UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

	Washington, 200.10		
	Form 10-Q		
QUARTERI 1934	LY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES	S EXCHANGE ACT OF	
	For the quarterly period ended June 30, 2013		
	OR		
☐ TRANSITIO	ON REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES	S EXCHANGE ACT OF	
	For the transition period from to		
	Commission file number: 001-35362		
	TRIPADVISOR, INC.		
	(Exact name of registrant as specified in its charter)		
	Delaware 80-074 (State or other jurisdiction of incorporation or organization) Identificat	nployer	
	141 Needham Street Newton, MA 02464 (Address of principal executive office) (Zip Code)		
	Registrant's telephone number, including area code: (617) 670-6300		
during the preceding 1	ck mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the 2 months (or for such shorter period that the registrant was required to file such reports), and (2) has bast 90 days. Yes \boxtimes No \square		34
to be submitted and po	ck mark whether the registrant has submitted electronically and posted on its corporate Web site, if any osted pursuant to Rule 405 of Regulation S-T (\S 232.405 of this chapter) during the preceding 12 months of the submit and post such files). Yes \boxtimes No \square		
	ck mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, o ge accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchan		See
Large accelerated filer		Accelerated filer	
Non-accelerated filer	☐ (Do not check if a smaller reporting company)	Smaller reporting company	
Indicate by check mar	k whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes \Box	No ⊠	
	Class Outstanding Shares a	at July 18, 2013	

Common Stock, \$0.001 par value per share Class B common stock, \$0.001 par value per share

Outstanding Shares at July 18, 2013 130,185,554 shares 12,799,999 shares

TripAdvisor, Inc. Form 10-Q For the Quarter Ended June 30, 2013

Table of Contents

Deat I. Firm del Information	Page
Part I—Financial Information	
Item 1. Unaudited Consolidated Financial Statements	
Consolidated Statements of Operations (Unaudited) for the Three and Six Months Ended June 30, 2013 and 2012	3
Consolidated Statements of Comprehensive Income (Unaudited) for the Three and Six Months Ended June 30, 2013 and 2012	4
Consolidated Balance Sheets (Unaudited) at June 30, 2013 and December 31, 2012	5
Consolidated Statement of Changes in Stockholders' Equity (Unaudited) for the Six Months Ended June 30, 2013	6
Consolidated Statements of Cash Flows (Unaudited) for the Six Months Ended June 30, 2013 and 2012	7
Notes to Unaudited Consolidated Financial Statements	8
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	24
<u>Item 3. Quantitative and Qualitative Disclosures about Market Risk</u>	40
<u>Item 4. Controls and Procedures</u>	40
Part II—Other Information	
<u>Item 1. Legal Proceedings</u>	41
<u>Item 1A. Risk Factors</u>	41
<u>Item 2. Unregistered Sales of Equity Securities and Use of Proceeds</u>	41
<u>Item 6. Exhibits</u>	42
<u>Signature</u>	44

TRIPADVISOR, INC. UNAUDITED CONSOLIDATED STATEMENTS OF OPERATIONS (In thousands, except per share data)

	Three months ended June 30,		Jun	
	2013	2012	2013	2012
Revenue	\$192,615	\$141,408	\$362,023	\$273,535
Related-party revenue from Expedia	54,322	55,740	114,833	107,328
Total revenue	246,937	197,148	476,856	380,863
Costs and expenses:				
Cost of revenue (exclusive of amortization) (1)	4,284	2,926	7,928	5,660
Selling and marketing (2)	82,574	64,243	161,865	131,632
Technology and content (2)	31,903	21,574	60,718	39,415
General and administrative (2)	25,552	18,252	47,985	34,506
Depreciation	6,876	4,715	13,200	8,996
Amortization of intangible assets	1,630	1,760	2,739	3,599
Total costs and expenses	152,819	113,470	294,435	223,808
Operating income	94,118	83,678	182,421	157,055
Other income (expense):				ŕ
Interest expense, net	(2,066)	(2,405)	(4,472)	(5,337)
Other, net	(2,150)	(4,539)	(3,611)	(3,843)
Total other expense, net	(4,216)	(6,944)	(8,083)	(9,180)
Income before income taxes	89,902	76,734	174,338	147,875
Provision for income taxes	(22,914)	(23,569)	(45,051)	(46,539)
Net income	66,988	53,165	129,287	101,336
Net (income) loss attributable to noncontrolling interests		(146)		(206)
Net income attributable to TripAdvisor, Inc.	\$ 66,988	\$ 53,019	\$129,287	\$101,130
Earnings Per Share attributable to TripAdvisor, Inc.:	·			
Basic	\$ 0.47	\$ 0.38	\$ 0.90	\$ 0.74
Diluted	\$ 0.46	\$ 0.37	\$ 0.89	\$ 0.73
Weighted Average Common Shares Outstanding:	,		,	
Basic	143,531	139,278	143,297	136,516
Diluted	145,664	141,737	145,160	138,947
(1) Excludes amortization as follows:				
Amortization of acquired technology included in amortization of intangibles	\$ 255	\$ 182	\$ 411	\$ 364
Amortization of website development costs included in depreciation	•		8,787	
Amortization of website development costs included in depreciation	4,630	2,979		5,692
(2) Indude the bond community of all and	\$ 4,885	\$ 3,161	\$ 9,198	\$ 6,056
(2) Includes stock-based compensation as follows:	ф <u>2244</u>	¢ 000	¢ 4550	¢ 2.001
Selling and marketing	\$ 2,244	\$ 923	\$ 4,559	\$ 2,001
Technology and content	\$ 4,024	\$ 2,426	\$ 10,422	\$ 3,938
General and administrative	\$ 3,940	\$ 3,419	\$ 8,838	\$ 5,521

TRIPADVISOR, INC. UNAUDITED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (In thousands)

	Three months ended June 30,		Six mont	
	2013	2012	2013	2012
Net income	\$66,988	\$53,165	\$129,287	\$101,336
Other comprehensive income (loss), net of tax:				
Foreign currency translation adjustments	95	(1,196)	(3,561)	231
Available-for-sale investments:				
Change in net unrealized gain (loss)	(593)	_	(545)	_
Less: reclassification adjustment for gains (losses) included in net income				
Net change (net of tax effect of \$404 and \$372)	(593)	_	(545)	_
Total other comprehensive income (loss), net of tax	(498)	(1,196)	(4,106)	231
Comprehensive income	66,490	51,969	125,181	101,567
Less: Comprehensive (income) loss attributable to noncontrolling interests		(146)		(206)
Comprehensive income attributable to TripAdvisor, Inc.	\$66,490	\$51,823	\$125,181	\$101,361

TRIPADVISOR, INC. UNAUDITED CONSOLIDATED BALANCE SHEETS (In thousands, except share and per share data)

ACCEPTE	June 30, 2013	December 31, 2012
ASSETS		
Current assets:	¢ 200.050	¢ 267 E1E
Cash and cash equivalents	\$ 200,058	\$ 367,515
Short-term marketable securities (note 4)	196,037	118,970
Accounts receivable, net of allowance of \$2,918 and \$2,818 at June 30, 2013 and December 31, 2012, respectively	137,981	81,459
Receivable from Expedia, Inc., net (note 14)	35,239	23,971
Taxes receivable	18,072	24,243
Deferred income taxes, net	7,026	5,971
Prepaid expenses and other current assets	17,059	10,365
Total current assets	611,472	632,494
Long-term marketable securities (note 4)	219,865	99,248
Property and equipment, net	54,601	43,802
Deferred income taxes, net	1,050	502
Other long-term assets	14,460	13,274
Intangible assets, net (note 7)	52,122	38,190
Goodwill (note 7)	502,015	471,684
TOTAL ASSETS	\$1,455,585	\$1,299,194
	41,100,000	4 -,,
LIABILITIES AND EQUITY		
Current liabilities:		
Accounts payable	\$ 9,153	\$ 12,796
Deferred merchant payables (note 2)	31,251	1,303
Deferred revenue	42,416	31,563
Credit facility borrowings (note 5)	24,633	32,145
Borrowings, current (note 5)	40,000	40,000
Taxes payable	12,006	14,597
Accrued expenses and other current liabilities (note 8)	81,840	63,236
Total current liabilities	241,299	195,640
Deferred income taxes, net	8,064	11,023
Other long-term liabilities	34,263	25,563
Borrowings, net of current portion (note 5)	320,000	340,000
Total Liabilities	603,626	572,226
Commitments and Contingencies (note 9)		
Stockholders' equity:		
Preferred stock \$0.001 par value	_	_
Authorized shares: 100,000,000		
Shares issued and outstanding: 0 and 0		
Common stock \$0.001 par value	131	130
Authorized shares: 1,600,000,000	131	150
Shares issued: 131,127,919 and 130,060,138		
Shares outstanding: 130,452,919 and 130,060,138	10	4.5
Class B common stock \$0.001 par value	13	13
Authorized shares 400,000,000		
Shares issued and outstanding: 12,799,999 and 12,799,999		=04.0=4
Additional paid-in capital	573,472	531,256
Retained earnings	325,725	196,438
Accumulated other comprehensive loss	(4,975)	(869
Treasury stock—Common stock, at cost, 675,000 and 0 shares, at June 30, 2013 and December 31, 2012 respectively	(42,407)	
Total stockholders' equity	851,959	726,968
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	\$1,455,585	\$1,299,194

TRIPADVISOR, INC. UNAUDITED CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' EQUITY FOR THE SIX MONTHS ENDED JUNE 30, 2013

(In thousands, except share data)

				Class I	R		Additional paid-in	Retained	(ımulated other orehensive			
	Common st	ock		common stock		capital	earnings	(loss) income		Treasury stock		Total	
	Shares	An	ount	Shares	An	nount					Shares	Amount	
Balance as of December 31, 2012	130,060,138	\$	130	12,799,999	\$	13	\$531,256	\$196,438	\$	(869)		<u>\$</u>	\$726,968
Net income attributable to													
TripAdvisor, Inc.								129,287					129,287
Currency translation adjustments										(3,561)			(3,561)
Unrealized gain on marketable													
securities, net of tax										(545)			(545)
Issuance of common stock related to exercise of options and vesting of													
RSUs	1,067,781		1				19,847						19,848
Repurchase of common stock											(675,000)	(42,407)	(42,407)
Tax benefits on equity awards							4,852						4,852
Minimum withholding taxes on net													
share settlements of equity awards							(5,534)						(5,534)
Stock-based compensation expense							23,051						23,051
Balance as of June 30, 2013	131,127,919	\$	131	12,799,999	\$	13	\$573,472	\$325,725	\$	(4,975)	(675,000)	\$(42,407)	\$851,959

TRIPADVISOR, INC. UNAUDITED CONSOLIDATED STATEMENTS OF CASH FLOWS (in thousands)

	Six months ended June 30,	
	2013	2012
Operating activities:		
Net income	\$ 129,287	\$101,336
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation of property and equipment, including amortization of internal-use software and website development	13,200	8,996
Stock-based compensation	23,819	11,460
Amortization of intangible assets	2,739	3,599
Amortization of deferred financing costs	401	475
Amortization of discounts and premiums on marketable securities, net	2,637	
Deferred tax benefit	(3,976)	(74)
Excess tax benefits from stock-based compensation	(5,242)	(2,011)
Provision (recovery) for doubtful accounts	656	(1,300)
Foreign currency transaction (gains) losses, net	2,539	2,323
Other, net	570	509
Changes in operating assets and liabilities, net of effects from acquisitions:	(== 00=)	(0= 4==)
Accounts receivable	(55,685)	(27,455)
Related parties	(11,273)	(34,356)
Taxes receivable	6,171	(5,950)
Prepaid expenses and other assets	(6,884)	(1,879)
Accounts payable	(4,831)	7,268
Deferred merchant payables	15,535	900
Taxes payable	11,905	4,128
Accrued expenses and other liabilities	3,983	15,940
Deferred revenue	7,736	7,568
Net cash provided by operating activities	133,287	91,477
Investing activities:		- ,
Acquisitions, net of cash acquired	(31,576)	_
Capital expenditures, including internal-use software and website development	(23,598)	(13,249)
Purchases of marketable securities	(346,593)	_
Sales of marketable securities	68,053	_
Maturities of marketable securities	76,929	_
Proceeds from Expedia, Inc. related to Spin-Off	_	7,028
Other, net	350	_
Net cash used in investing activities	(256,435)	(6,221)
Financing activities:		
Repurchase of common stock	(34,115)	_
Proceeds from credit facilities	6,701	7,627
Payments on credit facilities	(14,728)	(10,000)
Principal payments on long-term debt	(20,000)	(10,000)
Proceeds from exercise of stock options and warrants	19,847	225,769
Payment of minimum withholding taxes on net share settlements of equity awards	(5,534)	(3,035)
Excess tax benefits from stock-based compensation	5,242	2,011
Net cash (used) provided by financing activities	(42,587)	212,372
Effect of exchange rate changes on cash and cash equivalents	(1,722)	(2,919)
Net (decrease) increase in cash and cash equivalents	(167,457)	294,709
Cash and cash equivalents at beginning of year	367,515	183,532
Cash and cash equivalents at end of period	\$ 200,058	\$478,241

TRIPADVISOR, INC. UNAUDITED NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

NOTE 1: ORGANIZATION AND BASIS OF PRESENTATION

We refer to TripAdvisor, Inc. and our wholly-owned subsidiaries as "TripAdvisor," "the Company," "us," "we" and "our" in these notes to the consolidated financial statements.

During 2011, Expedia, Inc., or Expedia, announced its plan to separate into two independent public companies in order to better achieve certain strategic objectives of its various businesses. We refer to this transaction as the "Spin-Off." On December 20, 2011, following the close of trading on the NASDAQ Global Select Market ("NASDAQ"), the Spin-Off was completed, and TripAdvisor began trading as an independent public company on December 21, 2011. Expedia effected the Spin-Off by means of a reclassification of its capital stock that resulted in the holders of Expedia capital stock immediately prior to the time of effectiveness of the reclassification having the right to receive a proportionate amount of TripAdvisor capital stock. A one-for-two reverse stock split of outstanding Expedia capital stock occurred immediately prior to the Spin-Off, with cash paid in lieu of fractional shares. In connection with the Spin-Off, Expedia contributed or transferred all of the subsidiaries and assets relating to Expedia's TripAdvisor Media Group, which were comprised of the TripAdvisor Holdings, LLC combined financial statements, to TripAdvisor and TripAdvisor or one of its subsidiaries assumed all of the liabilities relating to Expedia's TripAdvisor Media Group.

Our common stock trades on the NASDAQ under the trading symbol "TRIP."

Description of Business

TripAdvisor is an online travel company, empowering users to plan and have the perfect trip. TripAdvisor's travel research platform aggregates reviews and opinions of members about destinations, accommodations (hotels, bed and breakfasts, specialty lodging and vacation rentals), restaurants and activities throughout the world through our flagship TripAdvisor brand. TripAdvisor-branded websites include tripadvisor.com in the United States and localized versions of the website in 29 countries, including in China under the brand daodao.com. Beyond travel-related content, TripAdvisor websites also include links to the websites of our travel advertisers allowing travelers to directly book their travel arrangements. In addition to the flagship TripAdvisor brand, we manage and operate 20 other travel media brands: www.airfarewatchdog.com, www.airfarewatchdog.com, www.airfarewatchdog.com, www.airfarewatchdog.com, www.airfarewatchdog.com, <a href="https://www.airfarewatchdog.com, www.airfarewatchdog.com, <a href="https://www.airfarewatchdog.com, www.airfarewatchdog.com, www.airfarewatchdog.com, www.airfarewatchdog.com, www.airfarewatchdog.com, <a href="https://www.airfarewatchdog.com, www.airfarewatchdog.com, <a href="https://www.airfarewatchdog.com, www.airfarewatchdog.com, www.airfarewatchdog.com, <a href="https://www.airfarewatchdog.com, www.airfarewatchdog.com, <a

Basis of Presentation

The accompanying unaudited consolidated financial statements have been prepared by us in accordance with generally accepted accounting principles, or GAAP, for interim financial reporting and the instructions to Form 10-Q and Rule 10-01 of Regulation S-X. Accordingly, certain information and footnote disclosures normally included in financial statements prepared under generally accepted accounting principles for complete periods have been condensed or omitted pursuant to such regulations. In the opinion of management, all adjustments considered necessary for a fair presentation of our financial position, results of operations and cash flows have been included. All such adjustments are of a normal recurring nature. These unaudited condensed consolidated financial statements should be read in conjunction with our audited consolidated financial statements for the year ended December 31, 2012 in our Annual Report on Form 10-K. The results for interim periods are not necessarily indicative of the results to be expected for the full year. The financial statements and related financial disclosures have been presented on a consolidated basis.

Consolidation

Our consolidated financial statements include the accounts of TripAdvisor, our wholly owned subsidiaries, and entities we control, or in which we have a variable interest and are the primary beneficiary of expected cash profits or losses. We record noncontrolling interest in our consolidated financial statements to recognize the minority ownership interest in our consolidated subsidiaries. Noncontrolling interest in the earnings and losses of consolidated subsidiaries represent the share of net income or loss allocated to members or partners in our consolidated entities. Significant intercompany transactions between the TripAdvisor consolidated entities and accounts have been eliminated.

Certain of our subsidiaries that operate in China have variable interests in affiliated entities in China in order to comply with Chinese laws and regulations, which restrict foreign investment in Internet content provision businesses. Although we do not own the capital stock of some of our Chinese affiliates, we consolidate their results as we are the primary beneficiary of the cash losses or profits of these variable interest affiliates and have the power to direct the activities of these affiliates. Although we currently operate at a loss in the Chinese market our variable interest entities are not material for all periods presented.

Accounting Estimates

We use estimates and assumptions in the preparation of our consolidated financial statements in accordance with GAAP. Our estimates and assumptions affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of our consolidated financial statements. These estimates and assumptions also affect the reported amount of net income or loss during any period. Our actual financial results could differ significantly from these estimates. The significant estimates underlying our consolidated financial statements include revenue recognition; recoverability of long-lived assets, valuation and impairment of marketable securities, intangible assets and goodwill; income taxes; useful lives of property and equipment; purchase accounting and stock-based compensation.

Reclassifications

We have reclassified certain amounts related to our prior period results to conform to our current period presentation.

Seasonality

Expenditures by travel advertisers tend to be seasonal. Traditionally, our strongest quarter has been the third quarter, which is a key travel research period, with the weakest quarter being the fourth quarter. However, adverse economic conditions or continued growth of our international operations with differing holiday peaks may influence the typical trend of our seasonality in the future.

NOTE 2: SIGNIFICANT ACCOUNTING POLICIES

Deferred Merchant Payables

We receive cash from travelers at the time of booking related to our transaction-based businesses and we record these amounts, net of commissions, on our consolidated balance sheets as deferred merchant payables. We pay after the travelers' use and subsequent billing from the hotel or vacation rental owners. Therefore, we receive cash from the traveler prior to paying the hotel and vacation rental owners, and this operating cycle represents a working capital source of cash to us. As long as our transaction-based businesses grow, we expect that changes in working capital related to these transactions will positively impact operating cash flows. As of June 30, 2013, our deferred merchant payables balance was \$31.3 million and the related transactions generated positive operating cashflow of \$15.5 million during the six months ended June 30, 2013. A payable balance of \$14.5 million was acquired with our business acquisitions during the three months ended June 30, 2013. For additional information on our business acquisitions refer to "Note 6—Acquisitions" below. The deferred merchant payables balance at December 31, 2012 was \$1.3 million.

There have been no material changes or other additions to our significant accounting policies since December 31, 2012. For additional information about our critical accounting policies and estimates, refer to "Note 2—*Significant Accounting Policies*," included in our Annual Report on Form 10-K for the year ended December 31, 2012.

Recently Adopted Accounting Pronouncements

Reporting of Amounts Reclassified Out of Accumulated Other Comprehensive Income

In February 2013, the Financial Accounting Standards Board, or FASB, issued new accounting guidance which adds new disclosure requirements for items reclassified out of accumulated other comprehensive income. The new guidance requires that companies present, either in a single note or parenthetically on the face of the financial statements, the effect of significant amounts reclassified based on its source and is effective for public companies in interim and annual reporting periods beginning after December 15, 2012. Accordingly, we have adopted these presentation requirements during the first quarter of 2013. The adoption of this new guidance did not have a material impact on our consolidated financial statements or disclosure.

NOTE 3: STOCK BASED AWARDS AND OTHER EQUITY INSTRUMENTS

Stock-based compensation expense relates primarily to expense for restricted stock units ("RSUs") and stock options. Our outstanding RSUs and stock options generally vest over four years.

For the three and six months ended June 30, 2013, we recognized total stock-based compensation expense of \$10.2 million and \$23.8 million, respectively. The total income tax benefit related to stock-based compensation expense was \$4.0 million and \$9.3 million for the three and six months ended June 30, 2013, respectively. For the three and six months ended June 30, 2012, we recognized total stock-based compensation expense of \$6.8 million and \$11.5 million, respectively. The total income tax benefit related to stock-based compensation expense was \$2.5 million and \$4.0 million for the three and six months ended June 30, 2012, respectively.

TripAdvisor, Inc. 2011 Stock and Annual Incentive Plan, As Amended

On December 20, 2011, our 2011 Stock and Annual Incentive Plan became effective. At our annual meeting of stockholders held on June 28, 2013 (the "Annual Meeting"), our stockholders approved an amendment to our 2011 Stock and Annual Incentive Plan to increase the aggregate number of shares of common stock authorized for issuance thereunder by 15,000,000 shares. We refer to our 2011 Stock and Annual Incentive Plan, as amended by the amendment as the "2011 Incentive Plan." A summary of the material terms of the 2011 Incentive Plan can be found in "Proposal 3: Approval of The 2011 Stock and Annual Incentive Plan, as amended" in our Proxy Statement on Schedule 14A for the Annual Meeting. The summary of the material terms of the 2011 Incentive Plan is qualified in its entirety by the full text of the 2011 Incentive Plan and the first amendment thereto, which are included in this Quarterly Report on Form 10-Q as Exhibits 4.2 and 4.1, respectively.

2013 Stock Option Activity

The exercise price for all stock options granted by us to date has been equal to the market price of the underlying shares of common stock at the date of grant. In this regard, when making stock option awards, our practice is to determine the applicable grant date and to specify that the exercise price shall be the closing price of our common stock on the date of grant. Stock options granted during the first six months of 2013 have a term of ten years from the date of grant and generally vest over a four-year service period.

During the six months ended June 30, 2013, we granted 1,561,789 primarily service-based stock options under the 2011 Incentive Plan with a weighted average grant-date fair value per option of \$23.15. We will amortize the fair value, net of estimated forfeitures, as stock-based compensation expense over the vesting term of generally four years on a straight-line basis, with the amount of compensation expense recognized at any date at least equaling the portion of the grant-date fair value of the award that is vested at that date. We use historical data to estimate pre-vesting option forfeitures and record share-based compensation expense only for those awards that are expected to vest. Changes in estimated forfeitures are recognized through a cumulative catch-up adjustment in the period of change which also impacts the amount of stock compensation expense to be recognized in future periods.

A summary of the status and activity for stock option awards relating to our common stock for the six months ended June 30, 2013, is presented below:

	Options Outstanding (In thousands)	Weighted Average Exercise Price Per Share	Weighted Average Remaining Contractual Life (In years)	 Aggregate Intrinsic Value thousands)
Options outstanding at January 1, 2013	8,654	\$ 31.41		
Granted	1,562	46.35		
Exercised	875	22.69		\$ 26,740
Cancelled	321	36.23		
Options outstanding at June 30, 2013	9,020	\$ 34.67	6.0	\$ 236,321
Exercisable as of June 30, 2013	3,756	\$ 29.03	3.5	\$ 119,593
Vested and expected to vest after June 30, 2013	8,439	\$ 34.12	6.0	\$ 223,343

Aggregate intrinsic value represents the difference between the closing stock price of our common stock and the exercise price of outstanding, in-themoney options. Our closing stock price as reported on NASDAQ as of June 28, 2013 was \$60.87.

The estimated fair value of the options granted under the 2011 Incentive Plan was calculated using a Black-Scholes Merton option-pricing model ("Black-Scholes model"). The Black-Scholes model incorporates assumptions to value stock-based awards, which includes the risk-free rate of return, expected volatility, expected term and expected dividend yield.

Our risk-free interest rate is based on the rates currently available on zero-coupon U.S. Treasury issues, in effect at the time of the grant, whose remaining maturity period most closely approximates the stock option's expected term assumption. We estimated the volatility of our common stock by using an average of historical stock price volatility of publicly traded companies that we consider peers based on daily price observations over a period equivalent to or approximate to the expected term of the stock option grants. The decision to use a weighted average volatility factor of a peer group was based upon the relatively short period of availability of data on our common stock. We estimated our expected term using the simplified method for all stock options as we do not have sufficient historical exercise data on our common stock. Our expected dividend yield is zero, as we have not paid any dividends on our common stock to date.

The fair value of stock option grants under the 2011 Incentive Plan has been estimated at the date of grant using the Black–Scholes model with the following weighted average assumptions for the periods presented below:

		Three months ended June 30,		s ended 30,
	2013	2012	2013	2012
Risk free interest rate	1.07%	1.05%	1.07%	1.05%
Expected term (in years)	6.25	6.21	6.25	6.21
Expected volatility	51.82%	53.34%	51.73%	53.34%
Expected dividend yield	0%	0%	0%	0%

2013 RSU Activity

During the six months ended June 30, 2013, we granted 994,293 primarily service-based RSUs under the 2011 Incentive Plan for which the fair value was measured based on the quoted price of our common stock at the date of grant. We will amortize the fair value, net of estimated forfeitures, as stock-based compensation expense over the vesting term of generally four years on a straight-line basis, with the amount of compensation expense recognized at any date at least equaling the portion of the grant-date fair value of the award that is vested at that date.

The following table presents a summary of RSU activity on our common stock:

	RSUs Outstanding (in thousands)	A (Da	eighted verage Grant- ate Fair Per Share
Unvested RSUs outstanding as of January 1, 2013	446	\$	26.11
Granted	994		47.05
Vested and released (1)	306		22.01
Cancelled	40		41.23
Unvested RSUs outstanding as of June 30, 2013	1,094	\$	45.13

(1) Inclusive of 117,403 RSUs withheld to satisfy employee minimum tax withholding requirements using net shares settlement method. Total payments for the employees' tax obligations to the taxing authorities were \$5.5 million and \$3.0 million for the six months ended June 30, 2013 and 2012, respectively, and are reflected as a financing activity within the consolidated statements of cash flows. RSUs that are withheld to satisfy employee minimum tax withholding requirements remain in the authorized but unissued pool under the 2011 Incentive Plan.

A summary of the unrecognized compensation expense, net of estimated forfeitures, and the weighted average period remaining at June 30, 2013 related to our non-vested stock options and RSU awards is presented below (in thousands, except year information):

	Stock	
	Options	RSUs
Unrecognized compensation expense (net of forfeitures)	\$77,179	\$32,234
Weighted average period remaining (in years)	2.89	3.51

All shares of common stock issued in respect of the exercise of options or other equity awards since Spin-Off have been issued from authorized, but unissued common stock.

NOTE 4: FINANCIAL INSTRUMENTS

Cash, Cash Equivalents and Marketable Securities

The following tables show our cash and available-for-sale securities' amortized cost, gross unrealized gains, gross unrealized losses and fair value by significant investment category recorded as cash and cash equivalents or short and long-term marketable securities as of June 30, 2013 and December 31, 2012 (in thousands):

				June 30, 2013	3		
	Amortized Cost	Unrealized Gains	Unrealized Losses	Fair Value	Cash and Cash Equivalents	Short-Term Marketable Securities	Long-Term Marketable Securities
Cash	\$154,775	\$ —	\$ —	\$154,775	\$ 154,775	\$ —	\$ —
Level 1:							
Money market funds	10,963			10,963	10,963		
Subtotal	10,963			10,963	10,963		
Level 2:							
U.S. agency securities	79,048	11	(126)	78,933	_	29,375	49,558
U.S. treasury securities	49,616	_	(1)	49,615	34,320	15,295	
Certificates of deposit	23,701	28		23,729		21,127	2,602
Commercial paper	15,492	4	_	15,496	_	15,496	_
Corporate securities	278,457	16	(1,023)	277,450	_	109,745	167,705
Municipal securities	5,001		(2)	4,999		4,999	
Subtotal	451,315	59	(1,152)	450,222	34,320	196,037	219,865
Total	\$617,053	\$ 59	\$ (1,152)	\$615,960	\$ 200,058	\$ 196,037	\$219,865
				December 31, 20			
	Amortized Cost	Unrealized Gains	Unrealized Losses	Fair Value	Cash and Cash Equivalents	Short-Term Marketable Securities	Long-Term Marketable Securities
Cash	\$141,460	\$ —	\$ —	\$141,460	\$ 141,460	\$ —	\$ —
Level 1:							
Money market funds	215,052			215,052	215,052		
Subtotal	215,052			215,052	215,052		
Level 2:							
U.S. agency securities	13,634	4	(3)	13,635	_	7,635	6,000
Commercial paper	48,710	15	(22)	48,703	9,999	38,704	_
Corporate securities	162,050	12	(180)	161,882	1,004	67,630	93,248
Municipal securities	5,003		(2)	5,001		5,001	
Subtotal	229,397	31	(207)	229,221	11,003	118,970	99,248
Total	\$585,909	\$ 31	\$ (207)	\$585,733	\$367,515	\$ 118,970	\$ 99,248

Our cash and cash equivalents consist of cash on hand in global financial institutions, money market funds and marketable securities with maturities of 90 days or less at the date purchased. The remaining maturities of our long-term marketable securities range from one to three years and our short-term marketable securities include maturities that were greater than 90 days at the date purchased and have 12 months or less remaining at June 30, 2013.

We classify our cash equivalents and marketable securities within Level 1 and Level 2 as we value our cash equivalents and marketable securities using quoted market prices (Level 1) or alternative pricing sources (Level 2). The valuation technique we used to measure the fair value of money market funds were derived from quoted prices in active markets for identical assets or liabilities. Investments in U.S. Treasury securities are considered "Level 2" valuations because we have access to quoted prices, but do not have visibility to the volume and frequency of trading for all investments. Fair values for our U.S. agency securities, commercial paper, corporate securities, certificates of deposit and municipal securities are considered "Level 2" valuations because they are obtained from pricing sources for identical or comparable instruments, rather than direct observations of quoted prices in active markets.

There were no material realized gains or losses related to sales of our marketable securities for the three and six months ended June 30, 2013 and 2012.

As of June 30, 2013, we have marketable securities with a total fair value of \$332.5 million in a total gross unrealized loss position of \$1.2 million. We consider the declines in market value of our marketable securities investment portfolio to be temporary in nature and do not consider any of our investments other-than-temporarily impaired. When evaluating an investment for other-than-temporary impairment, we review factors such as the length of time and extent to which fair value has been below its cost basis, the financial condition of the issuer and any changes thereto, and our intent to sell, or whether it is more likely than not we will be required to sell the investment before recovery of the investment's cost basis. During the six months ended June 30, 2013 and 2012, we did not recognize any impairment charges. We did not have any investments in marketable securities that were in a continuous unrealized loss position for 12 months or greater at June 30, 2013 or December 31, 2012.

Derivative Financial Instruments

In the normal course of business, we are exposed to the impact of foreign currency fluctuations, which we attempt to mitigate through the use of derivative instruments. Accordingly, we have entered into forward contracts to reduce the effects of fluctuating foreign currency exchange rates on our cash flows denominated in foreign currencies. We do not use derivatives for trading or speculative purposes. In accordance with current accounting guidance on derivative instruments and hedging activities, we record all our derivative instruments as either an asset or liability measured at their fair value. Our derivative instruments are typically short-term in nature.

Our current forward contracts are not designated as hedges. Consequently, any gain or loss resulting from the change in fair value is recognized in the current period earnings. These gains or losses are offset by the exposure related to receivables and payables with our foreign subsidiaries. We recorded a net loss of \$0.1 million and a net gain of \$0.7 million for the three and six month periods ended June 30, 2013, respectively, related to our forward contracts in our consolidated statement of operations in Other, net. The net cash received or paid related to our derivative instruments are classified as operating in our consolidated statements of cash flows, which is based on the objective of the derivative instruments. No derivative instruments were entered into or settled during the three and six months ended June 30, 2012.

The following tables show the fair value and notional principal amounts of our outstanding or unsettled derivative instruments that are not designated as hedging instruments (in thousands):

	Balance Sheet Caption	<u></u>	June 30, 2013	
		Fair Value Asset	of Derivative (2) Liability	U.S. Dollar Notional
Foreign exchange-forward contracts (current)	Accrued and other current liabilities (1)	\$ —	\$ 1	\$ 23,042
	Balance Sheet Caption	_	December 31, 2012	
		Fair Value Asset	of Derivative (2) Liability	U.S. Dollar Notional
Foreign exchange-forward contracts (current)	Accrued and other current liabilities (1)	<u>\$</u>	\$ 64	\$ 2,710

- (1) Derivative contracts address foreign exchange fluctuations for the Euro versus the U.S. dollar.
- (2) The fair value of our derivatives are measured using Level 2 fair value inputs, as we use a pricing model that takes into account the contract terms as well as current foreign currency exchange rates in active markets.

Concentration of Credit Risk

Counterparties to currency exchange derivatives consist of major international financial institutions. We monitor our positions and the credit ratings of the counterparties involved and, by policy limits, the amount of credit exposure to any one party. While we may be exposed to potential losses due to the credit risk of non-performance by these counterparties, losses are not anticipated.

Other Financial Instruments

Other financial instruments not measured at fair value on a recurring basis include trade receivables, related party receivables, trade payables, deferred merchant payables, short-term debt, accrued and other current liabilities and long-term debt. With the exception of long-term debt, the carrying amount approximates fair value because of the short maturity of these instruments as reported on our consolidated balance sheets as of June 30, 2013 and December 31, 2012. The carrying value of the long-term borrowings outstanding on our Credit Agreement bears interest at a variable rate and therefore is also considered to approximate fair value.

We did not have any Level 3 assets or liabilities for the periods ended June 30, 2013 and December 31, 2012.

NOTE 5: DEBT

Term Loan Facility Due 2016 and Revolving Credit Facility

Overview

On December 20, 2011, in connection with the Spin-Off, we entered into a credit agreement, dated as of December 20, 2011, by and among TripAdvisor, TripAdvisor Holdings, LLC, and TripAdvisor LLC, the lenders party thereto, JPMorgan Chase Bank, N.A., as administrative agent, and J.P. Morgan Europe Limited, as London agent (this credit agreement, together with all exhibits, schedules, annexes, certificates, assignments and related documents contemplated thereby, is referred to herein as the "Credit Agreement"), which provides \$600 million of borrowing including:

- the Term Loan Facility, or Term Loan, in an aggregate principal amount of \$400 million with a term of five years due December 2016; and
- the Revolving Credit Facility in an aggregate principal amount of \$200 million available in U.S. dollars, Euros and British pound sterling with a term of five years expiring December 2016.

The Term Loan and any loans under the Revolving Credit Facility bear interest by reference to a base rate or a Eurocurrency rate, in either case plus an applicable margin based on our leverage ratio. We are also required to pay a quarterly commitment fee, on the average daily unused portion of the Revolving Credit Facility for each fiscal quarter and fees in connection with the issuance of letters of credit. The Term Loan and loans under the Revolving Credit Facility currently bear interest at LIBOR plus 175 basis points, or the Eurocurrency Spread, or the alternate base rate ("ABR") plus 75 basis points, and undrawn amounts are currently subject to a commitment fee of 30 basis points.

As of June 30, 2013, we are using a one-month interest period Eurocurrency spread, which is approximately 2.0% per annum. Interest is currently payable on a monthly basis while we are borrowing under the one-month interest rate period. The current interest rates are based on current assumptions, leverage and LIBOR rates and do not take into account that rates will reset periodically.

The Term Loan principal is currently repayable in quarterly installments on the last day of each calendar quarter equal to 2.5% of the original principal amount with the balance due on the final maturity date. Principal payments aggregating \$20 million were made during the six months ended June 30, 2013.

The Revolving Credit Facility includes \$40 million of borrowing capacity available for letters of credit and \$40 million for borrowings on same-day notice. As of June 30, 2013, there are no outstanding borrowings under our Revolving Credit Facility. During the three and six months ended June 30, 2013, we recorded total interest and commitment fees on our Credit Agreement of \$2.0 million and \$4.1 million, respectively, to interest expense on our consolidated statement of operations. During the three and six months ended June 30, 2012, we recorded total interest and commitment fees on our Credit Agreement of \$2.1 million and \$4.4 million, respectively, to interest expense on our consolidated statements of operations. All unpaid interest and commitment fee amounts as of June 30, 2013 and December 31, 2012 were not material.

Total outstanding borrowings under the Credit Agreement consist of the following (in thousands):

	June 30, 2013
Short-Term Debt:	
Revolving Credit Facility	\$ —
Term Loan	40,000
Total Short-Term Borrowings	\$ 40,000
Long-Term Debt:	
Term Loan	\$320,000
Total Long-Term Borrowings	\$320,000

The remaining future minimum principal payment obligations due under the Credit Agreement related to our Term Loan is as follows as of June 30, 2013 (in thousands):

		Payment
		Amount
201	13 (remaining six months)	\$ 20,000
201	14	\$ 40,000
201	15	\$ 40,000
201	16	\$260,000
Total		\$360,000

Prepayments

We may voluntarily repay any outstanding borrowing under the Credit Agreement at any time without premium or penalty, other than customary breakage costs with respect to eurocurrency loans.

Guarantees

All obligations under the Credit Agreement are unconditionally guaranteed by us and each of our existing and subsequently acquired or organized direct or indirect wholly-owned domestic and foreign restricted subsidiaries, subject to certain exceptions for subsidiaries that are controlled foreign corporations, foreign subsidiaries in jurisdictions where applicable law would otherwise be violated, and non-material subsidiaries.

Covenants

The Credit Agreement contains a number of covenants that, among other things, restrict our ability to: incur additional indebtedness, create liens, enter into sale and leaseback transactions, engage in mergers or consolidations, sell or transfer assets, pay dividends and distributions, make investments, loans or advances, prepay certain subordinated indebtedness, make certain acquisitions, engage in certain transactions with affiliates, amend material agreements governing certain subordinated indebtedness, and change our fiscal year. The Credit Agreement also requires us to maintain a maximum leverage ratio and a minimum cash interest coverage ratio, and contain certain customary affirmative covenants and events of default, including a change of control. If an event of default occurs, the lenders under the Credit Agreement will be entitled to take various actions, including the acceleration of all amounts due under Credit Agreement and all actions permitted to be taken by a secured creditor.

As of June 30, 2013 we believe we are in compliance with all of our debt covenants.

The full text of the Credit Agreement is incorporated by reference to Exhibit 4.2 of our Annual Report on Form 10-K for the year ending December 31, 2012.

Chinese Credit Facilities

In addition to our borrowings under the Credit Agreement, we maintain our Chinese Credit Facilities. As of June 30, 2013 and December 31, 2012, we had \$24.6 million and \$32.1 million of short-term borrowings outstanding, respectively.

Certain of our Chinese subsidiaries entered into a RMB 138,600,000 (approximately \$22 million), one-year revolving credit facility with Bank of America (the "Chinese Credit Facility—BOA") that is currently subject to review on a periodic basis with no-specific expiration period. During the third quarter of 2012, this credit line was increased to RMB 189,000,000 (approximately \$30 million). During the six months ended June 30, 2013, we made a payment inclusive of interest of RMB 68,283,570 (approximately \$10.9 million). We currently have \$11.3 million of outstanding borrowings from the Chinese Credit Facility—BOA as of June 30, 2013. Our Chinese Credit Facility—BOA currently bears interest at a rate based on 100% of the People's Bank of China's base rate and was 5.6% as of June 30, 2013.

In addition, during April 2012, certain of our Chinese subsidiaries entered into a RMB 125,000,000 (approximately \$20 million) one-year revolving credit facility with J.P. Morgan Chase Bank ("Chinese Credit Facility-JPM"). This credit facility was renewed for an additional year in April 2013. During the six months ended June 30, 2013, we made a payment inclusive of interest of RMB 24,281,546 (approximately \$3.9 million). We currently have \$13.3 million of outstanding borrowings from the Chinese Credit Facility—JPM as of June 30, 2013. Our Chinese Credit Facility—JPM currently bears interest at a rate based on 100% of the People's Bank of China's base rate and was 5.6% as of June 30, 2013.

NOTE 6: ACQUISITIONS

During the six months ended June 30, 2013, we completed five acquisitions for total cash consideration paid of \$31.6 million, net of cash acquired. The total cash consideration is subject to adjustment based on the finalization of working capital adjustments and amounts retained with payment subject to certain indemnification obligations by the respective sellers for our benefit in future periods. We acquired TinyPost, the developer of a product that enables users to write over photos and turn them into stories, Jetsetter, a members-only private sale site for hotel bookings, CruiseWise, a cruise research and planning site, Niumba, a Spain-based vacation rental site, and GateGuru, a mobile app with flight and airport information around the world, all of which complement our existing brands in those areas of the travel ecosystem. The total purchase price of these acquisitions, all of which were accounted for as purchases of businesses under the acquisition method, have been allocated to the tangible and identifiable intangible assets acquired and the net liabilities assumed based on their respective fair values on the acquisition date. The purchase price allocation of our 2013 acquisitions are preliminary and subject to revision as more information becomes available, but in any case will not be revised beyond 12 months after the acquisition date and any change to the fair value of net liabilities acquired will lead to a corresponding change to the purchase price allocable to goodwill. Acquisition-related costs were expensed as incurred and were not significant.

The following table presents the purchase price allocations initially recorded on our consolidated balance sheet on June 30, 2013 for all 2013 acquisitions (in thousands):

	Total
Net liabilities (1)	\$(14,195)
Goodwill (2)	34,137
Intangible assets (3)	16,944
Deferred tax liability	(1,352)
Total purchase price consideration (4)	\$ 35,534

- (1) Includes cash acquired of \$2.9 million.
- (2) The goodwill represents the excess value over both tangible and intangible assets acquired. The goodwill in these transactions is primarily attributable to expected operational synergies, the assembled workforces, and the future development initiatives of the assembled workforces. Goodwill in the amount of \$18.1 million is expected to be deductible for tax purposes.
- (3) Identifiable definite-lived intangible assets were comprised of developed technology of \$1.6 million, trade names of \$7.0 million, customer relationships of \$7.9 million, and non-compete agreements of \$0.4 million. The overall weighted-average life of the identifiable definite-lived intangible assets acquired in the purchase of the companies was 8.3 years, which will be amortized on a straight-line basis over their estimated useful lives.
- (4) Subject to adjustment based on (i) final working capital adjustment calculations, and (ii) indemnification obligations of the acquired company stockholders.

Our consolidated financial statements include the operating results of all acquired businesses from the date of each acquisition. Pro forma results of operations for all 2013 acquisitions have not been presented as the financial impact to our consolidated financial statements both individually and in aggregate, are not material. There were no acquisitions completed during the six months ended June 30, 2012.

NOTE 7: GOODWILL AND INTANGIBLE ASSETS, NET

The change in goodwill for the six months ended June 30, 2013 consists of the following (in thousands):

	Total
Balance at December 31, 2012	\$471,684
Additions (1)	34,137
Purchase accounting adjustments (2)	(872)
Foreign exchange translation adjustments	(2,934)
Balance at June 30, 2013	\$502,015

- (1) The additions to goodwill relate to our business acquisitions. See "Note 6—*Acquisitions*," above for further information.
- (2) Purchase price allocation adjustment is related to our 2012 acquisition.

Intangible assets, which were acquired in business combinations and recorded at fair value on the date of purchase, consist of the following:

	June 30, 2013	December 31, 2012
	(In the	usands)
Intangible assets with definite lives	\$ 37,790	\$ 21,382
Less: accumulated amortization	(15,968)	(13,492)
Intangible assets with definite lives, net	21,822	7,890
Intangible assets with indefinite lives	30,300	30,300
	\$ 52,122	\$ 38,190

The following table presents the components of our intangible assets with definite lives:

			June 30, 2013		1	December 31, 2012	
	Weighted Ave Remaining Life (in years)	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount
			(In thousands)			(In thousands)	
Trade names and trademarks	8.1	\$21,217	\$ (10,384)	\$10,833	\$14,431	\$ (9,029)	\$ 5,402
Subscriber relationships	6.3	13,311	(4,270)	9,041	5,617	(3,511)	2,106
Technology and other	2.6	3,262	(1,314)	1,948	1,334	(952)	382
Total	6.8	\$37,790	\$ (15,968)	\$21,822	\$21,382	\$ (13,492)	\$ 7,890

Intangible assets with definite lives are amortized on a straight-line basis. The estimated amortization expense for intangible assets with definite lives for the remainder of 2013, the annual expense for the next five years, and the expense thereafter, assuming no subsequent impairment of the underlying assets, is expected to be as follows (in thousands):

2013 (remaining six months)	\$ 2,871
2014	4,553
2015	2,674
2016	2,177
2017	2,017
2018	2,017
2019 and thereafter	5,513
Total	\$21,822

NOTE 8: ACCRUED EXPENSES AND OTHER CURRENT LIABILITIES

Accrued expenses and other current liabilities consisted of the following:

	June 30, 2013	December 31, 2012
	(In	thousands)
Accrued salary, bonus, and related benefits	\$26,598	\$ 29,438
Accrued marketing costs	19,251	11,941
Accrued charitable foundation payments	4,252	6,757
Accrued treasury share purchases	8,292	_
Other	23,447	15,100
Total accrued expenses and other current liabilities	\$81,840	\$ 63,236

NOTE 9: COMMITMENTS AND CONTINGENCIES

Office Lease Commitments

On June 20, 2013, TripAdvisor LLC ("TA LLC"), our indirect, wholly owned subsidiary, entered into a lease (the "Lease"), with Normandy Gap-V Needham Building 3, LLC, as landlord (the "Landlord"). Pursuant to the Lease, the Landlord will build an approximately 280,000 square foot rental building in Needham, Massachusetts (the "Premises"), and thereafter lease the Premises to TA LLC as TripAdvisor's new corporate headquarters for an initial term of 15 years and 7 months. If the Landlord fails to deliver the Premises according to the schedule, subject to certain conditions, TA LLC may be entitled to additional free rent, or in extreme cases, a right to terminate the Lease. Under the Lease, TA LLC is required to pay an initial base rent of \$33.00 per square foot per year, increasing to \$34.50 per square foot by the final year of the initial term, as well as all real estate taxes and other building operating costs.

The aggregate future minimum lease payments are \$143.5 million and are currently scheduled to be paid, beginning in November 2015, as follows: \$1.1 million for 2015, \$9.3 million for 2016, \$9.3 million for 2017, and \$123.9 million for 2018 and thereafter. The Lease has escalating rental payments and initial periods of free rent and therefore the effects of variable rent payments will be expensed on a straight-line basis over the life of the Lease in accordance with GAAP. All real estate taxes and other building operating costs paid during the term of the Lease will be expensed as incurred. TA LLC is also obligated to deliver a letter of credit to the Landlord in the amount of \$0.8 million as security deposit, which amount is subject to increase under certain circumstances. Subject to certain conditions, TA LLC has certain rights under the Lease, including rights of first offer to lease additional space or to purchase the Premises if the Landlord elects to sell. TA LLC also has an option to extend the term of the Lease for two consecutive terms of five years each.

In connection with the Lease, on June 20, 2013, TripAdvisor entered into a Guaranty (the "Guaranty"), pursuant to which TripAdvisor provides full payment and performance guaranty for all of TA LLC's obligations under the Lease.

The full text of the Lease and the Guaranty are filed with this Quarterly Report on Form 10-Q as Exhibits 10.1 and 10.2, respectively.

The following table presents our estimated future minimum rental payments under all operating leases with non-cancelable lease terms having initial terms in excess of one year as of June 30, 2013 (in thousands):

	Total
	Payments
2013 (remaining six months)	\$ 5,125
2014	8,494
2015	5,774
2016	10,598
2017	10,598
2018 and thereafter	130,008
Total (1)	\$170,597

Excludes spending on anticipated leasehold improvements on our Needham, Massachusetts lease, including design, development, construction costs, the
purchase and installation of equipment, and receipt of related Landlord incentives which we estimate will begin in October 2014.

There have been no other material changes to our commitments and contingencies since December 31, 2012. (Refer to "Note 12—*Commitments and Contingencies*," in the Notes to our Consolidated Financial Statements in Item 8 of our Annual Report on Form 10-K for the year ended December 31, 2012.)

Legal Proceedings

In the ordinary course of business, we and our subsidiaries are parties to legal proceedings and claims involving alleged infringement of third-party intellectual property rights, defamation, and other claims. Rules of the U.S. Securities and Exchange Commission require the description of material pending legal proceedings, other than ordinary, routine litigation incident to the registrant's business, and advise that proceedings ordinarily need not be described if they primarily involve damages claims for amounts (exclusive of interest and costs) not individually exceeding 10% of the current assets of the registrant and its subsidiaries on a consolidated basis. In the judgment of management, none of the pending litigation matters that the Company and its subsidiaries are defending involves or is likely to involve amounts of that magnitude. There may be claims or actions pending or threatened against us of which we are currently not aware and the ultimate disposition of which could have a material adverse effect on us.

NOTE 10: INCOME TAXES

Each interim period is considered an integral part of the annual period and, accordingly, we measure our tax expense using an estimated annual effective tax rate. An enterprise is required, at the end of each interim reporting period, to make its best estimate of the annual effective tax rate for the full fiscal year and use that rate to provide for income taxes on a current year-to-date basis, as adjusted for discrete taxable events that occur during the interim period.

Our effective tax rate for the three and six months ended June 30, 2013 was 25.5% and 25.8%, respectively. Our effective tax rate for the three and six months ended June 30, 2012 was 30.7% and 31.5%, respectively. For the three and six months ended June 30, 2013, the effective tax rate is less than the federal statutory rate primarily due to earnings in jurisdictions outside the United States, where our effective tax rate is lower, which was partially offset by state income taxes, non-deductible stock compensation and accruals on uncertain tax positions. The decrease in the effective tax rate for 2013 compared to the 2012 rate was primarily due to an increase in earnings in jurisdictions outside the United States as well as an internal restructuring that occurred during the fourth quarter of 2012. This restructuring was undertaken within our non-US operations to align our global structure for more efficient treasury management and global cash deployment.

Our policy is to recognize accrued interest and penalties related to unrecognized tax benefits and income tax liabilities as part of our income tax expense. As of June 30, 2013, accrued interest is \$0.8 million, net of federal benefit, and no penalties have been accrued. We do not anticipate any material releases in the next twelve months.

For all periods prior to and through the Spin-Off date, we were a member of the Expedia consolidated tax group. Accordingly, Expedia filed a consolidated federal income tax return and certain state income tax returns with us for that period. Expedia has paid the entire income tax liability associated with these filings. As such, our income tax liability for this period was transferred to Expedia upon Spin-Off and was not included in income taxes payable as of December 31, 2011. Additionally, due to continuing ownership and business relationships after the Spin-Off, we may be considered to have a unitary relationship with Expedia from January 1, 2012 through December 11, 2012 for state income tax purposes. Consequently, we may file as part of a unitary combined group with Expedia for certain state tax returns for 2012.

As of June 30, 2013, the Company's tax years for 2009 through 2012 are subject to examination by the tax authorities in the United States, various states, and foreign jurisdictions. By virtue of previously filed consolidated income tax returns filed with Expedia, we are routinely under audit. We are currently under an IRS audit for the 2009 and 2010 tax years, and have various ongoing state income tax audits. As of June 30, 2013, no material assessments have resulted from these audits. These audits include questioning the timing and the amount of income and deductions and the allocation of income among various tax jurisdictions. Annual tax provisions include amounts considered sufficient to pay assessments that may result from the examination of prior year returns. We are no longer subject to tax examinations by tax authorities for years prior to 2005.

NOTE 11: STOCKHOLDERS' EQUITY

Common Stock and Class B Common Stock

Our authorized common stock consists of 1.6 billion shares of common stock with par value of \$0.001 per share, and 400 million shares of Class B common stock with par value of \$0.001 per share. Both classes of common stock qualify for and share equally in dividends, if declared by our Board of Directors. Common stock is entitled to one vote per share and Class B common stock is entitled to 10 votes per share on most matters. Holders of TripAdvisor common stock, acting as a single class, are entitled to elect a number of directors equal to 25% percent of the total number of directors, rounded up to the next whole number, which was two directors as of June 30, 2013. Class B common stockholders may, at any time, convert their shares into common stock, on a one for one share basis. Upon conversion, the Class B common stock is retired and is not available for reissue. In the event of liquidation, dissolution, distribution of assets or winding-up of TripAdvisor the holders of both classes of common stock have equal rights to receive all the assets of TripAdvisor after the rights of the holders of the preferred stock have been satisfied. There were 131,127,919 and 130,452,919 shares of common stock issued and outstanding, respectively, at June 30, 2013 and 12,799,999 shares of Class B common stock issued and outstanding at June 30, 2013.

Preferred Stock

In addition to common stock, we are authorized to issue up to 100 million preferred shares, with \$ 0.001 par value per share, with terms determined by our Board of Directors, without further action by our stockholders. At June 30, 2013, no preferred shares had been issued.

Accumulated Other Comprehensive Loss

Accumulated other comprehensive loss is primarily comprised of accumulated foreign currency translation adjustments, as follows (in thousands):

	June 30, 2013	ember 31, 2012
Net unrealized loss on securities, net of tax (1)	\$ (649)	\$ (104)
Cumulative foreign currency translation adjustments (2)	(4,326)	 (765)
Total accumulated other comprehensive loss	<u>\$(4,975)</u>	\$ (869)

- 1) Net of unrealized tax benefits of \$0.4 million and \$0.1 million at June 30, 2013 and December 31, 2012, respectively.
- (2) Our foreign subsidiary earnings are considered indefinitely reinvested; therefore; deferred taxes are not provided on foreign currency translation adjustments.

Treasury Stock

On February 15, 2013, our Board of Directors authorized the repurchase of \$250 million of our shares of common stock under a share repurchase program. We intend to use available cash and future cash from operations to fund repurchases under the share repurchase program. The repurchase program has no expiration but may be suspended or terminated by the Board of Directors at any time. The Executive Committee of our Board of Directors will determine the price, timing, amount and method of such repurchases based on its evaluation of market conditions and other factors, and any shares repurchased will be in compliance with applicable legal requirements, at prices determined to be attractive and in the best interests of both the Company and its stockholders.

During the six month period ended June 30, 2013, we repurchased 675,000 shares of outstanding common stock under the share repurchase program at an aggregate cost of \$42.4 million, of which \$8.3 million remains payable at June 30, 2013. As of June 30, 2013, we have a remaining amount from the authorized share repurchase program granted by the Board of Directors of \$207.6 million to repurchase shares of our common stock.

Dividends

During the period January 1, 2013 through June 30, 2013, our Board of Directors did not declare any dividends on our outstanding common stock.

NOTE 12: SEGMENT INFORMATION

We have one reportable segment: TripAdvisor. We determined our segment based on how our chief operating decision maker manages our business, makes operating decisions and evaluates operating performance. Our primary operating metric for evaluating segment performance is Adjusted EBITDA. We define Adjusted EBITDA as net income (loss) plus: (1) provision for income taxes; (2) other (income) expense, net; (3) depreciation of property and equipment, including internal use software and website development; (4) amortization of intangible assets; (5) stock-based compensation; and (6) non-recurring expenses. Such amounts are detailed in our segment reconciliation below. In addition, please see our discussion of Adjusted EBITDA in Part I, Item 2, "Management's Discussion and Analysis of Financial Condition and Results of Operations" below.

The following table is a reconciliation of Adjusted EBITDA to net income, the most directly comparable financial measure calculated and presented in accordance with GAAP, for the periods presented (in thousands):

	Three months ended June 30,		Six mont Jun	
	2013	2012	2013	2012
Adjusted EBITDA	\$112,832	\$ 96,921	\$222,179	\$181,110
Depreciation (1)	(6,876)	(4,715)	(13,200)	(8,996)
Amortization of intangible assets	(1,630)	(1,760)	(2,739)	(3,599)
Stock-based compensation	(10,208)	(6,768)	(23,819)	(11,460)
Other expense, net	(4,216)	(6,944)	(8,083)	(9,180)
Provision for income taxes	(22,914)	(23,569)	(45,051)	(46,539)
Net income	\$ 66,988	\$ 53,165	\$129,287	\$101,336

(1) Includes amortization of internal use software and website development costs.

NOTE 13: EARNINGS PER SHARE

Basic Earnings Per Share

We compute basic earnings per share by dividing net income attributable to TripAdvisor by the weighted average number of common shares outstanding during the period. For the three and six months ended June 30, 2013 and 2012, we computed the weighted average number of common shares outstanding during the period using the total of common stock and Class B common stock outstanding as of December 31, 2012 and 2011, respectively, plus the weighted average of any additional shares issued and outstanding during the three and six months ended June 30, 2013 and 2012, respectively.

Diluted Earnings Per Share

We compute diluted earnings per share by dividing net income attributable to TripAdvisor by the sum of the weighted average number of common and common equivalent shares outstanding during the period. For the three and six months ended June 30, 2013 and 2012, we computed the weighted average number of common and common equivalent shares outstanding during the period using the sum of (i) the number of shares of common stock and Class B common stock used in the basic earnings per share calculation as indicated above, (ii) if dilutive, the incremental weighted average common stock that we would issue upon the assumed exercise of common equivalent shares related to stock options and stock warrants and the vesting of restricted stock units using the treasury stock method during the three and six months ended June 30, 2013 and 2012, respectively, and (iii) if dilutive, performance based awards based on the number of shares that would be issuable as of the end of the reporting period assuming the end of the reporting period was also the end of the contingency period.

Under the treasury stock method, the assumed proceeds calculation includes the actual proceeds to be received from the employee upon exercise, the average unrecognized compensation cost during the period and any tax benefits credited upon exercise to additional paid-in-capital. The treasury stock method assumes that a company uses the proceeds from the exercise of an award to repurchase common stock at the average market price for the period. Windfall tax benefits created upon the exercise of an award would be added to assumed proceeds, while shortfalls charged to additional paid-in-capital would be deducted from assumed proceeds. Any shortfalls not covered by the windfall tax pool would be charged to the income statement and would be excluded from the calculation of assumed proceeds, if any.

Below is a reconciliation of the weighted average number of shares of common stock outstanding in calculating diluted earnings per share (in thousands, except for per share information):

		nths ended e 30,	Six months ended June 30,		
	2013	2012	2013	2012	
Numerator:					
Net income attributable to TripAdvisor	\$ 66,988	\$ 53,019	\$129,287	\$101,130	
Denominator:					
Weighted average shares used to compute Basic EPS	143,531	139,278	143,297	136,516	
Weighted average effect of dilutive securities:					
Stock options	1,940	1,394	1,671	1,263	
RSUs	193	100	192	147	
Stock warrants		965		1,021	
Weighted average shares used to compute Diluted EPS	145,664	141,737	145,160	138,947	
Basic EPS	\$ 0.47	\$ 0.38	\$ 0.90	\$ 0.74	
Diluted EPS	\$ 0.46	\$ 0.37	\$ 0.89	\$ 0.73	

The following potential common shares related to stock options and RSUs were excluded from the calculation of diluted net income per share because their effect would have been anti-dilutive for the periods presented (in thousands):

		Three Months Ended June 30,		Six Months Ended June 30,	
	2013 (1)	2012 (2)	2013 (1)	2012 (2)	
Stock options	1,741	3,901	3,159	2,861	
RSUs	48		24	30	
Total	1,789	3,901	3,183	2,891	

- (1) These totals do not include 210,000 performance based options and 44,000 performance based RSUs representing the right to acquire 254,000 shares of common stock for which all targets required to trigger vesting have not been achieved as of June 30, 2013; therefore, such awards were excluded from the calculation of weighted average shares used to compute diluted earnings per share for those reporting periods.
- (2) These totals do not include performance based RSUs representing the right to acquire 400,000 shares of common stock for which all targets required to trigger vesting had not been achieved as of June 30, 2012; therefore, such awards were excluded from the calculation of weighted average shares used to compute diluted earnings per share for those reporting periods.

The earnings per share amounts are the same for common stock and Class B common stock because the holders of each class are legally entitled to equal per share distributions whether through dividends or in liquidation.

NOTE 14: RELATED PARTY TRANSACTIONS

Expedia

Subsequent to the Spin-Off, we remained a related party with Expedia due to common ownership or Liberty Interactive Corporation's ownership of Expedia stock, as applicable, and continue to work together pursuant to various commercial agreements between subsidiaries of TripAdvisor, on the one hand, and subsidiaries of Expedia, on the other hand.

Related-party revenue from Expedia was \$54.3 million and \$114.8 million for the three and six months ended June 30, 2013, respectively, and was \$55.7 million and \$107.3 million for the three and six months ended June 30, 2012, respectively, which primarily consists of click-based advertising and other advertising services provided to Expedia and its subsidiaries and is recorded at contract value, which we believe is a reasonable reflection of the value of the services provided. Related-party revenue represented 22.0% and 24.1% of our total revenue for the three and six months ended June 30, 2013, respectively, and 28.3% and 28.2% of our total revenue for the three and six months ended June 30, 2012, respectively. Other related-party operating expenses which were included within selling and marketing expense were \$1.4 million and \$3.2 million for the three and six months ended June 30, 2013, respectively, and were \$1.8 million and \$3.5 million for the three and six months ended June 30, 2012, respectively, which primarily consisted of marketing expense for exit windows. The net related party receivable balances with Expedia reflected in our consolidated balance sheets as of June 30, 2013 and December 31, 2012 were \$35.2 million and \$24.0 million, respectively.

Liberty Interactive Corporation

As of June 30, 2013, Liberty Interactive Corporation, or Liberty, beneficially owned 18,159,752 shares of our common stock and 12,799,999 shares of our Class B common stock, which shares constitute 13.9% of the outstanding shares of common stock and 100% of the outstanding shares of Class B Common Stock. Assuming the conversion of all of the Liberty's shares of Class B common stock into common stock, Liberty would beneficially own 21.6% of the outstanding common stock (calculated in accordance with Rule 13d-3). Because each share of Class B common stock generally is entitled to ten votes per share and each share of common stock is entitled to one vote per share, Liberty may be deemed to beneficially own equity securities representing approximately 56.5% of our voting power as of June 30, 2013.

We had no related party transactions with Liberty during the six months ended June 30, 2013 and 2012, respectively.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Forward-Looking Statements

This Quarterly Report on Form 10-Q contains "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. These forward-looking statements reflect the views of our management regarding current expectations and projections about future events and are based on currently available information. Actual results could differ materially from those contained in these forward-looking statements for a variety of reasons, including, but not limited to, those discussed in our Annual Report on Form 10-K for the year ended December 31, 2012, Part I, Item 1A, "Risk Factors," as well as those discussed elsewhere in this report. Other unknown or unpredictable factors also could have a material adverse effect on our business, financial condition and results of operations. Accordingly, readers should not place undue reliance on these forward-looking statements. The use of words such as "anticipates," "estimates," "expects," "intends," "plans" and "believes," among others, generally identify forward-looking statements; however, these words are not the exclusive means of identifying such statements. In addition, any statements that refer to expectations, projections or other characterizations of future events or circumstances are forward-looking statements. These forward-looking statements are inherently subject to uncertainties, risks and changes in circumstances that are difficult to predict. We are not under any obligation to, and do not intend to, publicly update or review any of these forward-looking statements, whether as a result of new information, future events or otherwise, even if experience or future events make it clear that any expected results expressed or implied by those forward-looking statements will not be realized. Please carefully review and consider the various disclosures made in this report and in our other reports filed with the U.S. Securities and Exchange Commission ("SEC") that attempt to advise interested parties of the risks and factors that may affect our bus

The information included in this management's discussion and analysis of financial condition and results of operations should be read in conjunction with our consolidated financial statements and the notes included in this Quarterly Report on Form 10-Q, and the audited consolidated financial statements and notes and Management's Discussion and Analysis of Financial Condition and Results of Operations contained in our Annual Report on Form 10-K for the year ended December 31, 2012.

Overview

We are the world's largest online travel company, empowering users to plan and have the perfect trip. Our travel research platform aggregates reviews and opinions from our community about destinations, accommodations (including hotels, resorts, motels, B&Bs, specialty lodging and vacation rentals), restaurants and activities throughout the world through our flagship TripAdvisor brand. Our branded websites include tripadvisor.com in the United States and localized versions of the website in 29 other countries, including China under the brand daodao.com. Our TripAdvisor branded websites globally averaged more than 220 million monthly unique visitors for the quarter ended June 30, 2013, according to Google Analytics. We feature over 100 million reviews and opinions and we have built a base of approximately 53 million marketable members. Marketable members are members that have registered on our site and given us permission to email them. While we believe growth in marketable members increases user engagement and traffic to our site we do not consider it to be a key driver of our overall revenue. Beyond travel-related content, our websites also include links to the websites of our customers, including travel advertisers, allowing travelers to directly book their travel arrangements. In addition to the flagship TripAdvisor brand, we now manage and operate 20 other travel media brands: www.airfarewatchdog.com, www.flipkey.com, <a href="https://www.f

For additional information about our portfolio of brands, see "Other Travel Brands and Websites" in Part I, Item 1, "Business," in our Annual Report on Form 10-K for the year ended December 31, 2012.

Business Model

We derive substantially all of our revenue from the sale of advertising, primarily through click-based advertising and, to a lesser extent, display-based advertising. The remainder of our revenue is generated through a combination of subscription-based offerings, selling room nights on our transactional sites, and other revenue including content licensing. For the six months ended June 30, 2013, we earned \$362 million in revenue from click-based advertising, \$56 million in revenue from display-based advertising and \$59 million in revenue from subscription-based offerings, transaction revenue and other revenue.

- Click-Based Advertising Revenue. Our largest source of revenue is click-based advertising, which includes links to our partners' booking sites and contextually-relevant branded and unbranded textlinks. Our click-based advertising partners are predominantly online travel agencies and direct suppliers in the hotel, airline and cruise product categories. Click-based advertising is generally priced on a cost-per-click, or CPC, basis, with payments from advertisers based on the number of users who click on each type of link. Most of our click-based advertising contracts can be terminated by the advertisers at will or on short notice.
- Display-Based Advertising Revenue. We earn revenue from a variety of display-based advertising placements on our websites through which our advertising partners can promote their brands in a contextually-relevant manner. While our display-based advertising clients are predominately direct suppliers in the hotel, airline and cruise categories and online travel agencies, we also accept display advertising from marketing organizations, casinos, resorts and attractions, as well as advertisers from non-travel categories. We generally sell our display-based advertising on a cost per thousand impressions, or CPM, basis. Our display-based advertising products also include a number of custom-built products including the sponsorship of certain site features and functionality, as well as certain customized co-branded features.
- Subscription-Based, Transaction and Other Revenue. We offer advertising via a subscription model that is sold for a flat fee per time period. Managed by our TripAdvisor for Business division, this advertising product, Business Listings, is currently offered to hotels, B&Bs and other specialty lodging properties and allows subscribers to list a website URL, email address and phone number on TripAdvisor-branded websites as well as to post special offers for travelers. Our Vacation Rentals product allows individual vacation property owners and property managers to pay a subscription fee or to use a new free-to-list option to list properties on our U.K.-based Holiday Lettings, U.S.-based FlipKey and our recently-acquired Spain-based Niumba websites, as well as on select TripAdvisor-branded websites. Other sources of revenue include selling room nights on our transactional sites, including our recently-acquired Jetsetter business and Tingo, as well as content licensing arrangements with third-party sites.

Executive Summary

Our financial results are currently principally dependent on our ability to drive our click-based advertising revenue. We continue to invest in areas of potential click-based revenue growth, including our social, mobile and global initiatives, while also focusing on growing both our subscription-based products, such as Vacation Rentals and Business Listings, and our transaction-based businesses. We have leveraged our position as the largest online travel company to become an important partner for online advertisers—including hotels, online travel agencies and other travel-related service providers—by providing our customers with access to our large audience of highly-qualified, highly-engaged users. The key drivers of our click-based and display-based advertising revenue are described below, as well as a summary of our key growth areas and the current trends impacting our business.

Key Drivers of Click-Based Advertising Revenue

For the three and six months ended June 30, 2013, 74% and 76%, respectively, of our total revenue came from our CPC-based lead generation product. For the three and six months ended June 30, 2012, 77% and 78%, respectively, of our total revenue came from our CPC-based lead generation product. The key drivers of our click-based advertising revenue include the growth in monthly unique hotel shoppers and revenue per hotel shopper.

- *Hotel shoppers*: Total traffic growth, or growth in monthly visits from unique visitors, is reflective of our overall brand growth. We track and analyze subsegments of traffic and their correlation to revenue generation and utilize hotel shoppers as an indicator of revenue growth. We use the term "hotel shoppers" to refer to users who view a listing of hotels in a city or visitors who view a specific hotel page. Hotel shoppers tend to be seasonal and also tend to vary based on general economic conditions. Our number of hotel shoppers increased 41% for both the three and six month periods ended June 30, 2013, respectively, and increased 30% and 32% for the three and six months ended June 30, 2012, respectively, according to our log files.
- Revenue per hotel shopper: Revenue per hotel shopper is a metric we use to analyze how effectively we are able to monetize hotel shoppers based on a combination of user conversion and pricing. User conversion is a measure of how many hotel shoppers ultimately click on a CPC link that generates revenue for us. User conversion on our site is primarily driven by three factors: merchandising, commerce coverage and choice. We define merchandising as the number and location of ads that are available on a page; we define commerce coverage as whether we have a client who can take an online booking for a particular property; and we define choice as the number of clients available for any given property, allowing the user to shop for the best price. Pricing is the effective CPC that online travel agencies and hoteliers are willing to pay us for a hotel shopper lead. Revenue per hotel shopper decreased 16% and 13% for the three and six months ended June 30, 2013, respectively, and decreased 8% for both the three and six month periods ended June 30, 2012, respectively.

In summary, our CPC revenue depends on the number of hotel shoppers that are interested in a property, whether there is a commerce link available for that hotel shopper to click on for that property, whether there are several commerce choices available for that property so the hotel shopper can shop around and what our customers are willing to pay us for the lead.

Key Drivers of Display-Based Advertising Revenue

For the three and six months ended June 30, 2013, 13% and 12%, respectively, of our total revenue came from our display-based advertising product. The key drivers of our display-based advertising revenue include the growth in number of impressions, or the number of times an ad is displayed on our site, and the cost per thousand impressions, or CPM (or pricing). Our number of impressions increased 32% and 24% for the three and six months ended June 30, 2013, respectively, and increased 7% and 12% for the three and six months ended June 30, 2012, respectively, while pricing decreased 11% and 8% for the three and six months ended June 30, 2013, respectively, and increased 3% and 0% for the three and six months ended June 30, 2012, respectively.

Key Growth Areas

We continue to invest in areas of potential growth, including our social, mobile and global initiatives as well as our subscription-based products, such as Vacation Rentals and Business Listings.

Social. Our Wisdom of Friends initiative is a core component of our strategic growth plan; 76% of respondents to a recent Nielsen study cited "recommendations from people I know" as the information source that they trust most. We believe that having a strong social presence improves engagement on our sites and improves the sites' "stickiness" amongst the users. As a result, we continue to deepen our integration with Facebook. As of June 30, 2013, and according to AppData, an independent application tracking traffic service, TripAdvisor has averaged nearly 38 million monthly Facebook users via its TripAdvisor Facebook application id. We offer these Facebook users a more personalized and social travel planning experience that enables travelers to engage first with their own Facebook friends' reviews and opinions when planning their perfect trip on TripAdvisor.

Mobile. Mobile is an investment area that is geared towards creating a more complete user experience by reinforcing the TripAdvisor brand when users are in-market. In the quarter ended June 30, 2013, we saw strong mobile user uptake, as aggregate downloads of our TripAdvisor, City Guides, and SeatGuru mobile apps doubled year-over-year. Including downloads of Jetsetter and GateGuru, TripAdvisor reached 50 million downloads and average unique monthly visitors via smartphone and tablet devices grew 216% year-over-year from 25.0 million to 79.0 million for the quarter ended June 30, 2013, according to company logs. We believe that travelers will increasingly use mobile devices, including smartphones and tablets, to conduct travel research and planning.

Vacation Rentals. Our Vacation Rentals product addresses a highly-fragmented \$85 billion per-year vacation rental industry, according to a 2010 Radius Global Market study. As of December 31, 2012, we had amassed over 300,000 properties, up more than 50% during the year, across our TripAdvisor Vacation Rentals, U.S.-based FlipKey, and U.K.-based Holiday Lettings, as well as through partnership agreements with other vacation rental businesses. We offer individual property owners and property managers the ability to list using a subscription-based fee structure or a free-to-list, commission-based option and we believe our highly-engaged and motivated user community creates a competitive advantage for us in this market.

Business Listings. Our Business Listings product enables hotel and accommodation owners to list pertinent property information on TripAdvisor, bringing them closer to potential customers and thereby increasing direct bookings. In the year ended December 31, 2012, we grew our Business Listings customer base over 40% to 50,000 subscribers, representing just over 7% of our current hotel and accommodation listings on TripAdvisor branded sites. We continue to expand our sales force and introduce new product features and functionality to grow our subscriber base.

Current Trends Affecting Our Business

Increasing Competition. The travel review industry and, more generally, the business of collecting and aggregating travel-related resources and information, continue to be increasingly competitive. In recent years, an increasing number of companies, such as search companies Google Inc. and Baidu.com, Inc. and several large online travel agencies, have begun to collect and aggregate travel information and resources. We plan to continue to invest in order to remain the leading source of travel reviews as well as continuing to enhance our content and user experience.

Increasing Use of Internet and Social Media to Access Travel Information. Commerce, information and advertising continue to migrate to the Internet and away from traditional media outlets. We believe that this trend will create strategic growth opportunities, allowing us to attract new consumers and develop unique and effective advertising solutions. Consumers are increasingly using online social media, such as Facebook, as a means to communicate and exchange information, including travel information and opinions. We have made significant efforts related to social networking in order to leverage the expanding use of this channel and enhance traffic diversification and user engagement. We are also continually adapting our user experience in response to a changing internet environment and usage trends. For example, in 2012 we invested in building and introducing to users hotel metasearch functionality for our smartphone platforms and in early June 2013 completed the process of implementing hotel metasearch functionality on our desktop and tablet platforms. Refer to our metasearch discussion below in "Our Strategy" in this "Management's Discussion and Analysis of Financial Condition and Results of Operations" section for additional information on our hotel metasearch transition.

Increasing Mobile Usage. Users are increasingly using smartphone and tablet computing devices to access the Internet. To address these growing user demands, we continue to extend our platform to develop smartphone and tablet applications to deliver travel information and resources. Although the substantial majority of our smartphone users also access and engage with our websites on personal computers and tablets where we display advertising, our users could decide to increasingly access our products primarily through smartphone devices. We have just begun to display graphic advertising on smartphones, however, our smartphone monetization strategies are still developing, as smartphone monetization was less than 20% of desktop monetization of hotel shoppers during the six months ended June 30, 2013 while tablets monetize more closely to desktops. Mobile growth and development remains a key company priority and we will continue to invest and innovate in this growing platform to help us maintain and grow our user base and engagement over the long term.

Click-Based Advertising Revenue. In recent years, the majority of our revenue growth resulted from higher click-based advertising revenue due to increased traffic on our websites and an increase in the volume of clicks on our advertisers' placements. Although click-based advertising revenue growth has generally been driven by traffic volume, we remain focused on the various factors that could impact revenue growth, including, but not limited to, the growth in hotel shoppers, CPC pricing fluctuations, the overall economy, the ability of advertisers to monetize our traffic, the quality and mix of traffic to our websites, and the quality and mix of traffic from our advertising placements to advertisers, as well as advertisers' evolving approach to transaction attribution models and return on investment targets. We monitor and regularly respond to changes in these factors in order to strategically improve our user experience, customer satisfaction and monetization in this dynamic environment. For example, in order to improve user experience, we introduced metasearch functionality to our hotel shoppers as discussed below in "—Our Strategy."

Global Economic Conditions. In late 2008 and throughout 2009, weak global economic conditions created uncertainty for travelers and suppliers, and put pressure on discretionary spending on travel and advertising. Since 2010, the travel industry has been gradually improving. However, global economic conditions remain uncertain, and in particular, we anticipate travel expenditures in Europe to continue to be adversely effected by the economic issues overseas.

Our Strategy

In expanding our global reach, we leverage significant investments in technology, operations, brand-building, and advertiser and other partner relationships. For example, we are able to aggregate a large base of consumer reviews, in a variety of languages, across our global core platform. We expect to continue leveraging this investment when launching additional points of sale in new countries, introducing new product features and adding new business model offerings.

- *Investment in Technology*. We believe that our ability to continue to innovate by providing additional functionality to our main Internet sites, while at the same time continuing to extend that functionality to additional platforms such as smartphones and tablets, will enable us to continue providing an industry-leading user experience. We have a strong culture of speed-to-market with our innovations. By innovating and releasing updates quickly, we believe that we can continue to grow our site visitors and over 100 million reviews and opinions, increase revenue and effectively compete with our competitors.
- Social Platform. We intend to continue to expand our social integration and member acquisition efforts with social media, including Facebook, Twitter and other social sharing platforms. We believe that this integration will enable us to continue to grow and maintain engagement with our user base and increase our content. For example, when searching for a hotel on TripAdvisor, users can see which hotels their Facebook friends and friends-of-friends have visited and reviewed. Users can share their reviews and ratings with their friends and publish their activity to their timelines. Users can also update their travel map and rate new experiences on Facebook.
- *Metasearch*. A metasearch display is an interface that shows hotel or flight availability and pricing information from multiple sources, without requiring the user to visit another website. We have offered a flight metasearch product since 2009, expanding internationally to 19 points of sale in 2012 and increasing to 25 points of sale at June 30, 2013. In 2012, we invested in building hotel metasearch functionality for our smartphone platforms and introducing it to 100% of our global smartphone traffic. In early June 2013, we completed the process of fully implementing hotel metasearch functionality onto our desktop and tablet platforms. We believe metasearch functionality provides a better user experience while delivering more valuable leads to advertising partners, however at a lower volume per session. Click-based advertising customers are receiving higher converting leads and therefore are paying higher per-unit prices, which we expect will offset the decreased lead volume. We expect to continue to invest in evolving our metasearch functionality on our site, because we believe that providing hotel and flight pricing and availability deliver a better user experience.
- Investment in Search Engine Marketing. One of the ways that we look to penetrate new markets is to leverage our expertise in search engine marketing, or SEM. SEM is a form of Internet marketing that involves the promotion of websites by increasing their visibility in search engine results pages through the use of paid placement, contextual advertising, and paid inclusion. In certain markets we may bid on keywords to break even or at a loss in order to drive traffic, build our brand, gain more users to our product, collect content and scale more quickly. We think SEM is an important channel because it delivers a significant number of brand impressions and can be a cost-effective method to get people to try our sites.
- Enhance International Offerings. We are focused on strengthening our broad global footprint. We are continuing to improve localization and grow our user base in Europe, Asia and South America, especially in emerging markets, such as Brazil, Russia and China. These and the other newer sites in Asia-Pacific represent a longer-term opportunity for us. We believe that China represents a large international opportunity for our business. We currently have two lead product offerings in the Chinese market—DaoDao and Kuxun—both headquartered in Beijing. We continue to invest heavily and operate at a loss in the Chinese market.
- Acquisitions. We have a history of successfully acquiring and integrating companies that expand our footprint either geographically or in market sectors that are complementary to our flagship properties. We intend to continue to seek acquisition targets.

Segment

We have one reportable segment. The segment is determined based on how our chief operating decision maker manages our business, makes operating decisions and evaluates operating performance.

Employees

As of June 30, 2013, we had 1,799 employees. Of these employees, 1,040 were based in the United States. None of our employees are represented by a labor union or are subject to a collective bargaining agreement. We believe that relations with our employees are good.

Seasonality

Expenditures by travel advertisers tend to be seasonal. Traditionally, our strongest quarter has been the third quarter, which is a key travel research period, with the weakest quarter historically being the fourth quarter. However, adverse economic conditions or continued growth of our international operations with differing holiday peaks may influence the typical trend of our seasonality in the future.

Critical Accounting Policies and Estimates

Critical accounting policies and estimates are those that we believe are important in the preparation of our consolidated financial statements because they require that management use judgment and estimates in applying those policies. We prepare our consolidated financial statements and accompanying notes in accordance with GAAP. Preparation of the consolidated financial statements and accompanying notes requires that management make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities as of the date of the consolidated financial statements as well as revenue and expenses during the periods reported. Management bases its estimates on historical experience, where applicable and other assumptions that it believes are reasonable under the circumstances. Actual results may differ from estimates under different assumptions or conditions.

There are certain critical estimates that we believe require significant judgment in the preparation of our consolidated financial statements. We consider an accounting estimate to be critical if:

- It requires us to make an assumption because information was not available at the time or it included matters that were highly uncertain at the time we were making the estimate; and
- Changes in the estimate or different estimates that we could have selected may have had a material impact on our financial condition or results of
 operations.

There have been no material changes to our critical accounting policies and estimates as compared to the critical accounting policies and estimates described in our Annual Report on Form 10-K for the fiscal year ended December 31, 2012. Refer to "Note 2—Significant Accounting Policies," for additions to our critical accounting policies for the six months ended June 30, 2013.

New Accounting Pronouncements

For a discussion of new accounting pronouncements, see "Note 2—Significant Accounting Policies," in the notes to the consolidated financial statements in this Quarterly Report on Form 10-Q. We are an "issuer" (as defined in Section 2(a) of the Sarbanes-Oxley Act of 2002), and, as such, are required to comply with all new and revised accounting standards applicable to public companies.

Results of Operations Selected Financial Data (in thousands, except per share data)

	Three mont	hs ende	d June 30, 2012	0/ Characa	Six months ended June 30, 2013 2012		0/ Character
		housand		% Change		usands)	% Change
Revenue	\$ 192,615		141,408	36%	\$ 362,023	\$ 273,535	32%
Related-party revenue from Expedia	54,322	_	55,740	(3)%	114,833	107,328	<u>7</u> %
Total revenue	246,937		197,148	25%	476,856	380,863	25%
Costs and expenses:							
Cost of revenue (exclusive of amortization) (1)	4,284		2,926	46%	7,928	5,660	40%
Selling and marketing (2)	82,574		64,243	29%	161,865	131,632	23%
Technology and content (2)	31,903		21,574	48%	60,718	39,415	54%
General and administrative (2)	25,552		18,252	40%	47,985	34,506	39%
Depreciation	6,876		4,715	46%	13,200	8,996	47%
Amortization of intangible assets	1,630	_	1,760	<u>(7)</u> %	2,739	3,599	(24)%
Total costs and expenses	152,819	_	113,470	35%	294,435	223,808	32%
Operating income	94,118		83,678	12%	182,421	157,055	16%
Other income (expense):							
Interest expense, net	(2,066)	(2,405)	(14)%	(4,472)	(5,337)	(16)%
Other, net	(2,150	_	(4,539)	(53)%	(3,611)	(3,843)	(6)%
Total other expense, net	(4,216) _	(6,944)	(39)%	(8,083)	(9,180)	(12)%
Income before income taxes	89,902		76,734	17%	174,338	147,875	18%
Provision for income taxes	(22,914) _	(23,569)	(3)%	(45,051)	(46,539)	(3)%
Net income	66,988		53,165	26%	129,287	101,336	28%
Net (income) loss attributable to noncontrolling interest		_	(146)	(100)%		(206)	(100)%
Net income attributable to TripAdvisor, Inc.	\$ 66,988	\$	53,019	26%	\$129,287	\$ 101,130	28%
Earnings Per Share attributable to TripAdvisor, Inc.							
Basic	\$ 0.47	\$	0.38	24%	\$ 0.90	\$ 0.74	22%
Diluted	\$ 0.46	\$	0.37	24%	\$ 0.89	\$ 0.73	22%
Weighted Average Common Shares Outstanding:							
Basic	143,531		139,278	3%	143,297	136,516	5%
Diluted	145,664		141,737	3%	145,160	138,947	4%
Other Financial Data:							
Adjusted EBITDA (3)	\$ 112,832	\$	96,921	16%	\$ 222,179	\$ 181,110	23%
(1) Excludes amortization as follows:							
Amortization of acquired technology included in amortization of intangibles	\$ 255	\$	182		\$ 411	\$ 364	
Amortization of website development costs included in	4.620		2.070		0.707	F (02)	
depreciation	4,630	_	2,979		8,787	5,692	
(2) Includes stock based componentian as falley as	\$ 4,885	\$	3,161		\$ 9,198	\$ 6,056	
(2) Includes stock-based compensation as follows:	¢ 2244	ተ	022		¢ 4550	¢ 2.001	
Selling and marketing	\$ 2,244		923		\$ 4,559	\$ 2,001	
Technology and content	\$ 4,024		2,426		\$ 10,422	\$ 3,938	
General and administrative	\$ 3,940	\$	3,419		\$ 8,838	\$ 5,521	

⁽³⁾ See "Adjusted EBITDA" below for more information and for a reconciliation of Adjusted EBITDA to net income, the most directly comparable financial measure calculated and presented in accordance with GAAP.

Adjusted EBITDA

To provide investors with additional information regarding our financial results, we have disclosed Adjusted EBITDA in this Quarterly Report on Form 10-Q, a non-GAAP financial measure. We have provided reconciliations below of Adjusted EBITDA to net income, the most directly comparable GAAP financial measure. A "non-GAAP financial measure" refers to a numerical measure of a company's historical or future financial performance, financial position, or cash flows that excludes (or includes) amounts that are included in (or excluded from) the most directly comparable measure calculated and presented in accordance with GAAP in such company's financial statements.

We define Adjusted EBITDA as net income (loss) plus: (1) provision for income taxes; (2) other (income) expense, net; (3) depreciation of property and equipment, including internal use software and website development; (4) amortization of intangible assets; (5) stock-based compensation; and (6) non-recurring expenses. Adjusted EBITDA is the primary metric by which management evaluates the performance of its business and on which internal budgets are based. In particular, the exclusion of certain expenses in calculating Adjusted EBITDA facilitates operating performance comparisons on a period-to-period basis. Adjusted EBITDA eliminates items that are either not part of our core operations, such as non-recurring expenses, or those costs that do not require a cash outlay, such as stock-based compensation. Adjusted EBITDA also excludes depreciation and amortization expense, which is based on our estimates of the useful life of tangible and intangible assets. These estimates could vary from actual performance of the asset, are based on historical costs and other factors and may not be indicative of current or future capital expenditures. We believe that by excluding certain items, such as stock-based compensation and non-recurring expenses, Adjusted EBITDA corresponds more closely to the cash that operating income generated from our business and allows investors to gain an understanding of the factors and trends affecting the ongoing cash earnings capabilities of our business, from which capital investments are made and debt is serviced.

Our use of Adjusted EBITDA has limitations as an analytical tool, and you should not consider it in isolation or as a substitute for analysis of our results reported in accordance with GAAP. Some of these limitations are:

- Adjusted EBITDA does not reflect our cash expenditures or future requirements for capital expenditures or contractual commitments;
- · Adjusted EBITDA does not reflect changes in, or cash requirements for, our working capital needs;
- Adjusted EBITDA does not consider the potentially dilutive impact of stock-based compensation;
- Although depreciation and amortization are non-cash charges, the assets being depreciated and amortized may have to be replaced in the future, and Adjusted EBITDA does not reflect cash capital expenditure requirements for such replacements or for new capital expenditure requirements;
- · Adjusted EBITDA does not reflect tax payments that may represent a reduction in cash available to us; and
- Other companies, including companies in our own industry, may calculate Adjusted EBITDA differently than we do, limiting its usefulness as a
 comparative measure.

Because of these limitations, you should consider Adjusted EBITDA alongside other financial performance measures, including various cash flow metrics, net income and our other GAAP results.

The following table is a reconciliation of Adjusted EBITDA to net income, the most directly comparable financial measure calculated and presented in accordance with GAAP, for the periods presented (in thousands):

	Three mon		Six months ended June 30,		
	2013			2012	
Adjusted EBITDA	\$112,832	\$ 96,921	\$222,179	\$ 181,110	
Depreciation (1)	(6,876)	(4,715)	(13,200)	(8,996)	
Amortization of intangible assets	(1,630)	(1,760)	(2,739)	(3,599)	
Stock-based compensation	(10,208)	(6,768)	(23,819)	(11,460)	
Other expense, net	(4,216)	(6,944)	(8,083)	(9,180)	
Provision for income taxes	(22,914)	(23,569)	(45,051)	(46,539)	
Net income	\$ 66,988	\$ 53,165	\$129,287	\$101,336	

(1) Includes amortization of internal use software and website development costs.

Revenue

We derive substantially all of our revenue through the sale of advertising, primarily through click-based advertising and, to a lesser extent, display-based advertising. In addition, we earn revenue through a combination of subscription-based offerings related to our Business Listings and Vacation Rentals products, transaction revenue from selling room nights on our transactional sites, and other revenue including content licensing.

	Three months ended June 30,			Six mon		
	2013	2012	% Change	2013	e 30, 2012	% Change
	(\$ in n	nillions)		(\$ in n	nillions)	
Click-based advertising	\$ 183	\$ 151	21%	\$ 362	\$ 296	22%
Display-based advertising	31	27	18%	56	48	16%
Subscription, transaction and other	33	19	68%	59	37	60%
Total revenue	\$ 247	\$ 197	25%	\$ 477	\$ 381	25%

Revenue increased \$50 million and \$96 million during the three and six months ended June 30, 2013, respectively, when compared to the same periods in 2012, primarily due to an increase in click-based advertising revenue of \$32 million and \$66 million, respectively. The primary driver of the increase in click-based advertising revenue was an increase in hotel shoppers of 41% for the three and six months June 30, 2013, respectively, partially offset by lower revenue per hotel shopper of 16% and 13% for the three and six months ended June 30, 2013, respectively, primarily due to a combination of lower user conversion related to our transition to hotel metasearch, growth in hotel shoppers on smartphones, which have a lower monetization rate than desktops and tablets, and growth in emerging international markets that are currently monetizing at lower levels than our mature markets. We expect the lower user conversion trend to continue due to our recent transition to hotel metasearch, which we anticipate will negatively impact revenue growth over the balance of the year. Display-based advertising increased by \$4 million and \$8 million during the three and six months ended June 30, 2013, respectively, primarily as a result of a 32% and 24% increase in the number of impressions during the three and six months ended June 30, 2013, respectively. Subscription, transaction and other revenue increased by \$14 million and \$22 million during the three and six month periods ended June 30, 2013, respectively, primarily due to growth in Business Listings.

The following table presents our revenue by geographic region, which reflects how we measure our business internally. Revenue by geography is based on the location of our websites:

		Three months ended June 30,		Six months ended June 30,		
	2013	2012	% Change	2013	2012	% Change
	(\$ in r	nillions)		(\$ in n	nillions)	
North America (1)	\$ 134	\$ 107	25%	\$ 255	\$ 208	23%
EMEA (2)	73	63	16%	145	122	19%
APAC (3)	30	19	55%	57	36	57%
LATAM (4)	10	8	34%	20	15	35%
Total revenue	\$ 247	\$ 197	25%	\$ 477	\$ 381	25%

- (1) United States and Canada*
- (2) Europe, Middle East and Africa
- (3) Asia-Pacific
- (4) Latin America

^{*} Included in international revenue for discussion purposes.

International revenue increased \$25 million and \$52 million during the three and six months ended June 30, 2013, respectively, compared to the same periods in 2012. International revenue represented 49% of total revenue during the three and six months ended June 30, 2013, and represented 49% and 48% of total revenue during the three and six months ended June 30, 2012, respectively. The increase in international revenue, in absolute dollars and as a percentage of total revenue, is primarily due to additional investment in international expansion and growth in international hotel shoppers.

In addition to the product and geographic revenue discussion above, related-party revenue from Expedia, which consists primarily of click-based advertising, is as follows:

	Three months ended		Six months ended			
	June	30,		June 30,		
	2013	2012	% Change	2013	2012	% Change
	(\$ in mil	lions)	· · · · · · · · · · · · · · · · · · ·	(\$ in mi	llions)	
Related party revenue from Expedia	\$ 54	\$ 56	(3)%	\$ 115	\$ 107	7%
% of revenue	22.0%	28.3%		24.1%	28.2%	

Related-party revenue from Expedia decreased \$2 million and increased \$8 million during the three and six month periods ended June 30, 2013, respectively, when compared to the same periods in 2012. Related-party revenue from Expedia increased \$8 million during the six months ended June 30, 2013 when compared to the same period in 2012, primarily due to higher click volume sent to Expedia and higher CPC pricing paid by Expedia during the first three months of 2013, which was partially offset during the three months ended June 30, 2013, primarily due to lower click volume sent to Expedia primarily related to our transition to hotel metasearch. For information on our relationship with Expedia, including recent material transactions, refer to "Note 14—Related Party Transactions" in the notes to our consolidated financial statements.

Cost of Revenue

Cost of revenue consists of expenses that are closely correlated or directly related to revenue generation, including ad serving fees, flight search fees, credit card fees and data center costs.

	Three mor	Three months ended June 30,		Six months ended		
	June			June 30,		
	2013	2012	% Change	2013	2012	% Change
	(\$ in m	illions)	<u></u>	(\$ in mi	illions)	
Cost of revenue	\$ 4	\$ 3	46%	\$ 8	\$ 6	40%
% of revenue	1.7%	1.5%		1.7%	1.5%	

Cost of revenue increased \$1 million and \$2 million during the three and six month periods ended June 30, 2013, respectively, when compared to the same periods in 2012, primarily due to increased data center costs.

Selling and Marketing

Sales and marketing expenses primarily consist of direct costs, including search engine marketing, or SEM, other traffic acquisition costs, syndication costs and affiliate program commissions, brand advertising and public relations. In addition, our indirect sales and marketing expense consists of personnel and overhead expenses, including salaries, commissions, benefits, stock-based compensation expense and bonuses for sales, sales support, customer support and marketing employees.

		Three months ended June 30,		Six mont		
	2013	2012	% Change	2013	2012	% Change
	(\$ in mi	llions)		(\$ in m	illions)	
Direct costs	\$ 52	\$ 42	24%	\$ 103	\$ 89	16%
Personnel and overhead	31	22	38%	59	43	38%
Total selling and marketing	\$ 83	\$ 64	29%	\$ 162	\$ 132	23%
% of revenue	33.4%	32.6%		33.9%	34.6%	

Direct selling and marketing costs increased \$10 million and \$14 million during the three and six month periods ended June 30, 2013, respectively, when compared to the same periods in 2012, primarily due to increased search engine marketing costs and brand advertising costs, partially offset by a decrease in spending in social media costs. Personnel and overhead costs increased \$9 million and \$16 million during the three and six month periods ended June 30, 2013, respectively, when compared to the same periods in 2012, primarily due to an increase in headcount to support business growth, including international expansion, and employees joining us through recent business acquisitions.

Technology and Content

Technology and content expenses consist of personnel and overhead expenses, including salaries and benefits, stock-based compensation expense and bonuses for salaried employees and contractors engaged in the design, development, testing and maintenance of our websites. Other costs include licensing and maintenance expense.

	Three months ended June 30,				hs ended 2 30,		
	2013	2012	% Change	2013	2012	% Change	
	(\$ in mi	llions)		(\$ in m	illions)		
Personnel and overhead	\$ 30	\$ 20	49%	\$ 57	\$ 36	58%	
Other	2	2	29%	4	3	5%	
Total technology and content	\$ 32	\$ 22	48%	\$ 61	\$ 39	54%	
% of revenue	12.9%	10.9%		12.7%	10.3%		

Technology and content costs increased \$10 million and \$22 million during the three and six months ended June 30, 2013, respectively, when compared to the same periods in 2012, primarily due to increased personnel costs from increased headcount to support business growth, including international expansion, enhanced site features, extending our products onto smartphone and tablet platforms, and development of our new hotel metasearch product, as well as an increase in stock based compensation and additional personnel costs related to employees joining us through recent business acquisitions.

General and Administrative

General and administrative expense consists primarily of personnel and related overhead costs, including executive leadership, finance, legal and human resource functions and stock-based compensation as well as professional service fees and other fees including audit, legal, tax and accounting, and other costs including bad debt expense and our charitable foundation costs.

	Three months ended June 30,			Six montl June		
	2013 (\$ in mil	2012 lions)	% Change	2013 (\$ in mi	2012 llions)	% Change
Personnel and overhead	\$ 16	\$ 12	33%	\$ 31	\$ 23	36%
Professional service fees and other	10	6	54%	17	12	45%
Total general and administrative	\$ 26	\$ 18	40%	\$ 48	\$ 35	39%
% of revenue	10.3%	9.3%		10.1%	9.1%	

General and administrative costs increased \$8 million and \$13 million during the three and six months ended June 30, 2013, respectively, when compared to the same periods in 2012, primarily due to increased personnel costs related to an increase in stock-based compensation, as well as increased headcount to support business growth and additional professional service fees in order to support our operations as a standalone public company and an increase in our bad debt provision.

Depreciation

	Three month	ıs ended		Six month	ıs ended	
	June 3	30,		June	30,	
	2013	2012	% Change	2013	2012	% Change
	(\$ in milli	ions)		(\$ in mi	llions)	<u></u>
Depreciation	\$ 7	\$ 5	46%	\$ 13	\$ 9	47%
% of revenue	2.8%	2.4%		2.8%	2.4%	

Depreciation increased \$2 million and \$4 million during the three and six months ended June 30, 2013, respectively, when compared to the same periods in 2012, primarily due to increased amortization related to website development costs.

Amortization of Intangible Assets

	Three months ended June 30,			Six months ended June 30,		
	2013	2012	% Change	2013	2012	% Change
	(\$ in millions)			(\$ in millions)		
Amortization of intangible assets	\$ 2	\$ 2	(7)%	\$ 3	\$ 4	(24)%
% of revenue	0.7%	0.9%		0.6%	1.0%	

The change in intangible asset amortization is not material for the periods presented. Refer to "Note 7—*Goodwill and Intangible Assets, Net*" for estimated intangible amortization for the year ending December 31, 2013.

Interest (Expense) Income, Net

Interest expense is primarily related to interest incurred on our Term Loan and credit facilities for the periods presented.

Other, Net

Other, net is primarily comprised of net foreign exchange gains and losses for the periods presented.

Provision for Income Taxes

	Three mont	Three months ended June 30,		Six months ended		
	June			June 30,		
	2013	2012	% Change	2013	2012	% Change
	(\$ in mil	(\$ in millions)		(\$ in millions)		
Provision for income taxes	\$ 23	\$ 24	(3)%	\$ 45	\$ 47	(3)%
Effective tax rate	25.5%	30.7%		25.8%	31.5%	

For the three and six months ended June 30, 2013, our effective tax rate is less than the federal statutory tax rate primarily due to earnings in jurisdictions outside the United States, where our effective tax rate is lower, which was partially offset by state income taxes, non-deductible stock compensation and accruals on uncertain tax positions. The change in the effective tax rate for 2013 compared to the 2012 rate was primarily due to an increase in earnings in jurisdictions outside the United States as well as the internal restructuring that was implemented in the last quarter of 2012.

Related Party Transactions

For information on our relationships with Expedia and Liberty Interactive Corporation, including recent material transactions, refer to "Note 14—Related Party Transactions" in the notes to our consolidated financial statements.

Stock-Based Compensation

Refer to "Note 3—*Stock Based Awards and Other Equity Instruments*" for further information on current year equity award activity, including the issuance of 1,561,789 primarily service based stock options with a weighted average grant-date fair value per option of \$23.15 and 994,293 RSUs with a weighted average grant-date fair value of \$47.05 during the six months ended June 30, 2013.

Financial Position, Liquidity and Capital Resources

Our principal sources of liquidity are cash flows generated from operations. At June 30, 2013 and December 31, 2012, we had \$616 million and \$586 million, respectively, of cash, cash equivalents and short and long-term available-for-sale marketable securities. As of June 30, 2013, approximately \$346 million of our cash, cash equivalents, and short and long-term marketable securities were held by our international subsidiaries, primarily in the United Kingdom, and are related to earnings we intend to reinvest permanently outside the United States. Should we distribute earnings of foreign subsidiaries in the form of dividends or otherwise, we may be subject to U.S. income taxes. Cumulative undistributed earnings of foreign subsidiaries that we intend to indefinitely reinvest outside of the United States totaled approximately \$434 million as of June 30, 2013. Should we distribute, or be treated under certain U.S. tax rules as having distributed, the earnings of foreign subsidiaries in the form of dividends or otherwise, we may be subject to U.S. income taxes. Determination of the amount of any unrecognized deferred income tax liability on this temporary difference is not practicable because of the complexities of the hypothetical calculation. Cash held is primarily denominated in U.S. dollars.

Historically, the cash we generate has been sufficient to fund our working capital requirements, capital expenditures and to meet our long-term debt obligations and commitments. Management believes that our cash and cash equivalents and available-for-sale marketable securities, combined with expected cash flows generated by operating activities and available cash from our credit facilities will be sufficient to fund our ongoing working capital requirements, capital expenditures, business growth initiatives, meet our long term debt obligations and commitments, share repurchases and fund any potential acquisitions for at least the next 12 months. However, if during that period or thereafter, we are not successful in generating sufficient cash flow from operations or in raising additional capital when required in sufficient amounts and on terms acceptable to us, we may be required to reduce our planned capital expenditures and scale back the scope of our business growth initiatives, either of which could have a material adverse effect on our future financial condition or results of operations.

Term Loan Facility Due 2016 and Revolving Credit Facility

Overview

On December 20, 2011, in connection with the Spin-Off, we entered into a credit agreement, dated as of December 20, 2011, by and among TripAdvisor, TripAdvisor Holdings, LLC, and TripAdvisor LLC, the lenders party thereto, JPMorgan Chase Bank, N.A., as administrative agent, and J.P. Morgan Europe Limited, as London agent (this credit agreement, together with all exhibits, schedules, annexes, certificates, assignments and related documents contemplated thereby, is referred to herein as the "Credit Agreement"), which provides \$600 million of borrowing including:

- the Term Loan Facility, or Term Loan, in an aggregate principal amount of \$400 million with a term of five years due December 2016; and
- the Revolving Credit Facility in an aggregate principal amount of \$200 million available in U.S. dollars, Euros and British pound sterling with a term of five years expiring December 2016.

The Term Loan and any loans under the Revolving Credit Facility bear interest by reference to a base rate or a Eurocurrency rate, in either case plus an applicable margin based on our leverage ratio. We are also required to pay a quarterly commitment fee on the average daily unused portion of the Revolving Credit Facility for each fiscal quarter and fees in connection with the issuance of letters of credit. The Term Loan and loans under the Revolving Credit Facility currently bear interest at LIBOR plus 175 basis points, or the Eurocurrency Spread, or the alternate base rate ("ABR") plus 75 basis points, and undrawn amounts are currently subject to a commitment fee of 30 basis points.

As of June 30, 2013, we are using a one-month interest period Eurocurrency spread, which is approximately 2.0% per annum. Interest is currently payable on a monthly basis while we are borrowing under the one-month interest rate period. The current interest rates are based on current assumptions, leverage and LIBOR rates and do not take into account that rates will reset periodically. A 25 basis point change in the interest rate on the current Term Loan balance would result in an increase or decrease to interest expense of approximately \$0.9 million per annum.

The Term Loan principal is currently repayable in quarterly installments on the last day of each calendar quarter equal to 2.5% of the original principal amount with the balance due on the final maturity date. Principal payments aggregating \$20 million were made during the six months ended June 30, 2013.

The Revolving Credit Facility includes \$40 million of borrowing capacity available for letters of credit and \$40 million for borrowings on same-day notice. As of June 30, 2013, there are no outstanding borrowings under our Revolving Credit Facility.

During the three and six months ended June 30, 2013, we recorded total interest and commitment fees on our Credit Agreement of \$2.0 million and \$4.1 million, respectively, to interest expense on our consolidated statement of operations. During the three and six months ended June 30, 2012, we recorded total interest and commitment fees on our Credit Agreement of \$2.1 million and \$4.4 million, respectively, to interest expense on our consolidated statements of operations. All unpaid interest and commitment fee amounts as of June 30, 2013 and December 31, 2012 were not material.

The remaining future minimum principal payment obligations due under the Credit Agreement related to our Term Loan is as follows as of June 30, 2013 (in millions):

		Payment Amount	
2013 (remaining six months)	\$	20	
2014	\$	40	
2015	\$	40	
2016	<u>\$</u>	260	
Total	<u>\$</u>	360	

Covenants

The Credit Agreement contains a number of covenants that, among other things, restrict our ability to: incur additional indebtedness, create liens, enter into sale and leaseback transactions, engage in mergers or consolidations, sell or transfer assets, pay dividends and distributions, make investments, loans or advances, prepay certain subordinated indebtedness, make certain acquisitions, engage in certain transactions with affiliates, amend material agreements governing certain subordinated indebtedness, and change our fiscal year. The Credit Agreement also requires us to maintain a maximum leverage ratio and a minimum cash interest coverage ratio, and contain certain customary affirmative covenants and events of default, including a change of control. If an event of default occurs, the lenders under the Credit Agreement will be entitled to take various actions, including the acceleration of all amounts due under Credit Agreement and all actions permitted to be taken by a secured creditor.

As of June 30, 2013 we believe we are in compliance with all of our debt covenants.

Refer to "Note 5—*Debt*" in the notes to the consolidated financial statements for additional information on our Credit Agreement. The full text of the Credit Agreement is incorporated by reference on our Annual Report on Form 10-K for the year ended December 31, 2012, as Exhibit 4.2.

Chinese Credit Facilities

In addition to our borrowings under the Credit Agreement, we maintain our Chinese Credit Facilities. As of June 30, 2013 and December 31, 2012, we had \$24.6 million and \$32.1 million of short term borrowings outstanding, respectively.

Certain of our Chinese subsidiaries entered into a RMB 138,600,000 (approximately \$22 million), one-year revolving credit facility with Bank of America (the "Chinese Credit Facility—BOA") that is currently subject to review on a periodic basis with no-specific expiration period. During the third quarter of 2012 this credit line was increased to RMB 189,000,000 (approximately \$30 million). During the six months ended June 30, 2013, we made a payment inclusive of interest of RMB 68,283,570 (approximately \$10.9 million). We currently have \$11.3 million of outstanding borrowings from this credit facility as of June 30, 2013. Our Chinese Credit Facility—BOA currently bears interest based at 100% of the People's Bank of China's base rate and was 5.6% as of June 30, 2013.

In addition, during April 2012, certain of our Chinese subsidiaries entered into a RMB 125,000,000 (approximately \$20 million) one-year revolving credit facility with J.P. Morgan Chase Bank ("Chinese Credit Facility-JPM"). This credit facility was renewed for an additional year in April 2013. During the six months ended June 30, 2013, we made a payment inclusive of interest of RMB 24,281,546 (approximately \$3.9 million). We currently have \$13.3 million of outstanding borrowings from this credit facility as of June 30, 2013. Our Chinese Credit Facility—JPM currently bears interest based at 100% of the People's Bank of China's base rate and was 5.6% as of June 30, 2013.

Contractual Obligations, Commercial Commitments and Off-Balance Sheet Arrangements

Office Lease Commitments

On June 20, 2013, TripAdvisor LLC ("TA LLC"), our indirect, wholly owned subsidiary, entered into a lease (the "Lease"), with Normandy Gap-V Needham Building 3, LLC, as landlord (the "Landlord"). Pursuant to the Lease, the Landlord will build an approximately 280,000 square foot rental building in Needham, Massachusetts (the "Premises"), and thereafter lease the Premises to TA LLC as TripAdvisor's new corporate headquarters for an initial term of 15 years and 7 months. If the Landlord fails to deliver the Premises according to the schedule, subject to certain conditions, TA LLC may be entitled to additional free rent, or in extreme cases, a right to terminate the Lease. Under the Lease, TA LLC is required to pay an initial base rent of \$33.00 per square foot per year, increasing to \$34.50 per square foot by the final year of the initial term, as well as all real estate taxes and other building operating costs.

The aggregate future minimum lease payments are \$143.5 million and are currently scheduled to be paid, beginning in November 2015, as follows: \$1.1 million for 2015, \$9.3 million for 2016, \$9.3 million for 2017, and \$123.9 million for 2018 and thereafter. The Lease has escalating rental payments and initial periods of free rent and therefore the effects of variable rent payments will be expensed on a straight-line basis over the life of the Lease in accordance with GAAP. All real estate taxes and other building operating costs paid during the term of the Lease will be expensed as incurred. TA LLC is also obligated to deliver a letter of credit to the Landlord in the amount of \$0.8 million as security deposit, which amount is subject to increase under certain circumstances. Subject to certain conditions, TA LLC has certain rights under the Lease, including rights of first offer to lease additional space or to purchase the Premises if the Landlord elects to sell. TA LLC also has an option to extend the term of the Lease for two consecutive terms of five years each.

In connection with the Lease, on June 20, 2013, TripAdvisor entered into a Guaranty (the "Guaranty"), pursuant to which TripAdvisor provides full payment and performance guaranty for all of TA LLC's obligations under the Lease.

The full text of the Lease and the Guaranty are filed with this Quarterly Report on Form 10-Q as Exhibit 10.1 and 10.2, respectively.

The following table presents our estimated future minimum rental payments under all operating leases with non-cancelable lease terms having initial terms in excess of one year as of June 30, 2013 (in millions):

		otal ments
2013 (remaining six months)	\$	5
2014		8
2015		6
2016		11
2017		11
2018 and thereafter	_	130
Total (1)	\$	171

(1) Excludes spending on anticipated leasehold improvements on our Needham, Massachusetts lease, including design, development, construction costs, and the purchase and installation of equipment, and the receipt of related Landlord incentives, which we estimate will begin in October 2014.

There have been no other material changes outside the normal course of business to our contractual obligations and commercial commitments since December 31, 2012. Refer to "Contractual Obligations, Commercial Commitments and Off-Balance Sheet Arrangements" in Item 7—Management's Discussion and Analysis of Financial Condition and Results of Operations of our Annual Report on Form 10-K for the year ended December 31, 2012.

Off-Balance Sheet Arrangements

As of June 30, 2013, we did not have any off-balance sheet arrangements that have, or are reasonably likely to have, a current or future effect on our financial condition, results of operations, liquidity, capital expenditures or capital resources.

Contingencies

In the ordinary course of business, we and our subsidiaries are parties to legal proceedings and claims involving alleged infringement of third-party intellectual property rights, defamation, and other claims. Rules of the SEC require the description of material pending legal proceedings, other than ordinary, routine litigation incident to the registrant's business, and advise that proceedings ordinarily need not be described if they primarily involve damages claims for amounts (exclusive of interest and costs) not individually exceeding 10% of the current assets of the registrant and its subsidiaries on a consolidated basis. In the judgment of management, none of the pending litigation matters that the Company and its subsidiaries are defending involves or is likely to involve amounts of that magnitude. There may be claims or actions pending or threatened against us of which we are currently not aware and the ultimate disposition of which could have a material adverse effect on us.

Consolidated Cash Flow Statements Discussion:

Our cash flows are as follows (in millions):

	Six months ended			
	June	June 30,		
	2013	2012		
Net cash provided by (used in):				
Operating activities	\$ 133	\$ 91		
Investing activities	\$ (256)	\$ (6)		
Financing activities	\$ (43)	\$ 212		

Operating Activities

For the six months ended June 30, 2013, net cash provided by operating activities increased by \$42 million or 46% when compared to the same period in 2012, primarily due to an increase in net income of \$28 million, an increase in non-cash items not affecting cash flows of \$13 million, which is primarily related to increased stock based compensation and depreciation. Working capital movements were essentially flat for the six months ending June 30, 2013 and 2012, which reflected an increase in deferred merchant payables in 2013 of \$15.5 million offset by a decrease in operating cash related to the timing of customer receipts, related party cash receipts and tax and vendor payments.

Investing Activities

For the six months ended June 30, 2013, net cash provided by investing activities decreased by \$250 million when compared to the same period in 2012, primarily due to a net increase in marketable securities of \$202 million, cash paid for acquisition of businesses of \$32 million, net of cash acquired, and an increase in capital expenditures of \$10 million. In addition, we received \$7 million during the three months ended March 31, 2012 from Expedia related to the Spin-Off, which did not reoccur in 2013.

Financing Activities

For the six months ended June 30, 2013, net cash provided by financing activities decreased by \$255 million when compared to the same period in 2012 primarily due to an increase of \$10 million in principal payments on our term loan, payments of \$34 million for common stock share repurchases under our authorized share repurchase program, a reduction of \$206 million in proceeds related to the exercise of our stock options and warrants, as all remaining warrants were settled in the first six months of 2012, and a \$15 million repayment of our outstanding borrowings on our Chinese Credit Facilities in 2013. This was offset a \$10 million repayment of our outstanding borrowing on our Revolving Credit Facility in 2012 that did not reoccur in 2013.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Market Risk Management

We are exposed to certain market risks, including changes in interest rates and foreign currency exchange rates that could adversely affect our results of operations or financial condition. We manage our exposure to these risks through established policies and procedures and by assessing the anticipated near-term and long-term fluctuations in interest rates and foreign currency exchange rates. Our objective is to mitigate potential income statement, cash flow and market exposures from changes in interest and foreign exchange rates.

There has been no material change in our market risk profile during the six months ended June 30, 2013. For additional information, see Item 7A, Quantitative and Qualitative Disclosures About Market Risk, in Part II of our Annual Report on Form 10-K for the year ended December 31, 2012.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

As of June 30, 2013, our management, with the participation of our President and Chief Executive Officer and our Chief Financial Officer, evaluated the effectiveness of our disclosure controls and procedures pursuant to Rule 13a-15(b) promulgated under the Securities Exchange Act of 1934, as amended, or the Exchange Act. Based upon that evaluation, our President and Chief Executive Officer and our Chief Financial Officer concluded that, as of June 30, 2013, our disclosure controls and procedures were effective in ensuring that material information required to be disclosed in the reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's, or the SEC's, rules and forms, including ensuring that such material information is accumulated and communicated to our management, including our President and Chief Executive Officer and our Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosure.

Changes in internal control over financial reporting.

There were no changes to our internal control over financial reporting that occurred during the quarter ended June 30, 2013 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II—OTHER INFORMATION

Item 1. Legal Proceedings

In the ordinary course of business, we and our subsidiaries are parties to legal proceedings and claims involving alleged infringement of third-party intellectual property rights, defamation, and other claims. Rules of the SEC require the description of material pending legal proceedings, other than ordinary, routine litigation incident to the registrant's business, and advise that proceedings ordinarily need not be described if they primarily involve damages claims for amounts (exclusive of interest and costs) not individually exceeding 10% of the current assets of the registrant and its subsidiaries on a consolidated basis. In the judgment of management, none of the pending litigation matters that TripAdvisor and our subsidiaries are defending involves or is likely to involve amounts of that magnitude. There may be claims or actions pending or threatened against us of which we are currently not aware and the ultimate disposition of which could have a material adverse effect on us.

Item 1A. Risk Factors

In addition to the other information set forth in this Quarterly Report on Form 10-Q, you should carefully consider the factors discussed in Part I—Item 1A., "Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2012 which could materially affect our business, financial condition or future results. During the quarter ended June 30, 2013, there have been no material changes in our risk factors from those disclosed in Part 1, Item 1A., "Risk Factors" in our Annual Report on Form 10-K for the fiscal year ended December 31, 2012.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Unregistered Sales of Equity Securities

During the quarter ended June 30, 2013, we did not issue or sell any shares of our common stock, Class B common stock or other equity securities pursuant to unregistered transactions in reliance upon an exemption from the registration requirements of the Securities Act of 1933, as amended.

Share Repurchases

During the three months ended June 30, 2013, we repurchased 675,000 shares of outstanding common stock under the share repurchase program, as set forth in the table below:

	Total Number of Shares	Average Price Paid per	Total Number of Shares Purchased as Part of Publicly Announced Plans or	Maximum Number (or Approximate Dollar Value) of Shares that May Yet Be Purchased Under the Plans or
Period	Purchased	Share (1)	Programs (2)	Programs
April 1 to April 30	_	_	_	\$250,000,000
May 1 to May 31				\$250,000,000
June 1 to June 30	675,000	\$ 62.82	675,000	\$207,593,148
Total	675,000		675,000	

- (1) These amounts include fees and commissions associated with the share repurchase.
- (2) In February 2013, we announced that our Board of Directors authorized the repurchase of \$250 million of our shares of common stock under a share repurchase program. We intend to use available cash and future cash from operations to fund repurchases under the share repurchase program. The repurchase program has no expiration but may be suspended or terminated by the Board of Directors at any time. The Executive Committee of our Board of Directors will determine the price, timing, amount and method of such repurchases based on its evaluation of market conditions and other factors, and any shares repurchased will be in compliance with applicable legal requirements, at prices determined to be attractive and in the best interests of both the company and its stockholders.

Item 6. Exhibits

The exhibits listed below are filed as part of this Quarterly Report on Form 10-Q.

Exhibit No.	Exhibit Description	Filed <u>Herewith</u>	<u>Form</u>	Incorporated SEC File No.	by Reference Exhibit	Filing Date
4.1‡	First Amendment to TripAdvisor, Inc. 2011 Stock and Annual Incentive Plan	X			·	
4.2‡	TripAdvisor, Inc. 2011 Stock and Annual Incentive Plan		S-8	333-178637	4.5	12/20/11
10.1	Lease between TripAdvisor LLC and Normandy GAP-V Needham Building 3, LLC, dated as of June 20, 2013 $$	X				
10.2	Guaranty by TripAdvisor, Inc., dated as of June 20, 2013	X				
31.1	Certification of the Chief Executive Officer pursuant Section 302 of the Sarbanes-Oxley Act of 2002	X				
31.2	Certification of the Chief Financial Officer pursuant Section 302 of the Sarbanes-Oxley Act of 2002	X				
32.1**	Certification of the Chief Executive Officer pursuant Section 906 of the Sarbanes-Oxley Act of 2002	X				
32.2**	Certification of the Chief Financial Officer pursuant Section 906 of the Sarbanes-Oxley Act of 2002	X				
101*	The following financial statements from the Company's Quarterly Report on Form 10-Q for the period ended June 30, 2013, formatted in XBRL: (i) Consolidated Statements of Operations (Unaudited), (ii) Consolidated Statements of Comprehensive Income (Unaudited), (iii) Consolidated Balance Sheets (Unaudited), (iv) Consolidated Statement of Changes in Stockholders' Equity (Unaudited), (v) Consolidated Statements of Cash Flows (Unaudited), and (vi) Notes to Unaudited Consolidated Financial Statements.					

^{*} XBRL information is furnished and not filed for purposes of Sections 11 and 12 of the Securities Act of 1933 and Section 18 of the Securities Exchange Act of 1934, and is not subject to liability under those sections, is not part of any registration statement or prospectus to which it relates and is not incorporated or deemed to be incorporated by reference into any registration statement, prospectus or other document.

^{**} Furnished herewith.

[‡] Indicates a management contract or a compensatory plan, contract or arrangement.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

	Julie M.B. Bradley Chief Financial Officer		
By:	/S/	JULIE M.B. BRADLEY	
TripA	dvisor, Inc.		

July 24, 2013

FIRST AMENDMENT TO TRIPADVISOR, INC. 2011 STOCK AND ANNUAL INCENTIVE PLAN

- A. The TripAdvisor, Inc. 2011 Stock and Annual Incentive Plan (the "Plan") is hereby amended as follows:
 - 1. Section 3(a) of the Plan is hereby amended by deleting the first sentence thereof and substituting therefor the following:

 "The maximum number of Shares that may be delivered pursuant to Awards under the Plan shall be the sum of (i) the number of Shares that may be issuable upon exercise or vesting of the Adjusted Awards and (ii) 25,000,000."
 - 2. Section 3(b)(i) of the Plan is hereby amended by deleting such Section in its entirety and substituting therefor the following:
 - "(i) Options and/or Stock Appreciation Rights covering in excess of 3,000,000 Shares."
- B. Except as otherwise amended, the Plan is hereby confirmed in all other respects.
- C. This First Amendment shall become effective upon the approval of the Company's stockholders at the Company's 2013 Annual Meeting,

Exhibit 10.1

EXECUTION COPY

CORPORATE HEADQUARTERS LEASE

FOR

TRIPADVISOR LLC

CENTER 128

NEEDHAM, MASSACHUSETTS

TRIPADVISOR LEASE

CENTER 128 NEEDHAM, MASSACHUSETTS

TABLE OF CONTENTS

ADTICLE 1 DACIC TEDMS AND EVHIDITS	1
ARTICLE 1. BASIC TERMS AND EXHIBITS	1
1. Defined Terms	1
2. Exhibits	10
ARTICLE 2. PREMISES AND APPURTENANT RIGHTS 2.1. Lease of Premises	
2.1. Lease of Premises	12
2.2. Appurtenant Rights	12
2.3. Reservations	20
2.4. Project Development Documents	20
2.5. Landlord's Interest in the Building	21
ARTICLE 3. LEASE TERM	24
3.1 Lease Term	24
3.2 Building Construction and Delivery	24
3.3 Extension Options	24
ARTICLE 4. RENT	24
4.1 Base Rent	24
4.2 Additional Rent	25
4.3 Method of Payment	27
4.4 Audit Rights	28
4.5 Cooperation With Tenant's Accounting Requirements	29
ARTICLE 5. TAXES	30
5.1 Taxes	30
5.2 Definition of "Taxes"	30
5.3 Personal Property Taxes	32
5.4 TIF Incentives	32
ARTICLE 6. UTILITIES	32
6.1 Utilities	32
ARTICLE 7. INSURANCE	35
7.1 Tenant's Coverage	35

	7.2 Action Increasing Rates	36
	7.3 Waiver of Subrogation	36
	7.4 Landlord's Insurance and Indemnity	37
ART	ICLE 8. OPERATING EXPENSES	37
	8.1 Operating Expenses	37
ART	ICLE 9. USE OF PREMISES	38
	9.1 Permitted Uses	38
	9.2 Indemnification	38
	9.3 Compliance With Legal Requirements	39
	9.4 Hazardous Substances	40
	9.5 Signage	42
	9.6 Landlord's Access	42
	9.7 Security	43
ART	ICLE 10. CONDITION AND MAINTENANCE OF PREMISES AND PROPERTY	44
	10.1 Condition of Premises and Property	44
	10.2 Tenant Property	44
	10.3 Landlord's Obligations for Building Services, Repair and Maintenance	45
	10.4 Tenant's Obligations for Repair and Maintenance	50
	10.2. Tenant Work	52
	10.6 Condition upon Termination	57
ART	ICLE 11. ROOFTOP EQUIPMENT	57
	11.1 Rooftop Installations	57
	11.2 Installation and Maintenance of Rooftop Equipment	58
	11.3 Operation of Rooftop Equipment	58
ART	ICLE 12. DAMAGE OR DESTRUCTION; CONDEMNATION	59
	12.1 Damage or Destruction of Premises	59
	12.2 Eminent Domain	61
ART	ICLE 13. ASSIGNMENT AND SUBLETTING	62
	13.1 Landlord's Consent Required	62
	13.2 Permitted Transfers	62

13.3 Consent Procedures

63

	13.4 Sharing of Transfer Profits	64
	13.5 No Release	64
	13.6 Additional Provisions	65
ART	TICLE 14. EVENTS OF DEFAULT AND REMEDIES	65
	14.1 Events of Default	65
	14.2 Right to Terminate	66
	14.3 Remedies for Default	67
	14.4. Late Charge	69
	14.5. Interest	69
	14.6. Letter of Credit	70
	14.7. Guaranty	72
	14.8. Holdover	72
ART	CICLE 15. PROTECTION OF LENDERS	73
	15.1 Subordination and Superiority of Lease	73
	15.2 Attornment	73
	15.3 Rent Assignment	74
	15.4 Other Instruments	74
	15.5 Estoppel Certificates	75
ART	TICLE 16. MISCELLANEOUS PROVISIONS	75
	16.1 Landlord's and Tenant's Consent	75
	16.2 Notice of Landlord's Default	75
	16.3 Quiet Enjoyment	75
	16.4. Limitations on Liability	76
	16.5. Notices	76
	16.6. Notice of Lease	76
	16.7. Corporate Authority	77
	16.8. Joint and Several Liability	77
	16.9. Force Majeure	77
	16.10. Limitation of Warranties	77
	16.11. Brokers	78
	16.12. Applicable Law and Construction	78
	16.13. Construction on the Property or Adjacent Property	79

80

16.14. Confidentiality of Information

TRIPADVISOR LEASE

CENTER 128 NEEDHAM, MASSACHUSETTS

This Lease (the "Lease") is hereby entered into as of the Lease Date set forth below by and between the undersigned Landlord and Tenant.

ARTICLE 1. BASIC TERMS AND EXHIBITS

1. <u>Defined Terms</u>. The following capitalized terms used in this Lease shall have the meanings set forth below. A reference table for all defined terms used in this Lease is attached hereto as <u>Exhibit O</u>. All references in this Lease to Sections shall mean the Sections of this Lease, unless otherwise expressly specified.

1.1 <u>Lease Date and Parties:</u>

Lease Date: June 20, 2013

Landlord: Normandy GAP-V Needham Building 3, LLC, a Delaware limited liability company

Tenant: TripAdvisor LLC, a Delaware limited liability company

Guarantor: TripAdvisor, Inc., a Delaware corporation

1.2 <u>Terms related to the Building:</u>

Building: The six-(6)-story building, with a mechanical penthouse level, to be constructed by Landlord under

<u>Section 3.2</u> and <u>Exhibit B</u> on the Building Site (as defined below). For clarity in certain contexts, the Building is also sometimes referred to herein as the "<u>TA Building</u>" or the "<u>Original Building</u>."

Building Site: The portion of the Project Land on which the Building is located together with the immediately

surrounding exterior areas on the Building Site that exclusively serve the Building, as shown on the

Building Site Plan and more particularly described in this Lease.

Building Site Plan: The detailed site plan for the Building Site attached hereto as Exhibit A-2, subject to the final design

process under Exhibit B.

Preliminary Plans: The preliminary plans, specifications, and renderings attached hereto as Exhibit B-1.

Base Building Work: The base building work and initial site work for the Building to be performed by Landlord under

Exhibit B, as more particularly defined in the definition of "Base Building Work" in Exhibit B.

TI Work: The Tenant Work (as defined in <u>Section 10.5</u>) for the Building to be performed by Tenant under <u>Exhibit B</u>,

pursuant to the final TIW Construction Documents (as defined in **Exhibit B**).

1.3. <u>Terms related to the Project:</u>

Project: The phased development project known as Center 128 in Needham, Massachusetts to be developed on the

Project Land pursuant to the Project Condominium Documents (as defined below). The Project, when fully developed, is expected to contain (a) a total of approximately 740,000 gross square feet in four (4) new buildings (inclusive of the TA Building), (b) a hotel containing approximately 128 guest rooms and related hotel facilities and amenities (currently under construction), (c) two (2) multi-level parking garages adjacent to the Project buildings (shown as Garage A and Garage B on the attached Project Site Plan), as well as surface parking, and (d) open space, landscaping, and other amenities and improvements now or hereafter required or permitted by the applicable Master Plan Permits (as defined in the Project

Condominium Documents).

Project Land: The land described in Exhibit A.

Special Permit Amendment: The Special Permit Amendment referenced in Paragraph 4 of Exhibit B.

Project Site Plan: The general site plans for the Project attached hereto as Exhibit A-1, as the same may be revised from time

to time (a) pursuant to the final design process and the Special Permit Amendment and

(b) pursuant to the remaining Project development rights reserved to the Project Declarant (as defined below) under the Project Condominium Documents, as the same may be affected by the exercise (if applicable) of Tenant's Expansion/Purchase Rights as more particularly provided in this Lease.

The common areas and facilities of the Project from time to time that are for the use of the Building in common with all or certain other Project buildings pursuant to the Project Condominium Documents, including without limitation the roadways, sidewalks, common stairways, common driveways and nonexclusive parking areas and non-exclusive garages and other improvements on the Condominium Land, as more particularly provided for under the Project Condominium Documents.

As defined in the Project Condominium Documents. The Limited Common Elements for the Building are set forth in the Building Condominium Instruments under Section 2.4(a) and referenced in the applicable

provisions of this Lease.

Project Condominium Documents: As defined in Exhibit A.

Project Declarant: The Declarant as defined under the Project Condominium Documents.

1.4 Terms related to the Premises:

Project Common Facilities:

Limited Common Elements:

Premises: The entire rentable area of the Building, subject to the exclusions set forth in Section 2.1.

Premises RSF: The entire RSF of the Building, comprising a total of 280,892 square feet of Premises RSF as set forth in

Paragraph 3.5(a) of Exhibit B, subject to adjustment (if any) in accordance with Paragraph 3.5(b) of

Exhibit B.

RSF: The rentable square footage based on the BOMA International Standard Method of Measurement for

Office Buildings ANSI/BOMA Z65.1-1996 for single tenant buildings (the "Measurement Standard"), as

set forth in Paragraph 3.5(a) of Exhibit B and subject to adjustment (if any) in accordance with

Paragraph 3.5(b) of Exhibit B.

Tenant's Pro Rata Share: The percentage determined by dividing (x) the Premises RSF by (y) the entire RSF of the Building.

Accordingly, Tenant's Pro Rata Share for the original Premises under this Lease shall be one hundred

percent (100%).

1.5 <u>Terms related to the Lease Term:</u>

Term: The Initial Term, as the same may be from time to time extended.

Initial Term: The period commencing on the Commencement Date (as defined below) and expiring on the Term

Expiration Date (as defined below).

TIW Access Date: The date on which the Base Building Work shall have achieved the conditions for Building

Enclosed/Weather Tight as set forth in Exhibit B-3, as provided in Paragraph 6.3 of Exhibit B.

Early Occupancy Period: See <u>Section 4.2(a)</u>.

Commencement Date: The date that is eight (8) months after the TIW Access Date.

Partial Rent Commencement Date: The date that is four (4) months after the Commencement Date. See <u>Section 4.1(a)</u>.

Rent Commencement Date: The date that is forty nine (49) days after the Partial Rent Commencement Date, subject to adjustment (if

any) in accordance with Paragraph 3.5(c) of Exhibit B. See Section 4.1(a).

Term Expiration Date: The last day of the Partial Final Lease Year (as defined below).

Lease Year: Each successive twelve-(12)-month period commencing on the Commencement Date and on each

anniversary thereof, provided that the "<u>Partial Final Lease Year</u>" shall mean the seven-(7)-month period immediately following the end of fifteenth (15th) Lease Year and (if such seven-(7)-month period does not end on the last day of a calendar month) shall further include the balance of the calendar month in which

such seven-(7)-month period ends.

Extension Terms:

Two (2) extension periods of five (5) years each, as set forth in, and subject to the provisions of, <u>Section 3.3</u> and <u>Exhibit D-1</u>.

1.6 <u>Base Rent (per square foot of Premises RSF per annum):</u>

For the Initial Term: Lease Year 1: \$33.00 per RSF per annum, subject to abatement as set forth in <u>Section 4.1(a)</u>.

\$33.00 per RSF per annum Lease Year 2: Lease Year 3: \$33.00 per RSF per annum Lease Year 4: \$33.00 per RSF per annum Lease Year 5: \$33.00 per RSF per annum Lease Year 6: \$33.75 per RSF per annum Lease Year 7: \$33.75 per RSF per annum \$33.75 per RSF per annum Lease Year 8: Lease Year 9: \$33.75 per RSF per annum Lease Year 10: \$33.75 per RSF per annum Lease Year 11: \$34.50 per RSF per annum Lease Year 12: \$34.50 per RSF per annum \$34.50 per RSF per annum Lease Year 13: Lease Year 14: \$34.50 per RSF per annum Lease Year 15: \$34.50 per RSF per annum Partial Final \$34.50 per RSF per annum

Lease Year:

For each Extension Term, if applicable:

Ninety five percent (95%) of the Fair Market Rent Rate for the Premises for the applicable Extension Term, as determined pursuant to <u>Exhibit D-1</u>.

1.7 Other Basic Economic Terms:

Rent: Base Rent and Additional Rent. (See <u>Sections 4.1</u> and <u>4.2</u>.)

Additional Rent: Tenant's Pro Rata Share of Total Operating Costs, together with all other amounts payable by Tenant under

this Lease other than Base Rent. As provided in Section 4.2(b), Total Operating Costs include

(i) Taxes (Article 5), (ii) utility expenses under Article 6 to the extent provided and paid for by Landlord,

(iii) costs of insurance under Section 7.4, and (iv) Operating Expenses (Article 8).

Expenses Paid Directly by Tenant: All utilities for the Premises (except as set forth in <u>Article 6</u>), and the services to the Premises from time to

time provided by Tenant in accordance with <u>Sections 10.3</u> and <u>10.4</u> and other provisions of this Lease.

Building Services: See <u>Sections 10.3</u> and <u>10.4</u> and <u>Exhibit E</u>.

TI Allowance: An amount equal to \$16,561,530.00, subject to adjustment (if any) in accordance with Paragraph 3.5(b)

and/or 3.5(c) of Exhibit B. See Paragraph 11.2 of Exhibit B.

Letter of Credit: An initial amount equal to \$807,564.50, subject to increase in accordance with <u>Section 14.6</u>.

Guaranty: The guaranty of Tenant's obligations under the Lease provided by Guarantor in accordance with

Section 14.7.

1.8 Appurtenant Rights:

General Common Areas: See Section 2.2(a).

Parking Rights: See Section 2.2(b).

Signage Rights: See Section 2.2(c).

Rooftop Equipment: See <u>Section 2.2(d)</u> and <u>Article 11</u>.

Amphitheater Area: See <u>Section 2.2(e)</u>.

1.9 <u>Expansion/Purchase Rights:</u> The following expansion and purchase rights affecting other Project buildings:

a) Building 2 Expansion Option: See <u>Exhibit C-1</u>.
 b) Building 2 Development Rights See <u>Exhibit C-2</u>.

Purchase Option

c) Building 2 Lease ROFO: See <u>Exhibit C-3</u>.
 d) Building 4 Lease ROFO: See <u>Exhibit C-4</u>.

1.10 ROFO on Building Sale:

See Exhibit D-2.

1.11 <u>Permitted Uses</u>:

Professional, business, and administrative office uses (including, as ancillary thereto, an employee cafeteria or café, lunch rooms, meeting areas, fitness center, and other ancillary uses of the Premises for Tenant's employees and not for the general public), in each case consistent with the operation of a first-class suburban office building, the Special Permit Amendment and other applicable Legal Requirements, the Project Condominium Documents, and the provisions of this Lease.

1.12 <u>Original Notice Addresses</u>:

For Tenant:

For Landlord: c/o Normandy Real Estate Partners

57 Maple Avenue Morristown, NJ 07960 Attn: General Counsel

with copies to:

c/o Normandy Real Estate Partners 57 Maple Avenue

Morristown, NJ 07960

Attn: Head of Property Management

and

c/o Greenfield Partners, LLC 50 North Water Street South Norwalk, CT 06854 Attn: Barry P. Marcus

and

DLA Piper LLP (US) 33 Arch Street Boston, MA 02110

Attn: William B. Forbush III, Esq.

Before the Commencement Date:

TripAdvisor

141 Needham Street

Newton, MA 02464

Attn: Chief Financial Officer

with copies to:

TripAdvisor

141 Needham Street

Newton, MA 02464

Attn: General Counsel

and

Paul Jakubowski, Esq. WilmerHale LLP 60 State Street Boston, MA 02109

After the Commencement Date:

TripAdvisor 400 First Avenue Needham, MA 02494 Attn: Chief Financing Officer

with copies to:

TripAdvisor 400 First Avenue Needham, MA 02494 Attn: General Counsel

and

Paul Jakubowski, Esq. WilmerHale LLP 60 State Street Boston, MA 02109

Columbia Group Realty Advisors, Inc. and Cushman & Wakefield of MA, Inc. (as Tenant's Brokers)

and

Richards Barry Joyce & Partners (as Landlord's Broker)

8

1.13 Brokers:

1.14 <u>Management Company</u>:

1.15 <u>Additional Terms</u>:

Financial Condition Standard:

Material Event of Default:

Normandy Real Estate Management, subject to Article 10.

As of the applicable date under the Lease provision referring to this term, the Financial Condition Standard shall be deemed satisfied if either (a) Tenant has a Net Worth (as defined in Section 14.6) of at least Five Hundred Million Dollars (\$500,000,000), as evidenced by then current financial information provided by Tenant and reasonably acceptable to Project Declarant and Landlord, occupant (b) Tenant then has a public company investment grade credit rating of not less than BBB as determined by Standard & Poors (or the equivalent credit rating as determined by Moody's or Fitch), and to the extent more than one such rating agency has rated Tenant's credit, such credit rating has been determined by each such rating agency.

As of the applicable date under the Lease provision referring to this term, an Event of Default (as defined in Section 14.1) has occurred and is continuing, other than (a) a non-monetary Event of Default that is of an immaterial and readily curable nature (such as a temporarily lapsed insurance certificate due to administrative oversight), provided that Tenant shall cure such Event of Default within thirty (30) days after notice that such an Event of Default has occurred; and (b) a monetary Event of Default for non-recurring Additional Rent charges not exceeding \$10,000 in the aggregate, provided that Tenant shall pay such delinquent amounts within thirty (30) days after notice that such an Event of Default has occurred.

2. <u>Exhibits</u>. The following Exhibits are attached to this Lease and shall be incorporated into this Lease by reference. All references in this Lease to Exhibits shall mean the Exhibits attached to this Lease, unless otherwise expressly specified.

Exhibit A: Project Land and Documents

Exhibit A-1: Project Site Plan Exhibit A-2: Building Site Plan

Exhibit B: Base Building Construction and TI Work Letter

Exhibit B-1: Preliminary Plans

Exhibit B-2: Scope of Base Building Work

Schedule 1: Base Building Work Alternates Schedule 2: Base Building Work Allowances

Exhibit B-3: Requirements for Building Enclosed/Weather Tight

Exhibit B-4: Construction Schedule

Exhibit B-5: Form of Confirmation of Commencement Date

Exhibit B-6: General Requirements for Tenant Work Construction Documents

Exhibit B-7: Additional Parking Costs

Exhibit B-8: Building Measurement Calculations Exhibit B-9: Special Permit Amendment Costs

Exhibit C-1: Building 2 Expansion Option

Schedule 1: Free-Standing Expansion Plan Schedule 2: Connected Expansion Plan Schedule 3: Integrated Expansion Plan

Exhibit C-2: Building 2 Development Rights Purchase Option

Schedule 1: Development Services for Building 2

Schedule 2: Form of Building 2 Development Rights Assignment Schedule 3: Form of Amendment to Master Deed for Building 2

Schedule 4: Form of Unit Deed for Building 2 Unit

Exhibit C-3: Building 2 Lease ROFO
Exhibit C-4 Building 4 Lease ROFO

Exhibit D-1: Extension Options

Exhibit D-2: ROFO on Building Sale

Schedule 1: Due Diligence Materials Schedule 2: Standard Provisions

Exhibit E: Initial List of Building Services
Exhibit E-1: Operating Expenses and Exclusions

Exhibit F: Rules and Regulations

Exhibit G: Tenant Work Insurance Schedule

Exhibit H: Form of Lender's Subordination, Nondisturbance and Attornment Agreement

Exhibit I: Letter of Credit Requirements
Exhibit J: Environmental Reports

Exhibit K: Form of Development Rights Assignment for Original Building Exhibit K-1 Form of Amendment to Master Deed for Original Building Unit

Exhibit K-2 Form of Unit Deed for Original Building Unit

Exhibit L: Form of Notice of Lease Exhibit M: Form of Guaranty

Exhibit N: Tenant Competitor Restriction
Exhibit O: Reference Table for Defined Terms

ARTICLE 2. PREMISES AND APPURTENANT RIGHTS

- 2.1. <u>Lease of Premises</u>. Landlord hereby leases the Premises to Tenant, and Tenant hereby leases the Premises from Landlord, for the Term set forth herein, upon and subject to the terms and conditions of this Lease. The Premises exclude (a) the exterior walls, windows, and roof of the Building, (b) the base building mechanical and service rooms within the Building, (c) the base building equipment and systems installed by Landlord as part of the Base Building Work, including without limitation base building pipes, ducts, conduits, wires and appurtenant fixtures located within the Building or on the Building Site and serving the Building and/or other parts of the Project, whether exclusively or in common, and (d) the Project Common Facilities and Limited Common Elements serving the Premises. If the Premises include less than the entire rentable area of any floor from time to time, then the Premises shall also exclude the common corridors, lobbies, elevator lobby, and restrooms located on such floor.
- 2.2. <u>Appurtenant Rights</u>. During the Term, Tenant shall have, as appurtenant to its lease of the Premises, the rights, in common with others (as applicable) and subject to the terms of this Lease, to use the following facilities serving the Building and/or other portions of the Project. Tenant shall be permitted access to the Building, the Premises, and the Project's parking facilities serving the Building on a 24-hour-per-day, 7-day-per-week basis during the Term, subject to the terms of this Lease.
- (a) <u>Building Site Facilities and General Common Areas</u>. During the Term, Tenant shall have (i) the exclusive right to use the Building's loading dock and other facilities serving the Building that are located on the Building Site and shown on the Building Site Plan (except that Landlord may use such facilities in connection with performing its service, repair, and other obligations for the Building under this Lease) and (ii) the non-exclusive right to use the walkways, driveways, and other Project Common Facilities that serve the Building (including, without limitation, the driveways and walkways on other portions of the Project Land) in common with others, in each case for their customary, intended purposes and subject to the terms of this Lease.
- (b) <u>Parking</u>. During the Term, Tenant shall have the right to use the following parking areas of the Project for daily and transient parking by Tenant's employees, visitors, and other business invitees:
 - (i) <u>Parking Allocation</u>. The Base Building Work to be constructed by Landlord under <u>Exhibit B</u> includes the first phase of Garage B and certain surface parking spaces along the Project roadways as indicated on the "Phase 2A" portion of the Preliminary Plans. The Preliminary Plans for the Initial Site Work (which is part of the Base Building Work described under <u>Exhibit B</u>) indicate approximately 1,047 parking spaces in the first phase of Garage B and approximately 14 surface parking spaces (for a total of approximately 1,061 parking spaces for the Initial Site Work hereunder), subject to final design and striping under <u>Exhibit B</u>. Accordingly, the parking spaces that are initially allocated to the Building under this <u>Section 2.2(b)</u> shall consist of (x) the number

of parking spaces in the first phase of Garage B that, when combined with such surface spaces, satisfies the Minimum Parking Ratio (as defined below) for the Premises RSF (as the final Premises RSF may be adjusted, if applicable, under Paragraph 3.5(b) of Exhibit B), and (y) the balance of the parking spaces (not to exceed the Additional Parking Ratio) in the first phase of Garage B. The "Garage B Spaces" means the number of parking spaces in Garage B that are from time to time allocated to the Building under the provisions of this Section 2.2(b). The "Minimum Parking Ratio" shall mean 3.6 parking spaces per 1,000 square feet of Premises RSF (rounded to the nearest integer), and the "Minimum Parking Allocation" shall mean the number of parking spaces required to satisfy the Minimum Parking Ratio (i.e., a total of approximately 1,011 parking spaces based on the Premises RSF set forth in Section 1.4). The "Additional Parking Ratio" shall mean 0.4 parking spaces per 1,000 square feet of Premises RSF (rounded to the nearest integer) (i.e., a total of approximately 113 parking spaces based on the Premises RSF set forth in Section 1.4). The "Additional Parking Allocation" shall mean the number of Garage B Spaces actually provided under clause (y) above (i.e., approximately 50 parking spaces based on the Premises RSF set forth in Section 1.4) and the capacity of Garage B set forth above, subject to final design and striping under Exhibit B). The "Total Parking Allocation" means, collectively, the Minimum Parking Allocation and the Additional Parking Allocation. The parties hereby acknowledge and agree that (A) Landlord has re-designed the first phase of Garage B to increase its capacity (by adding a partially below-grade parking level as shown on the Preliminary Plans) in order to accommodate Tenant's request for a portion of the Additional Parking Allocation, and (B

- (ii) <u>Additional Parking Costs</u>. Tenant shall be responsible for the costs (the "<u>Additional Parking Costs</u>") of providing Tenant with (A) the Additional Parking Allocation for the Building as provided in this subparagraph (ii) and (B) the Garage A Additional Parking Spaces under subparagraph (iii) below. The Additional Parking Costs for the Additional Parking Allocation spaces in Garage B are set forth on <u>Exhibit B-7</u>. Additional Parking Costs shall constitute BBW Tenant Costs that are paid by Tenant under Paragraph 11.1 of <u>Exhibit B.</u>
- (iii) <u>Garage A Additional Spaces</u>. To the extent the actual number of Additional Parking Allocation spaces initially provided in Garage B under <u>Section 2.2(b)(i)</u> above is less than the number required to fully satisfy the Additional Parking Ratio for the Original Building, then Project Declarant (pursuant to the application of its reserved rights under the Project Condominium Documents) shall cause any remaining spaces required to satisfy the Additional Parking Ratio for the Original Building (anticipated to be approximately 63 parking spaces based on the numbers recited in <u>Section 2.2(b)(i)</u> above) to be allocated to Tenant in Garage A (the "<u>Garage A Additional Parking Spaces</u>") if, as, and when Garage A may hereafter be constructed. The Additional Parking Costs for any Garage A Additional Parking Spaces so provided (the "<u>Garage A Additional Parking Costs</u>") shall be based on the per space cost for Garage A spaces, which shall be determined in the manner provided in Exhibit C-1 for

the Building 2 Additional Parking Costs, whether Garage A is constructed pursuant to Exhibit C-1 or otherwise. If Garage A is constructed pursuant to the applicable Building 2 Rights, then in connection with the construction of Garage A Tenant shall pay the Garage A Additional Parking Costs to Project Declarant for any Garage A Additional Parking Spaces so provided, as more particularly provided under the applicable Building 2 Rights. If Garage A is constructed but not pursuant to the Building 2 Rights, then Tenant shall pay the Garage A Additional Parking Costs to Project Declarant at least sixty (60) days prior to commencement of Garage A construction, on at least thirty (30) days' prior notice from Project Declarant of the estimated Garage A Additional Parking Costs (determined in the manner set forth in Exhibit C-1), subject to final reconciliation upon completion of Garage A in the manner set forth in Exhibit C-1. Any Garage A Additional Parking Spaces that are provided hereunder shall, upon the construction and opening of Garage A, be deemed included in Tenant's Total Parking Allocation for the Original Building. Nothing in this Section 2.2(b) shall be construed to impose on Project Declarant or Landlord any obligation to construct Garage A, except as may be required to be performed by Project Declarant in connection with Tenant's exercise of the applicable Building 2 Rights.

- (iv) <u>Adjustments for Future Development</u>. A portion of the Total Parking Allocation shall be subject to relocation in connection with the further expansion and development of the Project and its parking facilities, as provided below.
 - (A) Construction of Garage A. If and when Garage A is hereafter constructed (whether pursuant to Tenant's Building 2 Rights as provided below or otherwise), then to the extent that the capacity of Garage A exceeds the sixty nine (69) parking spaces allocated to the hotel (the "Garage A Hotel Spaces") and the appurtenant parking spaces for Building 2 (the number of such excess spaces in Garage A being referred to herein as the "Excess Garage A Capacity"), Project Declarant reserves the right to relocate a portion of the Total Parking Allocation to Garage A (after providing the Garage A Additional Parking Spaces as provided above) up to the extent required to utilize the remaining entire Excess Garage A Capacity, as and when directed by Project Declarant following the construction and opening of Garage A. In the event that Tenant has exercised the Building 2 Expansion Option under Exhibit C-1 or the Building 2 Development Rights Purchase Option under Exhibit C-2, then upon the construction and opening of Garage A pursuant thereto, any Excess Garage A Capacity that remains after provision for the Garage A Additional Parking Spaces as set forth above shall be used to re-allocate to Garage A the portion of Tenant's Total Parking Allocation for the Original Building that is required to use such entire remaining Excess Garage A Capacity (i.e., with the result that, apart from the Garage A Hotel Spaces, all spaces in Garage A are allocated to Tenant through the combination of the foregoing provisions).
 - (B) Expansion of Garage B. Project Declarant further reserves the right from time to time to enlarge and expand Garage B in connection with the development of other building(s) on the Project Land, provided that the number of Total Parking Allocation spaces for the Building is not reduced. Without limiting

the generality of the foregoing, the parties acknowledge and agree that construction of the second phase of Garage B is anticipated to provide for drive lanes between the two phases of Garage B which may thereby displace certain of Tenant's allocated parking spaces from the first phase to the second phase of Garage B and that Project Declarant will use reasonable and diligent efforts to cause such construction to be carried out in a manner designed to minimize interference with Tenant's on-going use of the first phase of Garage B during such construction activities.

- (C) <u>Allocation of Surface Spaces</u>. Project Declarant further reserves the right to allocate, designate, relocate, or replace one or more surface spaces for other Project buildings or purposes (e.g., a future building's allocation of such common spaces based on its gross square footage, adjacent roadway location, or construction-related requirements), as Project Declarant may direct, and re-allocate the portion of such Tenant spaces to Garage A, Garage B, and/or other surface parking areas that are reasonably proximate to the Building, in connection with such development.
- (D) <u>Construction Activities</u>. Project Declarant further reserves the right to temporarily relocate any parking spaces required hereunder that may from time to time be temporarily displaced by construction activities, or, if such relocated spaces are not available, render certain spaces temporarily unavailable during such construction activities provided that reasonable and diligent efforts are used to minimize such disruption as set forth in subparagraph (B) above.
- (E) <u>Visitor Parking Areas</u>. Project Declarant further reserves the right, whether at the request of Tenant under subparagraph (vi) below or otherwise, to designate certain spaces in the Project's common parking facilities for visitor parking purposes, with any such visitor spaces so requested by Tenant (or otherwise allocable to Tenant in Garage B or other parking facility serving the Building pursuant to the provisions of this <u>Section 2.2(b)</u>) being included in Tenant's applicable parking allocation hereunder.
- (F) Interim Surface Parking Lot for Remaining Spaces. If requested by Tenant, by written notice to Project Declarant and Landlord on or before December 31, 2016, Project Declarant will reasonably cooperate with Tenant to seek to obtain applicable permits for, and to design and construct (after BBW Substantial Completion has occurred under Exhibit B and prior to construction of Garage A) an interim surface parking lot on an unimproved portion of site for the future Garage A (in an area between the hotel's surface parking lot and Garage B), which interim surface parking lot (if constructed hereunder) shall be sufficient to provide for Tenant's temporary use a number of surface parking spaces at least equal to the number of parking spaces for the Additional Parking Ratio for the Original Building that were not provided in Garage B (i.e., approximately 63 spaces based on the numbers recited in Section 2.2(b)(i) above), subject to the following terms and conditions. Tenant shall pay to Project Declarant, in advance at the times and manner reasonably designated by Project

Declarant, one hundred percent (100%) of all hard and soft costs (including a 3% development fee on all such costs) incurred by Project Declarant to permit, design, and construct such interim surface parking lot (inclusive of all costs of related site improvements and related re-configurations of any roadways, sidewalks, or other Project Common Facilities in connection with such interim surface parking lot). Prior to commencing construction of such interim surface parking lot, Project Declarant shall provide Tenant with a reasonably detailed budget for such work. Any scheduling, funding, or other delivery requirements relating to the interim surface parking lot (including the timing and manner of Tenant's payment of any costs in excess of the estimated costs and the refund to Tenant of any unapplied deposits made by Tenant for the estimated costs of the interim surface parking lot) shall be as may be mutually agreed by Project Declarant and Tenant prior to commencement of construction of such work. Tenant reserves the right, by written notice to Project Declarant (with a copy to Landlord) at any time prior to commencement of construction of the interim surface parking lot, to withdraw its request that Project Declarant construct such interim surface parking lot, provided that (i) Tenant shall remain responsible for paying all costs incurred (including permitting, planning, pre-construction services, and other similar soft costs) in connection with Tenant's prior request for such interim surface parking lot and (ii) Project Declarant shall return to Tenant any pre-funded estimated hard costs of construction (if any) that Tenant may have paid to Project Declarant in anticipation of such work. If such interim surface parking lot is constructed, all costs to operate and maintain the interim surface parking lot and related walkways leading from such lot to the Building shall be allocated to and paid by Tenant as Project Common Facilities Expenses allocable to the Building under Section 4.2(d) and Article 8. If, as, and when construction of Garage A (or any construction staging for Garage A or Building 2 on such site) is scheduled to commence (whether pursuant to the Building 2 Rights or otherwise) Project Declarant reserves the right to permanently discontinue Tenant's use of such interim surface parking lot, effective upon not less forty (45) days prior written notice to Tenant, and without any obligation to provide any replacement parking spaces of any kind. Tenant's payment for and use of such interim surface parking lot shall have no effect on Project Declarant's obligations under Section 2.2(b)(iii) or (iv) above to provide, or on Tenant's obligation to accept and pay for, the Garage A Additional Parking Spaces if, as, and when provided under the terms of Section 2.2(b)(iii) or (iv) above. Landlord shall have no obligation or liability to Tenant of any kind in respect of the development, delivery, or operation of any such interim surface parking lot, and any delay or failure by Project Declarant in causing such interim surface parking lot to be provided shall not give rise to any default by Landlord under this Lease or otherwise affect any of Tenant's rights and obligations under this Lease. Project Declarant reserves the right, at its election, to defer the development of any such interim surface parking lot in the event that (1) Project Declarant determines that development of such interim surface parking lot would adversely impact the schedule for developing or obtaining permits or approvals for any other phase of the Project, (2) a Material Event of Default by Tenant then exists,

- or (3) the Lease is assigned to any Third Party or Tenant has entered into one or more subleases or other agreements with one or more Third Parties for the use of occupancy of more than fifty percent (50%) of the Premises.
- (v) General Parking Provisions. During the Term, Tenant's use of parking spaces for the Total Parking Allocation shall be without separate charge to Tenant, other than for (A) the Additional Parking Costs (including, as applicable, any Garage A Additional Parking Costs) as set forth above, and (B) the Building's allocable share of Project Common Facilities Expenses (as defined below) in respect of the Total Parking Allocation pursuant to the Project Condominium Documents, which shall be allocated to the Building and paid by Tenant as part of Operating Expenses under Article 8, including any additional costs for extra operational measures requested by Tenant pursuant to Section 2.2(b)(vi) below. Tenant's use of the Project's parking facilities hereunder shall be subject to, and undertaken in compliance with, the Project Condominium Documents and applicable rules and regulations thereunder with respect to the use of such Project parking facilities. The parking rights provided to Tenant under this Section 2.2(b) are provided solely for the purposes set forth above, shall not be used by Tenant or any other Tenant Party (as defined below) for any other purpose, and shall not be assigned, sublicensed, or otherwise transferred to any other party, except in connection with a sublease or assignment of this Lease permitted or approved in accordance with Article 13.
- (vi) <u>Garage Operational Matters</u>. During the Term, if Tenant from time to time believes that other Project users are materially infringing on Tenant's allocated parking spaces in the garage(s) hereunder, Tenant may from time to time request the implementation of reasonable controlled access mechanisms (e.g., card readers, gates, or parking pass systems) that are designed to ensure that the parking spaces from time to time allocated to Tenant in accordance with this <u>Section 2.2(b)</u> remain available for Tenant's use. In such event, Landlord shall reasonably cooperate with Tenant to develop such controlled access mechanisms in conjunction with Project Declarant (or, as applicable, the condominium association responsible for administering the parking garage facilities under the Project Condominium Documents) and any other Project users that from time to time share the applicable parking garage in common with Tenant (e.g., the implementation of a card reader system for Garage B during the period when both phases of Garage B are operational and used by Tenant and other Project users). During the period when Tenant is entitled to use all of the Garage A spaces (through the combination of rights provided above under this <u>Section 2.2(b)</u>) other than the Garage A Hotel Spaces, such controlled access measures may include, if requested by Tenant, the installation of an interior gate regulating the portion of Garage A allocated for Tenant's exclusive use, provided that such gate shall be removed at Tenant's expense at such time as Tenant shall cease to be entitled (through the combination of rights provided above under this <u>Section 2.2(b)</u>) to use all of the Garage A spaces other than the Garage A Hotel Spaces, whether due to the expiration or termination of the Lease, the occupancy of all or part of Building 2 by a third party, or otherwise. In addition, if Tenant from time to time requests that a reasonable number of its allocated parking spaces under <u>Section 2.2(b)</u> be designated for visitor parking purposes, Landlo

users to implement such visitor parking areas for Tenant and, as applicable, other Project users in accordance with the Project Condominium Documents. The implementation of any measures requested by Tenant under this Section 2.2(b)(vi) shall be at Tenant's sole cost and expense, whether as a direct reimbursement of the costs of installing, maintaining, replacing, and removing at the expiration or earlier termination of the Term any such special parking equipment or access systems or as on-going reimbursements of additional operating, maintenance, personnel, and other costs, as the case may be, which amounts in each case shall be added to the Project Common Facilities Expenses that are allocated to the Building and paid by Tenant as part of Operating Expenses under Article 8.

- (c) Exterior Building Sign. During the Term, Tenant shall have the right, at its sole cost and expense, to install and maintain (i) a prominent exterior sign identifying Tenant's name and corporate logo (using Tenant's graphics and colors) in at least one location on the Building's façade and (ii) a so-called monument sign identifying Tenant's name and corporate logo (using Tenant's graphics and colors) in a location on the Building Site reasonably approved by Landlord, in accordance with and subject to the following requirements. All such exterior signage shall comply with applicable Legal Requirements and shall be consistent with first-class single-building office tenant signage suitable for a first-class suburban office park, taking into account its prominent visibility from Route 128 and its proximity to other Project buildings (the "First-Class Signage Standard"). Prior to installing any such exterior signage, Tenant shall submit to Landlord for its review and approval (which shall not be withheld or unreasonably delayed following Tenant's submission if such exterior signage complies with the requirements of this Section 2.2(c)) detailed construction plans and specifications showing the location, size, height, graphics, materials, electric lighting, manner of installation, and other details of the proposed signage installation. Landlord acknowledges that such signage may include a lighted or back-lit sign consistent with a first-class suburban office park, subject to receiving all required governmental permits. Tenant shall be solely responsible, at its sole cost and expense, for installing, maintaining, operating, and repairing all such exterior signage in compliance with all Legal Requirements and in accordance with the First-Class Signage Standard. Tenant shall be responsible for obtaining all governmental permits and approvals required for the installation of such exterior signage. Landlord shall cooperate, in its capacity as owner of the Property, as may be reasonably requested by Tenant in connection with applications for approvals from municipal authorities for any exterior signage hereunder, provided that Tenant shall reimburse Landlord, as Additional Rent, for its reasonable, out-of-pocket third-party expenses (if any) incurred by Landlord in connection with such efforts. At the expiration or earlier termination of the Term, Tenant shall remove all such exterior signage (together with all related mounting brackets, supports, fasteners, wiring, and any other signage-related equipment), repair any damage to the Building or the Building Site caused by such installations and removal, and restore the affected areas to their prior condition. Except for Tenant's exterior signage rights hereunder, Landlord shall not grant exterior signage rights on the Building to any third party.
- (d) <u>Rooftop Equipment</u>. During the Term, Tenant shall have the right, at its sole cost and expense, to install and maintain a satellite dish or other rooftop telecommunications device, and other equipment on the roof of the Building, all as more particularly provided in <u>Article 11</u>.

- (e) <u>Amphitheater Area</u>. The "<u>Amphitheater Area</u>" means the portion of the Project Land immediately adjacent to the Building Site that is outlined as the "Amphitheater Area" on the Preliminary Plans. The site improvements for the Amphitheater Area are more particularly shown on the "Phase 2A" portion of the Preliminary Plans and are to be constructed by Landlord as part of the Initial Site Work in accordance with <u>Exhibit B</u>.
 - (i) <u>Tenant's Use of Amphitheater Area</u>. During the Term, Tenant shall have the right to use the Amphitheater Area for non-exclusive routine exterior usage (e.g., for employee lunch-time outdoor seating) and for occasional special event usage (e.g., organized employee gatherings or weekend special events) incident to Tenant's use and occupancy of the Premises, in each case in a manner suitable for a prominent exterior area of a first-class suburban office park, taking into account its proximity to the Building and other then existing Project buildings and facilities, and in compliance with applicable Legal Requirements and applicable provisions of the Project Condominium Documents (the "<u>First-Class Exterior Use Standard</u>"). The on-going costs of operating and maintaining the Amphitheater Area shall be allocated as provided in <u>Section 4.2(d)</u> below.
 - (ii) Limited Common Element; Allocation of Costs. The parties acknowledge that, pursuant to the Building Condominium Instruments under Section 2.4(a) below, (A) the Amphitheater Area constitutes a Limited Common Element of the Original Building Unit (as defined below) under the Project Condominium Documents and shall not become a Limited Common Element of any other Project building without the approval of the owner of the Original Building Unit, (B) any substantial alteration of the Amphitheater Area after its initial construction under Exhibit B (excluding routine maintenance and periodic replacement of worn landscaping elements) and any special event usage of the Amphitheater Area shall require the prior approval of the owner of the Original Building Unit, (C) the costs of operating, maintaining, and (as applicable) replacement of worn elements of the Amphitheater Area shall be allocated entirely to the Original Building (which amounts shall be included in Total Operating Expenses under this Lease), and (D) the Amphitheater Area shall be an open space within the Project available for use by occupants and visitors of other Project buildings for purposes of routine exterior usage (e.g., for employee lunch-time outdoor seating) in accordance with the First-Class Exterior Use Standard, with any proposed special event usage by occupants of any other Project building requiring the prior approval of the owner of the Original Building Unit. Tenant's rights to use of the Amphitheater Area shall be without a separate user charge under the provisions of the Project Condominium Documents, other than for Total Operating Costs attributable to the Amphitheater Area as provided in this Lease and costs and expenses arising from any special event usage by Tenant or any Tenant Party.
 - (iii) <u>Tenant Approval Rights for Alterations or Special Event Usage</u>. Landlord hereby agrees that, during the Term provided no Material Event of Default then exists, (A) any substantial alteration of the Amphitheater Area after its initial construction under <u>Exhibit B</u> (excluding routine maintenance and periodic replacement of worn landscaping elements with comparable items consistent with the First-Class Exterior Use Standard) shall require Tenant's prior approval, which approval shall not be unreasonably withheld

for alterations from time to time required to replace worn-out exterior site improvements, to remedy unsafe conditions, to comply with applicable Legal Requirements, and for other improvements consistent with the First-Class Exterior Use Standard, (B) any proposed special event usage by any occupant of any other Project building shall require Tenant's prior approval in its sole discretion, and (C) Landlord shall not approve the Amphitheater Area becoming a Limited Common Element of any other Project building without obtaining Tenant's prior approval in its sole discretion.

- (iv) <u>Routine Use by Other Project Buildings</u>. Tenant acknowledges and agrees that (A) the Amphitheater Area shall remain an open space within the Project, without fences or barriers, and shall be freely accessible by means of the common walkways leading around and through the Project, (B) employees and visitors of any Project buildings may from time to time use such area for non-exclusive routine exterior purposes (e.g., employee lunchtime outdoor seating) in accordance with the First-Class Exterior Use Standard, and (C) neither Landlord, Project Declarant (or its respective successors and assigns as owners of other Project phases), nor the condominium association under the Project Condominium Documents shall have any obligation to monitor or restrict access to the Amphitheater Area by Project occupants, their guests or invitees, or other third parties who may from time to time enter such areas.
- 2.3. Reservations. In addition to other rights reserved herein or by law, Landlord reserves the right from time to time, without unreasonable (except in emergency) interference with Tenant's rights hereunder, including without limitation Tenant's use and occupancy of the Premises and access thereto: (i) to perform repairs to or reconstructions of the Building and to install, use, maintain, repair, replace and relocate for service to the Premises and other parts of the Building, or either, pipes, ducts, conduits, wires and appurtenant fixtures, wherever located in the Premises, the Building, or elsewhere on the Building Site or the Project Land, provided, however, such work shall not materially adversely affect the use of the Premises or access thereto for their intended purposes (other than on a temporary basis to accommodate such work); (ii) to construct, alter, or relocate any portion of the Project Common Facilities from time to time in connection with the current or future development of the Project and the construction, maintenance, repair, and replacement from time to time of Project Common Facilities in accordance with Section 16.13, (iii) to grant easements and other rights with respect to the Project Land in accordance with Section 16.13, and (iv) to designate the street address of the Building prior to the Commencement Date (and thereafter to change such address with Tenant's written consent, not to be unreasonably withheld, conditioned or delayed). Installations, replacements and relocations within the Building under clause (i) shall be located as far as practicable in the base building core areas of the Building, above ceiling surfaces, below floor surfaces, or within perimeter walls of the Premises. Any entry by Landlord into the Building hereunder shall be subject to the provisions of Section 9.6, and Landlord shall use reasonable efforts to minimize disruption to Tenant's business activities to the degree reasonably practicable and endeavor to perform during non-business hours any such work that
- 2.4. <u>Project Development Documents</u>. For purposes of the phased development of the Project and separately owning and/or financing different Project buildings, the Project Land was submitted to a commercial condominium regime under the provisions of Chapter 183A of the

General Laws of Massachusetts (the "<u>Condominium Statute</u>") pursuant to the Project Condominium Documents as set forth in <u>Exhibit A</u>. This Lease, and Tenant's rights hereunder, are subject and subordinate to the Project Condominium Documents, the Master Plan Permits (as defined in the Project Condominium Documents) as affected by the Special Permit Amendment obtained under <u>Exhibit B</u>, all other matters of record, and all Legal Requirements (as defined below) affecting the Building, the Building Site, and the Project Land (collectively, as amended from time to time, the "<u>Project Documents</u>"), subject to and in accordance with the provisions of this Lease (including, without limitation, the provisions of this <u>Section 2.4</u> set forth below).

2.5. Landlord's Interest in the Building. Concurrently with the execution of this Lease, (1) the Project Declarant has assigned to Landlord, an affiliate of the Project Declarant, the development rights reserved under the Project Condominium Documents with respect to the construction of the Building (including site work and related parking areas set forth in Exhibit B) required under this Lease pursuant to that certain Deed of Easements and Assignment and Conveyance of Development Rights of even date herewith (the "Development Rights Assignment") (the form of which is attached hereto as Exhibit K), and (2) the parties have executed and recorded a Notice of Lease (in the form attached hereto as Exhibit L) in respect of this Lease (the "Notice of Lease"). Pursuant to the Project Condominium Documents and the Development Rights Assignment, after Landlord's construction of the portions of the Base Building Work sufficient under the Condominium Statute to permit the preparation of condominium unit floor plans and the recording of a unit deed for the condominium Unit for the Building, the Project Declarant will add the condominium unit for the Building (the "Building Unit" or the "Original Building Unit") to the Condominium pursuant to an amendment to the Master Deed substantially in the form attached hereto as Exhibit K-1 (the "Building Master Deed Amendment") and issue to Landlord and record a unit deed for the Building Unit substantially in the form attached hereto as Exhibit K-2 (the "Building Unit Deed"). The Development Rights Assignment, together with the Building Master Deed Amendment and the Building Unit Deed, are collectively referred to herein as the "Building Condominium Instruments." The parties hereby acknowledge and agree that the rights and obligations of Landlord and Tenant under this Lease are binding upon and inure to the benefit of Landlord (as holder of the development rights for the Building under the Development Rights Assignment) and Tenant, respectively, and shall automatically be binding upon and inure to the benefit of Landlord (as the holder of the Building Unit Deed upon its recording) and Tenant, respectively, with the same effect as if the Notice of Lease had been recorded immediately after the recording of the Building Master Deed Amendment and the Building Unit Deed, without the requirement for Landlord and Tenant to execute any further instrument, but in confirmation thereof at the request of either party the parties shall execute and record, immediately after the recording of the Building Master Deed Amendment and the Building Unit Deed as set forth above, an amended and restated Notice of Lease acknowledging the foregoing. As used in this Lease, references to "Property" or Landlord's interest in the Property shall be deemed to mean, as of the Lease Date, all of Landlord's right, title, and interest to the Building and the Building Site under the Development Rights Assignment and, upon the recording of the Building Master Deed Amendment and the Building Unit Deed as set forth above, all of Landlord's right, title, and interest in and to the Building Unit. Landlord reserves the right, if deemed necessary in connection with the construction financing for the Building, to cause the Building Site (and related areas of the Project Land necessary to perform the Base Building Work) to be subjected to a ground lease for the period prior to the recording of the Building Unit Deed, in which event this Lease shall be

subject and subordinate to such ground lease and the related ground leasehold mortgage as provided in <u>Article 15</u>, provided that such subordination shall be conditioned upon delivery of a subordination, non-disturbance and attornment agreement with respect to such ground lease as provided in <u>Article 15</u> which provides for the protection of Tenant's rights under this Lease in a manner substantially equivalent to the subordination, non-disturbance and attornment agreement for mortgages required under <u>Article 15</u>.

Landlord shall not amend or terminate any of the Project Condominium Documents in a manner that (i) materially adversely affects the use of the Premises for their intended purposes (or with access thereto or the appurtenant parking rights for the Building from time to time provided under the terms of Section 2.2(b) for their intended purposes) in each case as more particularly set forth under the terms and conditions of this Lease; (ii) conflicts with Tenant's rights or privileges under the express terms of this Lease, or (iii) imposes on Tenant any Materially Increased Obligation (as defined below), without the prior written consent of Tenant. As used herein, "Materially Increased Obligation" shall mean a material increase in the obligations under the Project Condominium Documents beyond the obligations from time to time arising as contemplated and provided for under the terms of the Project Condominium Documents existing as of the Lease Date (as amended by the Building Condominium Instruments and as may hereafter be further amended in accordance with the Master Deed, subject to the limitations set forth in Sections 8.08(q) and 8.12 of the Master Deed, the "Phased Project Condominium Documents"). For avoidance of doubt, the parties acknowledge and agree that the development of future phases of the Project under the Phased Project Condominium Documents, the construction, addition, alteration, replacement, or other changes of or to any buildings or common elements of the Project thereunder (including, without limitation, roadways, sidewalks, landscaped areas, and other facilities thereunder) in connection the development or re-development of the Project from time to time, and the allocation of costs in connection with such development or re-development of the Project and the common elements thereunder, all as contemplated and provided for under the terms of the Phased Project Condominium Documents, shall not be deemed to violate the standards set forth in the first sentence of this paragraph or to require any consent by Tenant with respect thereto. Landlord shall cause the trustee appointed by Landlord (as owner of the Building Unit) to the condominium association under the Project Condominium Documents to exercise the Building Unit's voting rights thereunder in conformity with the foregoing restrictions and otherwise in a manner consistent with the requirements of this Lease.

- (b) <u>Joinder by Project Declarant</u>. The Project Declarant, as holder of the remaining development rights for the Project under the Project Condominium Documents, has executed the Joinder attached to this Lease for the sole purposes of:
 - (1) acknowledging the Development Rights Assignment and Project Declarant's obligation to record the Building Master Deed Amendment and issue the Building Unit Deed to Landlord as set forth therein;
 - (2) granting to Tenant the Expansion/Purchase Rights under Exhibits C-1, C-2, and C-3 in respect of the Building 2 phase of the Project (the "Building 2 Rights") and under Exhibit C-4 in respect of the Building 4 phase of the Project (the "Building 4 Rights"), in each case subject to and in accordance with the terms more particularly set forth therein;

- (3) agreeing not to amend or terminate the Project Condominium Documents in a manner that (i) materially adversely affects the use of the Premises for their intended purposes (or with access thereto or the appurtenant parking rights for the Building from time to time provided under the terms of Section 2.2(b) for their intended purposes) in each case as more particularly set forth under the terms and conditions of this Lease; (ii) conflicts with the Building 2 Rights or the Building 4 Rights granted hereunder, or (iii) imposes on Tenant any Materially Increased Obligation, without the prior written consent of Tenant; provided that the development of future phases of the Project under the Project Condominium Documents, the construction, addition, alteration, replacement, or other changes of or to any buildings or common elements of the Project thereunder (including, without limitation, roadways, sidewalks, landscaped areas, and other facilities thereunder) in connection the development or re-development of the Project from time to time, and the allocation of costs in connection with such development or re-development of the Project and the common elements thereunder, all as contemplated and provided for under the terms of the Phased Project Condominium Documents, shall not be deemed to violate the standards set forth in this clause (3) or to require any consent by Tenant with respect thereto. By way of illustration, the addition or alteration of Project Common Facilities in connection with the development of a future phase of the Project, or the allocation to the Building of its Proportionate Share (as defined in the Master Deed) of the costs of maintaining and operating such added or altered Project Common Facilities, or the adjustment of the Proportionate Shares of the Units in accordance with Section 8.08(q)(ii) of the Master Deed shall not be deemed to constitute a Materially Increased Obligation;
 - (4) agreeing to the provisions of Sections 2.2(b) and 16.13 that are applicable to the Project Declarant; and
 - (5) agreeing to the Tenant Competitor Restriction set forth in Exhibit N.

Concurrently with the execution of this Lease, the Project Declarant has also executed the joinder to the Notice of Lease to give record notice of the Expansion/Purchase Rights and Tenant Competitor Restriction, which shall establish record priority for the respective Expansion/Purchase Rights and Tenant Competitor Restriction and, unless the same have lapsed or been terminated in accordance with their respective terms, shall be binding (as to Building 2) on the Project Declarant (and its successors and assigns) in respect to the Building 2 phase of the Project and (as to Building 4) on the Project Declarant (and its successors and assigns) in respect to the Building 4 phase of the Project. The provisions of Section 16.4 shall also be deemed to apply to the Project Declarant (and its successors and assigns) in respect of their respective obligations under this Section 2.4(b), as though for purposes thereof the limitations of liability under Section 16.4 referred to the Project Declarant's (and its successors' and assigns') respective interests in the Building 2 phase of the Project and the Building 4 phase of the Project, as applicable. Tenant acknowledges and agrees that (i) the rights granted to Tenant by the Project Declarant under this Section 2.4(b) may not be amended or modified except by written instrument executed by the Project Declarant (or its respective successors and assigns in respect

to the respective rights for other Project buildings) and (ii) the rights inuring to or reserved for the benefit of the Project Declarant under the express terms of this Lease (including, without limitation, under Section 2.2(b) above and Section 16.13 below) shall inure to the benefit of, and shall be directly enforceable by, the Project Declarant (or such respective successors and assigns). For so long as the Building 2 Rights, the Building 4 Rights, or the Tenant Competitor Restriction remain applicable to other portions of the Project, respectively, Landlord and Tenant shall from time to time provide the Project Declarant (or such respective successors and assigns) with such information as may from time to time be reasonably requested by Landlord or the Project Declarant (or such respective successors and assigns) to establish whether the conditions to the applicability of such rights or restrictions have been satisfied by Tenant.

ARTICLE 3. LEASE TERM

- 3.1 <u>Lease Term</u>. The Initial Term of this Lease is set forth in <u>Section 1.5</u>. Following the Commencement Date, at the request of either party, Landlord and Tenant shall enter into a recordable instrument confirming the occurrence of the Commencement Date in the form of <u>Exhibit B-5</u>, provided, however, that the failure by either party to execute such an instrument shall not be deemed to delay the occurrence of the Commencement Date as determined in accordance with the provisions of this Lease.
- 3.2 <u>Building Construction and Delivery.</u> Landlord shall perform the Base Building Work for the Building and Tenant shall perform the TI Work for the Building, all in accordance with and subject to the provisions of <u>Exhibit B</u>. The procedures for developing the final Construction Documents for the Base Building Work and for the final Construction Documents for the TI Work, along with relevant milestone dates and applicable Tenant and Landlord remedies with respect thereto, are all set forth in <u>Exhibit B</u>.
- 3.3 Extension Options. Tenant shall have two (2) options to extend the Initial Term of the Lease, each for a successive extension period of five (5) years, subject to and in accordance with the terms and conditions set forth in Exhibit D-1.

ARTICLE 4. RENT

- 4.1 <u>Base Rent</u>. Tenant covenants and agrees to pay to Landlord the Base Rent (at the rate set forth in <u>Section 1.6</u>) commencing on the Commencement Date (subject to <u>Section 4.1(a)</u>) and thereafter on the first day of each calendar month during the Term, in accordance with <u>Section 4.3</u> below.
- (a) <u>Rent Waiver Periods</u>. Notwithstanding the foregoing, subject to the provisions of <u>Section 14.3(c)</u> below (if applicable), Landlord agrees to waive payment of (i) the monthly installments of Base Rent (together with the monthly installments of Additional Rent for Taxes, Operating Expenses, and other Total Operating Costs defined below) for the four-(4)-month period commencing on the Commencement Date through the date immediately preceding the Partial Rent Commencement Date (the "<u>Gross Rent Waiver Period</u>") and (ii) the monthly installments of Base Rent (only) during the period commencing on the Partial Rent

Commencement Date through the date immediately preceding the Rent Commencement Date (the "Base Rent Waiver Period"), it being agreed that during the Base Rent Waiver Period Tenant shall pay all Additional Rent attributable to such period.

- (b) <u>Property Management Fee During Rent Waiver Periods</u>. The parties further agree that (x) no Property Management Fee (as defined in Paragraph 1(a) (17) of <u>Exhibit E-1</u>) shall be charged to Tenant during the Gross Rent Waiver Period, (y) during the Base Rent Waiver Period, the Property Management Fee shall be charged to Tenant at the monthly rate (pro-rated on a per diem basis) based on 2.5% of an imputed monthly Base Rent at the rate that would have applied during the first Lease Year under <u>Section 1.6</u> in the absence of the foregoing waiver of Base Rent, and (z) the Property Management Fee shall thereafter be charged to Tenant on a monthly basis in advance in accordance with Paragraph 1(a)(17) of <u>Exhibit E-1</u> (or, as applicable, <u>Section 10.3(a)(y)</u>).
- (c) <u>General.</u> If the Gross Rent Waiver Period ends on a day that is not the last day of a calendar month, then on the Partial Rent Commencement Date Tenant shall make a pro-rated payment of Additional Rent (calculated on a per diem basis) for the balance of the partial calendar month in which the Gross Rent Waiver Period expires. If the Base Rent Waiver Period ends on a day that is not the last day of a calendar month, then on the Rent Commencement Date Tenant shall make a pro-rated payment of Base Rent (calculated on a per diem basis) and continue to pay the Additional Rent for the balance of the partial calendar month in which the Base Rent Waiver Period expires. For any partial calendar month at the expiration or earlier termination of the Term, the regular monthly installment of Base Rent and Additional Rent shall be pro-rated on a per diem basis.
- 4.2 <u>Additional Rent</u>. Tenant covenants and agrees to pay to Landlord during the Term (except as provided in <u>Section 4.1(a)</u> for the Gross Rent Waiver Period) all Additional Rent (as defined in <u>Section 1.7</u>) as provided in this <u>Section 4.2</u> and the applicable provisions of the Lease referred to herein.
- (a) <u>Early Occupancy Period</u>. In the event that prior to the Commencement Date Tenant, at its election, first commences the regular conduct of its business in any portion(s) of the Premises comprising at least 10,000 RSF in the aggregate (which commencement of business shall not be deemed to have occurred by virtue of Tenant's performance of the TI Work for the Premises or the installation or testing of computers, equipment, or other personal property in the Premises in preparation for such commencement of business), then during the period (if any) from such commencement of business through the end of the day that immediately precedes the Commencement Date (the "<u>Early Occupancy Period</u>"), all of the provisions of the Lease applicable to the Premises (including without limitation under <u>Article 4</u>) shall be applicable to the Premises, other than the obligation to pay the Base Rent for the Premises during the Early Occupancy Period, and Tenant shall pay to Landlord all Additional Rent under this <u>Article 4</u> that is attributable to such Early Occupancy Period, including, without limitation, the Property Management Fee under Paragraph 1(a)(17) of <u>Exhibit E-1</u> at the monthly rate (pro-rated on a per diem basis) based on 2.5% of an imputed monthly Base Rent at the rate that would have applied during the first Lease Year under <u>Section 1.6</u> in the absence of the waiver of Base Rent. For any portion(s) of the Premises comprising less than 10,000 RSF in the aggregate that Tenant may occupy for the regular conduct of its business prior to the Commencement Date, such occupancy

shall be on all of the terms of this Lease, other than the obligation to pay Base Rent or Additional Rent, provided that Tenant shall be responsible for the costs of any utilities used by Tenant during such period (in the manner set forth in Section 7.1) and for the costs of any Building Services or services for parking facilities or other Project Common Facilities actually provided by Landlord under Section 10.3 during such period, which amounts to the extent incurred by Landlord shall be paid by Tenant as Additional Rent.

- (b) Estimated and Final Payments. For each calendar year or portion thereof during the Term, Landlord shall estimate the amounts of (i) all Taxes under Article 5, (ii) all utility costs (to the extent the same are not separately metered and required to be paid to the applicable utility company by Tenant) under Article 6, (iii) all insurance premiums to be paid by Landlord under Article 7, and (iv) all Operating Expenses under Section 8.1 (collectively, the "Total Operating Costs"). Commencing on the Commencement Date (except for the Gross Rent Waiver Period under Section 4.1(a)) and thereafter on the first day of each calendar month during the Term, Tenant shall pay one-twelfth of Tenant's Pro Rata Share of such estimated Total Operating Costs monthly in advance together with the monthly installment of Base Rent then due. Landlord shall endeavor to provide Tenant with such estimate on or before the Partial Rent Commencement Date (for the balance of such calendar year) and on or before each subsequent December 1 (for the next ensuing calendar year) during the Term of the Lease. Landlord may adjust its estimates of Total Operating Costs at any time based upon its experience and reasonable anticipation of costs. Such adjustments shall be effective as of the next Rent payment date after notice to Tenant. Within approximately one hundred twenty (120) days after the end of each calendar year following the Commencement Date, Landlord shall provide Tenant with a final statement of the Total Operating Costs paid or incurred by Landlord during the immediately preceding calendar year of the Term and a calculation of Tenant's Pro Rata Share of such Total Operating Costs; provided that any Project Common Facilities Expenses (as defined in Section 8.1) under the Project Condominium Documents shall be reconciled separately within sixty (60) days following Landlord's receipt of annual accountings thereunder during the Term. Within thirty (30) days after delivery of such statements, Tenant shall pay to Landlord any underpayment, or Landlord shall credit any overpayment to Tenant against Additional Rent next due, of Tenant's Pro Rata Share of such Total Operating Costs (or, if the Term has expired, promptly refund such amount to Tenant, after deduction of any amounts due to Landlord for unsatisfied Tenant obligations under the Lease). If the Term expires or this Lease is terminated as of a date other than the last day of a calendar year, Tenant's payment of Additional Rent pursuant to this Section 4.2(b) for such partial year shall be based on Landlord's best estimate of the items otherwise includable in Total Operating Costs and shall be made on or before the later of (i) thirty (30) days after Landlord delivers such estimate to Tenant or (ii) the last day of the Term, with an appropriate payment or refund (in the manner provided above) to be made upon Tenant's receipt of Landlord's statement of Total Operating Costs for such calendar year. This <u>Section 4.2(b)</u> shall survive expiration or earlier termination of the Term.
- (c) <u>General Provisions</u>. This Lease requires Tenant to pay directly to suppliers, vendors, carriers, contractors, and other parties, certain insurance premiums, utility costs, personal property taxes, maintenance and repair costs, and other expenses. In the event that, in accordance with the express provisions of this Lease, Landlord from time to time shall pay any such amounts, Tenant shall reimburse Landlord for such costs in full upon demand with the next monthly Rent payment. Unless this Lease expressly provides otherwise, Tenant shall pay all

Additional Rent within thirty (30) days after the same has been billed to Tenant. In no event shall Landlord's failure to demand payment of Additional Rent be deemed a waiver of Landlord's right to such payment.

- (d) Allocation of Certain Operating Costs. The parties acknowledge that, in the event that certain services or facilities are provided solely to the Building or jointly to the Building and other buildings in the Project at any time during the Term, then the applicable expenses for such services or facilities shall be charged entirely to those buildings, including the Building if applicable, receiving such services. By way of illustration, in the event that certain facilities initially serve only the Original Building (or, following the construction of Building 2, serve only the Original Building and Building 2) the costs relating to such facilities would be shared by the buildings served by such facilities, in accordance with the Project Condominium Documents governing the allocation of costs for facilities serving one or more Project buildings but not all Project buildings. Landlord and Tenant agree that (i) the costs of maintaining and operating the Amphitheater Area under Section 2.2(e) above, which is a Limited Common Element of the Building, shall include, without limitation, costs of trash removal, grass-cutting, snow removal, and general landscaping work, and that such costs shall be allocated entirely to the Building, (ii) the costs of maintaining and operating the first phase of Garage B, which is initially a Limited Common Element of only the Building, shall be initially allocated entirely to the Building and (during such portion of the Term in which Garage B may become Limited Common Elements of the Building and other Project building(s) as provided in Section 2.2(b). above) an allocable portion of the costs of maintaining and operating Garage B shall be allocated to the Building in the proportion that the number of parking spaces in Garage B that are from time to time allocated to Tenant under the applicable provisions of Section 2.2(b) above bears to the total number of parking spaces in Garage B (as the same may be expanded), (iii) pursuant to the applicable provisions of Section 2.2(b) above, if and when Garage A is constructed, an allocable portion of the costs of maintaining and operating Garage A shall be allocated to the Building under this Lease in the proportion that the number of parking spaces in Garage A that are allocated to the Building under the terms of this Lease bears to the total number of parking spaces in Garage A, (iv) the costs of maintaining and operating any interim surface parking lot (if any) installed under Section 2.2(b)(iv)(F) shall be allocated entirely to the Building, (v) the costs of maintaining and repairing the sidewalks that lead from Garage B (and, if applicable, from the interim surface parking lot under <u>Section 2.2(b)(iv)(F)</u>) to the Original Building, which sidewalks serve only the Building, shall be allocated entirely to the Original Building, and (vi) if, as, and when Garage A is constructed, the costs of maintaining and repairing the sidewalks that lead from Garage A to the Original Building and Building 2 shall be allocated between the Original Building and Building 2 in the respective proportions that the number of parking spaces in such parking facility that are allocable to each such building bears to the total number of parking spaces in such parking facility that are allocable to such buildings.
- 4.3 <u>Method of Payment</u>. Tenant shall pay the Base Rent to Landlord in advance in equal monthly installments by the first of each calendar month during the Term (except as provided in <u>Section 4.1(a)</u> above) and the monthly installment of Tenant's Pro Rata Share of Total Operating Costs as provided in <u>Section 4.2</u>. Tenant shall make a pro-rata payment (on a per diem basis) of Base Rent and Additional Rent for any period of less than a month at the beginning or end of the Term. All payments of Base Rent, Additional Rent and other sums due shall be paid, without demand, set-off or other deduction, except as otherwise expressly set forth

herein, in current U.S. exchange. Tenant shall pay the regular monthly installments of Base Rent under <u>Section 4.1</u> and Additional Rent under <u>Section 4.2(b)</u> by electronic fund transfer pursuant to wire instructions from time to time provided by Landlord unless and until otherwise directed by Landlord. All other amounts due to Landlord hereunder shall be paid by Tenant either in the manner provided under the preceding sentence or by check drawn on a clearinghouse bank and delivered to the Address of Landlord set forth in <u>Article 1</u> or to such other place or account as Landlord may from time to time direct.

It is intended that the Base Rent payable hereunder shall be a net return to Landlord throughout the Term, free of expense, charge, offset, diminution or other deduction whatsoever (except as expressly provided herein) on account of the Premises (excepting Landlord's financing expenses, federal and state income taxes of general application, and those expenses that this Lease expressly makes the responsibility of Landlord), and all provisions hereof shall be construed in light of such intent. Without limiting the foregoing, except as expressly provided in this Lease, Tenant's obligation to pay Rent shall be absolute, unconditional, and independent of any Landlord covenants and shall not be discharged or otherwise affected by any law or regulation now or hereafter applicable to the Premises, or any other restriction on Tenant's use, or (except as expressly provided herein) any casualty or taking, or any failure by Landlord to perform or other occurrence; and Tenant waives all rights now or hereafter existing to quit or surrender this Lease or the Premises or any part thereof, or to assert any counterclaim or defense in any action seeking to recover Rent (unless such counterclaim or defense would be lost by Tenant if not raised in such proceeding).

4.4 Audit Rights. Landlord shall keep books and records regarding Total Operating Costs for each year during the Term. All records shall be retained for at least three (3) years after the end of the applicable year. At the request of Tenant ("Tenant's Audit Notice") given within one hundred twenty (120) days after Landlord delivers Landlord's final statement of Total Operating Costs with respect to any calendar year during the Term, Tenant shall have the right to examine Landlord's books and records regarding Total Operating Costs for such year. Such right to examine the records shall be exercisable (a) upon reasonable advance notice to Landlord and at reasonable times during Landlord's business hours, (b) only during the ninety-(90)-day period (the "Audit Period") following Tenant's Audit Notice, and (c) only if no Material Event of Default then exists. Landlord shall make such books and records available at Landlord's office in Massachusetts or at the Project, or in electronically accessible form. The audit of Landlord's Total Operating Costs for such year may be conducted, at Tenant's election, either by members of Tenant's internal financial staff or by a certified public accountant experienced in suburban Boston office lease audits as to which Landlord does not reasonably object. If Tenant reasonably requests copies of any such materials, Tenant shall reimburse Landlord for the out-of-pocket costs thereof. Tenant shall not engage any such accountant or any other party on a contingency fee basis in connection with any examination of Landlord's books and records hereunder. As a condition precedent to performing any such examination of Landlord's books and records, Tenant's examiners shall be required to execute and deliver to Landlord an agreement in form reasonably acceptable to Landlord agreeing to keep confidential any non-public, confidential information about Landlord, the Building, or the Project in connection with such examination and not to disclose the results of such examination except as required

(or, if the Term has expired, promptly refund such amount to Tenant, after deduction of any amounts due to Landlord for unsatisfied Tenant obligations under the Lease). Notwithstanding the foregoing, Tenant's request to audit Landlord's books and records shall not extend the time within which Tenant is obligated to pay the amounts shown on Landlord's statement of Total Operating Costs, and Tenant may not make any request to audit Landlord's books and records at any time when Tenant is in default of such payments. In the event that, as a result of such an audit, Tenant has been overcharged by five percent (5%) or more of the Additional Rent actually due for Total Operating Costs for such year, Landlord shall (in addition to refunding or crediting the amount overcharged as provided above) reimburse Tenant for the reasonable third-party costs of engaging its auditor, not to exceed \$20,000. In all other cases, Tenant shall pay for the cost of such audit

If requested by Tenant from time to time, Landlord shall authorize Tenant, under Article 9 of the by-laws of the Project's condominium association and at Tenant's expense, to inspect or examine the books and records of the Project condominium association with respect of the Project Facilities Expenses allocated to Tenant hereunder. If Tenant reasonably determines that the Project condominium association has improperly allocated to the Building any element of Project Facilities Expenses that are in turn charged to Tenant under the provisions of this Lease, Tenant may from time to time so notify Landlord of such claimed improper allocation, and if requested by Tenant, Landlord shall use reasonable efforts at Tenant's expense (which may but need not include pursuing arbitration under Article 13 of the condominium association's by-laws) to correct any such misallocation under the applicable provisions of the Project Condominium Documents or, if such Landlord efforts are unavailing or if Landlord determines not to pursue such remedies, to permit Tenant, as Landlord's designated agent, to pursue the correction of such misallocation under the applicable provisions of the Project Condominium Documents, provided that Tenant shall be responsible for all costs thereof (including any late charges or interest assessed under Section 3.11 of the by-laws of the condominium association in respect to the disputed amounts in question) and the payment of all amounts due under the Lease in order to ensure that the Property remains free of all liens, including without limitation any liens arising under the Project Condominium Documents from the disputed amounts in question.

4.5 <u>Cooperation With Tenant's Accounting Requirements.</u> If requested by Tenant from to time, Landlord will provide, not more often than quarterly, financial information to Tenant with respect to Total Operating Costs incurred to date for the then-current year to the extent available to Landlord, as is reasonably required by Tenant's accountants and auditors for Tenant to comply with lease accounting requirements applicable to Tenant (provided that nothing herein shall be deemed to expand, modify or limit Tenant's rights under <u>Article 4</u> above, and any such information and Tenant's rights to the same shall be subject to the provisions of <u>Section 4.4</u> as if it were an audit of Landlord's books and records). Tenant shall reimburse Landlord for the reasonable out-of-pocket costs to provide such information as Additional Rent within thirty (30) days after invoice.

ARTICLE 5. TAXES

- 5.1 <u>Taxes.</u> Tenant covenants and agrees to pay to Landlord, as Additional Rent, Tenant's Pro Rata Share of the Taxes for each fiscal tax period, or ratable portion thereof, included in the Term (except for the Gross Rent Waiver Period under <u>Section 4.1(a)</u>). If Landlord receives a refund of any such Taxes (other than with respect to such Gross Rent Waiver Period), Landlord shall credit against Additional Rent next due or, at Landlord's election, refund to Tenant (or, if the Term has expired, remit to Tenant any such amounts not then fully credited or refunded hereunder, after deduction of any amounts due to Landlord for unsatisfied Tenant obligations under the Lease) Tenant's Pro Rata Share of the refund of the Taxes previously paid by Tenant, in each case after deducting Landlord's reasonable costs and expenses incurred in obtaining the refund (to the extent such costs and expenses were not previously included in Operating Expenses or Taxes), but in any event such refund to Tenant shall not exceed amounts paid by Tenant for Taxes on account of the period subject to such refund. Upon Tenant's request from time to time, Landlord shall furnish Tenant with a copy of the applicable real estate tax bill. Tenant shall make estimated payments on account of Taxes in monthly installments on the first day of each month, in amounts reasonably estimated from time to time by Landlord pursuant to <u>Section 4.2(b)</u>.
- 5.2 <u>Definition of "Taxes"</u>. The term "<u>Taxes</u>" means all real estate taxes, assessments, betterments, excises, user fees and all other governmental charges and fees of any kind or nature, or impositions or agreed payments in lieu thereof or voluntary payments made in connection with the provision of governmental services or improvements of benefit to the Building, the Building Site, or the Property, and all penalties and interest thereon (to the extent due to Tenant's failure to make timely payments hereunder), assessed or imposed against the Building, the Building Site, or the Property (including without limitation any personal property taxes levied on the Building or the Property or on fixtures or equipment used in connection therewith), other than a federal or state income tax of general application.
- (a) Exclusions. Notwithstanding the foregoing, Taxes shall not include (i) any of the foregoing which are levied or assessed against the Property to the extent (x) the same are not attributable to the Term (other than for the Early Occupancy Period, if applicable) or (y) the same are special development fees or governmental development assessments (as opposed to general real estate taxes) for the initial construction of on-site or off-site street or intersection improvements, roads, rights of way, lighting, and traffic signals for the initial development or construction of the Building or any other building of the Project, (ii) any inheritance, estate, gift, excise, franchise, income, gross receipts, capital levy, revenue, rent, state, payroll, stamp or profit taxes, however designated; (iii) any interest or penalties resulting from the late payment of taxes by Landlord (except to the extent due to Tenant's failure to make timely payments hereunder); or (iv) reserves for future Taxes.
- (b) <u>Allocations for Common Elements</u>. Subject to the foregoing exclusions, to the extent not already included in the municipality's assessment for the Building Unit, (i) Taxes imposed with respect to any Limited Common Elements that are exclusive to the Building (including, without limitation, the Amphitheater Area during the Term and the first phase of Garage B during the period such garage exclusively serves the Building) shall be allocated entirely to the Building, (ii) Taxes imposed with respect to any Limited Common Elements that are shared by the Building and any other Project building(s) (e.g., the shared Garage A under the circumstances set forth <u>in Section 2.2(b)</u>) shall be allocated among such buildings based on their respective Proportionate Shares (or, in the case of any shared garage facility, based on the respective number of parking spaces allocated to the respective buildings), and (iii) Taxes for any other Project Common Facilities shall be included in the Project Common Facilities Expenses allocable under <u>Article 8</u>.

- (c) <u>Changes in Taxation</u>. If during the Term the present system of taxation of real or personal property shall be changed so that, in lieu of or in addition to the whole or any part of such tax there shall be assessed, levied or imposed on the Property or Premises or on Landlord any kind or nature of federal, state, county, municipal or other governmental capital levy, income, sales, franchise, excise or similar tax, assessment, levy, charge or fee (as distinct from the federal and state income tax in effect on the Lease Date) measured by or based in whole or in part upon Building valuation, mortgage valuation, rents, services or any other incidents, benefits or measures of real property or real property operations, then any and all of such taxes, assessments, levies, charges and fees shall be included within the term of Taxes; provided, however, that Tenant's obligation with respect to such substitute taxes shall be limited to the amount thereof as computed at the rates that would be payable if the Building and Property were the only property of Landlord.
- (d) <u>Abatement Expenses</u>. Taxes shall also include reasonable expenses, including reasonable fees of attorneys, appraisers and other consultants, incurred by Landlord in connection with any efforts to obtain abatements or reduction or to assure maintenance of Taxes for any year wholly or partially included in the Term, whether or not successful and whether or not such efforts involved filing of actual abatement applications or initiation of formal proceedings.
- (e) Taxes Prior to Assessment of Building Unit. In the event that the condominium trust receives an assessment from the Town for any period of the Term or the Early Occupancy Period before the Building Unit is separately assessed by the Town for real estate tax purposes, Landlord will reasonably and equitably allocate to the Building, in accordance with the Building Condominium Instruments, (i) the portion of the assessment received by the condominium trust attributable to the Building, any other improvements on the Building Site, and any improvements on the Project Land that are Limited Common Elements of the Building or otherwise exclusively serve the Building (e.g., the municipal assessor's work sheets regarding the valuation of the Building, any other improvements on the Building Site, and the first phase of Garage B) and (ii) an allocable share of the assessment for the Project Land and other general common elements of the Project based on the proportion that the gross square footage of the Building bears to the gross square footage of the Project phases that have not been separately assessed, whether undeveloped or developed (including the Building), and provide Tenant with reasonable back-up documentation for such allocations of the assessment received by the condominium trust for such period.
- (f) <u>Abatements</u>. Landlord shall, upon the written request of Tenant provided no Event of Default then exists, commence a proceeding for abatement of real estate Taxes attributable to the Building Unit, provided Landlord shall thereafter have the right to settle such proceeding in its reasonable discretion. From and after the date on which Landlord acquires the unit deed for the Building Unit, Tenant may, in its sole discretion if Landlord declines to pursue an abatement proceeding for the applicable fiscal year during the Term, initiate an abatement proceeding for real estate Taxes payable for such year at Tenant's sole cost and expense by prior written notice to Landlord and in compliance with any Legal Requirements applicable to such

proceeding. If Tenant pursues an abatement of real estate Taxes pursuant to the immediately preceding sentence, then any abatement proceeds shall be payable to Landlord after deducting Tenant's reasonable expenses incurred in obtaining such abatement. In the event of any abatement of Taxes for a period occurring during the Term of this Lease, Tenant shall be entitled to Tenant's Pro Rata Share of any refund (after deducting Landlord's or Tenant's, as applicable, reasonable cost in obtaining an abatement, if any, to the extent not previously included in Operating Expenses or Taxes) but in any event such refund to Tenant shall not exceed the amounts on account of Taxes actually paid by Tenant with respect to the period subject to the abatement.

- 5.3 <u>Personal Property Taxes</u>. Tenant shall pay directly all taxes charged against Tenant's trade or business fixtures, furnishings, equipment, inventory, or other personal property (collectively, "<u>Tenant Property</u>."). Tenant shall use commercially reasonable efforts to have Tenant Property taxed separately from the Property. Landlord shall notify Tenant if any of Tenant Property is taxed with the Property, and Tenant shall pay such taxes to Landlord within thirty (30) days after such notice.
- 5.4 <u>TIF Incentives</u>. Any exemption from real property taxes for the Building or the Property due to any Tax Increment Financing Agreement entered into by the Tenant and the Town of Needham shall be allocated entirely to the Building, so that Taxes payable by Tenant during the Term reflect the full amount of the exemption to which Tenant is entitled under the terms of such Tax Increment Financing Agreement for the applicable periods provided thereunder.

ARTICLE 6. UTILITIES

- 6.1 <u>Utilities</u>. Tenant shall pay all charges for water, sewer, gas, electricity, telecommunications, and other utilities or like services used or consumed at the Premises, the Building, or the Building Site (each, a "<u>Utility Service</u>" and collectively the "<u>Utility Services</u>"), including without limitation any Utility Services used or consumed by all mechanical equipment serving the Premises, the Building, or the Building Site, wherever located, whether called use charge, tax, assessment, fee or otherwise as the same become due. Each company or third party providing the applicable Utility Service hereunder is referred to a "<u>Utility Service Provider</u>."
- (a) <u>Tenant's Direct-Metered Utilities</u>. Tenant shall pay, directly to the applicable Utility Service Provider, all costs and expenses associated with all separately metered Utility Services exclusively serving the Premises or (while Tenant is the sole tenant) the Building as and when due to such Utility Service Provider ("<u>Tenant's Direct-Metered Utilities</u>"). The costs of Tenant's Direct-Metered Utilities, to the extent paid by Tenant directly to the applicable Utility Service Provider, shall not constitute part of the reimbursable Operating Expenses under <u>Article 8</u>. Tenant shall be responsible for maintaining all utility deposits required by the applicable Utility Service Provider for such Tenant's Direct-Metered Utilities and will from time to time upon Landlord's reasonable request provide evidence of Tenant's payment of all charges to the applicable Utility Service Provider.

- (b) <u>Landlord Utilities</u>. For any Utility Services serving the Building or the Building Site that from time to time are not paid by Tenant as Tenant Direct-Metered Utilities (e.g., a Utility Service, if any, that is not metered separately to the Building or that may serve the Building or the Building Site in common with other portions of the Project) and are paid by Landlord ("<u>Landlord Utilities</u>"), Tenant shall pay to Landlord, as Additional Rent, all costs and expenses associated with such utility charges based on check-meters or similar sub-metering equipment (or, if none exists, on reasonable allocations prepared by Landlord's building engineer) and periodic reading of such meters, at the same rate paid by Landlord to the Utility Service Provider; provided that Additional Rent for any such metered (or sub-metered) utilities paid by Landlord may be reasonably estimated monthly by Landlord, subject to periodic reconciliations based on actual readings, and shall be paid monthly by Tenant within thirty (30) days after being billed with a final accounting based upon actual bills received from the applicable Utility Service Providers following the conclusion of each year. The costs of such Landlord Utilities shall be included as part of the reimbursable Operating Expenses under Article 8.
- (c) <u>Metering.</u> The Base Building Work will include a common switching point(s) at the Building for certain Utility Services for the Building (collectively, the "<u>Utility Switching Points</u>"), as and to the extent specified in <u>Exhibit B-2</u> to be part of Base Building Work. Metering equipment for or within the Building will be installed by the respective parties as specified under <u>Exhibit B-2</u>. Tenant shall pay for any and all costs to install and connect Utility Services from the Utility Switching Points to the Premises and other parts of the Building and the Building Site used by Tenant hereunder. Landlord shall be under no obligation as to any Utility Services beyond the foregoing responsibility to install the Utility Services to the Utility Switching Points, and Landlord shall not be liable for any interruption or failure in the supply of any utilities or Utility Services, except to the extent expressly set forth below.
- (d) <u>Utility Service Providers</u>. At either party's request from time to time (or at annual service meetings under <u>Section 10.3</u>), the parties shall review alternative Utility Service Providers for the Building. The parties acknowledge that such alternatives may from time to time include Utility Service Providers for Utility Services to the Building and other Project building(s) or facilities (e.g., for multi-year electricity service contracts for multiple buildings), provided that Landlord makes no representation or warranty as to the availability or terms from time to time of any such multi-year or multi-building arrangements for any such Utility Service. For any Tenant's Direct-Metered Utilities under <u>Section 6.1(a)</u>, Tenant shall have the right to select or approve (or thereafter change) the applicable Utility Service Provider. For any Landlord Utilities under <u>Section 6.1(b)</u>, to the extent permitted by law, Landlord shall have the right at any time and from time to time during the Term to contract for or purchase one or more of such Utility Services as Landlord may determine to reflect market terms and conditions; provided that if such Utility Service exclusively serves the Premises, the same shall be subject to Tenant's approval, not to be unreasonably withheld, conditioned, or delayed. In requesting Tenant's approval for any such proposed Utility Service Provider, Landlord shall provide Tenant with such information as Tenant may reasonably request regarding the proposed contract for such service. Provided there shall be no unreasonable interference with Tenant's operations within the Premises, Tenant agrees reasonably to cooperate with Landlord and the Utility Service Providers and at all times as reasonably necessary, and on reasonable advance notice, shall allow Landlord and the Utility Service Providers reasonable access to any utility

lines, equipment, feeders, risers, fixtures, wiring and any other such machinery or personal property within the Building or the Building Site and associated with the delivery of Utility Services. If Tenant desires to select or change a particular Utility Service Provider from time to time as set forth above, (i) as to any Utility Service Provider for Tenant's Direct-Metered Utilities under Section 6.1(a), Tenant shall provide Landlord with reasonable prior notice of such change and any alterations in connection therewith shall be subject to Tenant's compliance with Section 10.5 and other provisions of this Lease, (ii) as to any Utility Service Provider for Landlord Utilities under Section 6.1(b), Landlord shall not unreasonably withhold its approval of the same (subject to Tenant's compliance with Section 10.5 and other provisions of this Lease), and (ii) in all cases, Tenant shall be responsible for all costs relating thereto, including without limitation all costs of designing and installing new service lines or equipment, granting or otherwise coordinating under the applicable Project Documents any necessary utility easements, connecting to and/or removing other existing base building equipment or other equipment in the Building or elsewhere on the Building Site or the Project Land, and terminating the prior Utility Service Provider, and any such installation shall comply with the applicable provisions of Section 10.5 and other provisions of this Lease.

(e) Service Interruption. Tenant shall look solely to the applicable Utility Service Provider for all service issues with respect to all Tenant's Direct-Metered Utilities under Section 6.1(a), and Landlord shall (to the maximum extent permitted by law) have no liability to Tenant, nor shall Rent be abated, for any interruption, curtailment, or suspension of any Such service. In the event that there shall be an interruption, curtailment, or suspension of any Landlord Utilities under Section 6.1(b) (and no reasonably equivalent alternative service or supply is provided by Landlord or Tenant, e.g., Tenant's own emergency generator or other back-up power supply equipment) that renders all or a portion of the Premises untenantable for its intended purposes for a period of three (3) consecutive days (in the case of service interruption due to the negligence of Landlord or its employees, contractors, or agents ("Landlord Parties")) or seven (7) consecutive days (in the case of service interruption not due to such negligence) following receipt by Landlord of written notice from Tenant of such event, other than as caused, in whole or in part, by reasons beyond Landlord's reasonable control (such as a generally applicable power outage or other off-site occurrence) or by an act or omission in violation of this Lease by Tenant or by any negligence or willful misconduct of any of Tenant's agents, employees, contractors, invitees, subtenants, successors or others using the Premises under or through Tenant (collectively, with Tenant, the "Tenant Parties" or any one of them, including Tenant, a "Tenant Party") (an event that satisfies the foregoing conditions being referred to herein as a "Service Interruption"), then Tenant shall be entitled to an equitable abatement of Base Rent and Tenant's Pro Rata Share of Total Operating Costs, based on the nature and duration of the Service Interruption, the area of the Premises so affected, and the then current Rent amounts, for the period that commencing on the expiration of such thre

ARTICLE 7. INSURANCE

- 7.1 <u>Tenant's Coverage</u>. Tenant shall maintain during the Term insurance for the benefit of Tenant and Landlord (as their interests may appear) from insurers rated at least A-/X by A. M. Best (subject to the provisions of <u>Section 7.2</u> below), with terms and coverages reasonably satisfactory to Landlord and with such increases in limits as Landlord may from time to time reasonably request, consistent with requirements at other comparable first class suburban office buildings in the "Route 128 West, Class A Market" ("<u>Comparable Properties</u>"). Initially, Tenant shall maintain the following:
 - (a) Commercial general liability insurance naming Landlord, Landlord's property manager, Landlord's mortgagee(s), and other parties designated by Landlord from time to time as additional insureds, with coverage for premises/operations, personal injury, and contractual liability with combined single limits of liability of not less than \$10,000,000 for bodily injury and property damage per occurrence with a per location aggregate;
 - (b) Property insurance that shall be primary on all Tenant Property (including Tenant's FF&E) and, if applicable, on any Tenant-Insured Work (as defined below) in an amount adequate to cover their replacement cost, including a vandalism and malicious mischief endorsement, and sprinkler leakage coverage; business interruption insurance, loss of income and extra expense insurance covering all perils covered by a standard, "Special Form" (as defined from time-to-time by the insurance industry) property insurance policy. As used herein, "Tenant-Insured Work" shall mean any Tenant Work that is not required to be insured as Leasehold Improvements under Landlord's property insurance policy for the Building under Section 7.4(b) below. Tenant's insurance on any Tenant-Insured Work shall name Landlord and Landlord's mortgagee(s) from time to time as additional loss payees as their interests may appear. Tenant's property insurance under this subparagraph (b) shall cover special perils including theft and such other risks Landlord may from time to time reasonably designate if such risks are required by landlords to be insured by tenants of Comparable Properties under similar circumstances, for the full replacement cost value of the covered items and in amounts that meet any co-insurance clause of the policies of insurance;
 - (c) Workers' compensation insurance with statutory benefits, and employers' liability insurance with customary limits (in no event less than \$100,000 per employee and \$500,000 per policy); and
 - (d) Automobile liability insurance on vehicles operated by employees of Tenant including own and non-owned automobiles, if applicable.

In addition, Tenant shall carry such other coverages, and in such amounts, as are required by Landlord from time to time, so long as such coverages and amounts are within the range of coverages typically required for comparable tenancies in Comparable Properties or are required under the Project Condominium Documents.

Prior to the Commencement Date and on each anniversary of that date (or on the policy renewal date), Tenant shall give Landlord certificate(s) evidencing such coverage in a customary

form (for institutionally-owned properties) reasonably specified by Landlord and with an affirmative statement from the issuer of such policy that it may not be canceled or coverage limits reduced without at least thirty (30) days' prior written notice to Landlord and Tenant; provided that if the issuer will not provide such affirmative statement, then the statement will require the issuer to endeavor to provide notice of such cancellation or change to Landlord at least thirty (30) days in advance and Tenant shall provide Landlord with notice of such cancellation or change at least thirty (30) days in advance. Liability insurance maintained by Tenant shall be deemed to be primary insurance, and any liability insurance maintained by Landlord shall be deemed secondary to it.

Tenant may use blanket or excess umbrella coverage to satisfy any of the requirements of this <u>Section 7.1</u>, provided that the Premises is specifically named in any blanket coverage and the limits thereon are available on a per property basis and on such basis comply with the required limits set out herein and that any umbrella coverage is provided on a "following form" basis.

7.2 Action Increasing Rates. Tenant shall comply with Sections 9.1, 9.2, 9.3, and 9.4 and in addition shall not, directly or indirectly, use the Premises in any way that is prohibited by law. If Tenant, directly or indirectly, uses the Premises in any way that jeopardizes any insurance coverage carried by Landlord or Tenant as reasonably documented by evidence provided by Landlord to Tenant, then Tenant shall, if such use is in violation of the other terms and conditions of this Lease, promptly stop such use. Tenant shall, in any event, reimburse Landlord upon demand for all of Landlord's costs incurred in providing any insurance to the extent attributable to any special endorsement or increase in premium resulting from the particular business or operations of Tenant, and any special or extraordinary risks or hazards resulting therefrom, including without limitation, any risks or hazards associated with the generation, storage and disposal of hazardous or toxic materials. Notwithstanding the foregoing, Tenant's use of the Premises for the Permitted Uses specified in Section 1.11 (as opposed to Tenant's particular use) in compliance with the terms and conditions of this Lease shall not be deemed legally prohibited or dangerous to people or property for the purposes of this Section 7.2. Tenant shall cure any breach of this Lease on account of Tenant's failure to carry the insurance required by this Section 7.2 within ten (10) days after notice from Landlord and Tenant shall have no further notice or cure right under Article 14 for any such breach.

7.3 Waiver of Subrogation. Landlord and Tenant each waive any and every claim for recovery from the other for any and all loss of or damage to the Premises, the Building, or the Property or any part of it, or to any of its contents (including without limitation any Tenant Property), to the extent such loss or damage is covered by property insurance or would have been covered by property insurance required hereunder. Landlord waives any and every such claim against Tenant that would have been covered had the insurance policies required to be maintained by Landlord by this Lease been in force, to the extent that such loss or damage would have been recoverable under such policies. Tenant waives any and every such claim against Landlord that would have been recoverable under such policies required to be maintained by Tenant under this Lease been in force, to the extent that such loss or damage would have been recoverable under such policies. This mutual waiver precludes the assignment of any such claim by subrogation (or otherwise) to an insurance company (or any other person), and Landlord and Tenant each agree to give written notice of this waiver to each insurance company that has issued or shall issue any property insurance policy to it, and to have the policy properly endorsed, if

necessary, to prevent invalidation of the insurance coverage because of this waiver. Reference is further made to the provisions on waivers of claims and subrogation rights set forth in the Project Condominium Documents, which shall apply in accordance with their terms.

7.4 Landlord's Insurance and Indemnity. Landlord shall purchase and maintain, during the Term, policies with insurance companies and with such coverages, amounts, and deductibles as may be required under the Project Condominium Documents or by Landlord's mortgagee (or otherwise determined by Landlord to be prudent or appropriate in a manner consistent with Comparable Properties), for the following: (a) commercial general liability insurance for incidents occurring in or about the common areas of the Building or the Building Site; and (b) property insurance written on a "special form" policy (or its then equivalent) covering property damage to the Building and improvements on the Building Site (including the Leasehold Improvements, but excluding Tenant's FF&E and other Tenant Property) and loss of rental income (covering a period of not less than eighteen (18) months from the date of fire of other casualty), covering special perils for the full replacement cost value, together with such other policies for coverages, amounts, deductibles, and risks as may be required under the Project Condominium Documents or such mortgagee or otherwise determined by Landlord to be prudent or appropriate in a manner consistent with Comparable Properties. As used herein, "Leasehold Improvements" shall mean (i) the TI Work initially installed by Tenant under Exhibit B and (ii) any other Tenant Work that may be installed in the Building from time to time during the Term in accordance with and pursuant to plans approved by Landlord under Section 10.5, in each case exclusive of Tenant's FF&E and other Tenant Property. As set forth in Section 4.2(b), the cost of such insurance shall be included in Total Operating Costs, and Tenant's Pro Rata Share of the cost thereof shall be borne by Tenant. The Building's allocable share of insurance costs for the Project Common Facilities maintained under the Project Condominium Documents shall also be included in Operating Expenses under Article 8. Subject to the terms and limitations set forth in this Lease (including, without limitation, Sections 7.3 and 16.4), and except to the extent due to the negligence or willful misconduct of Tenant or any Tenant Party, Landlord shall indemnify, save harmless and defend Tenant from and against any claims, damage, loss, cost, or expense (including without limitation reasonable legal fees) made against Tenant for injury or damage to person or property in the common areas of the Building or the Building Site to the extent caused by the negligence or willful misconduct of Landlord or any Landlord Party.

ARTICLE 8. OPERATING EXPENSES

8.1 <u>Operating Expenses</u>. "<u>Operating Expenses</u>" shall mean (a) all costs and expenses associated with the operation, management, maintenance and repair of the Building, the Building Site, and the Property as more particularly described, and subject to the exclusions set forth, in <u>Exhibit E-1</u> and (b) the portion of the Common Expenses and Parking Expenses (as defined in the Project Condominium Documents) that is allocated and assessed to the Building for the Project Common Facilities and any Limited Common Elements in accordance with the Phased Project Condominium Documents as amended from time to time in accordance with the requirements set forth in <u>Section 2.4</u> above (the "<u>Project Common Facilities Expenses</u>"). The Project Common Facilities Expenses include, without limitation, the costs of operating and maintaining the Amphitheater Area (as provided in <u>Section 2.2(e)</u> and <u>Section 4.2(d)</u> above) and

the share of the Parking Expenses that is allocable under the Project Condominium Documents to Tenant's Total Parking Allocation under Section 2.2(b) and Section 4.2(d) above, subject to the limitations set forth in Exhibit E-1. For avoidance of doubt, with respect to the initial portion of the Term when the first phase of Garage B is a Limited Common Element of the Original Building and all parking spaces in the first phase of Garage B are allocated to Tenant under Section 2.2(b), all Parking Expenses for the first phase of Garage B (except as limited under the terms of Exhibit E-1) shall be allocated to Tenant as part of the Project Common Facilities Expenses hereunder. For any portion of the Term during which Tenant shares the use of Garage B (or Garage A, as applicable) as shared Limited Common Elements serving the Building and other Project buildings, the Parking Expenses for such shared parking facilities shall be allocated to Tenant based on the number of spaces allocated to Tenant under this Lease in the applicable garage. Tenant shall pay Tenant's Pro Rata Share of Operating Expenses in accordance with Section 4.2.

ARTICLE 9. USE OF PREMISES

9.1 <u>Permitted Uses</u>. Tenant may use the Premises only for the Permitted Uses described in <u>Section 1.11</u> and shall not use the Premises for any other purpose. Tenant shall keep the Premises equipped with appropriate safety appliances to the extent required by applicable laws or insurance requirements relating to Tenant's use of the Premises. Tenant shall comply with Landlord's rules and regulations (the "<u>Rules and Regulations</u>") reasonably promulgated from time to time for the Building and the Building Site and provided to Tenant, provided the same are not inconsistent with the provisions of this Lease, and Tenant shall use reasonable efforts to cause its agents, contractors, customers and business invitees to comply therewith. Landlord's initial Rules and Regulations for the Building and the Building Site are attached hereto as <u>Exhibit F</u> and shall further be deemed to include the Rules and Regulations applicable to the Project Common Facilities and other areas of the Project under the Project Condominium Documents (copies of which have been provided to Tenant). Without limiting the generality of the foregoing, Tenant shall not use the Premises or any Appurtenant Exterior Area (as defined below) in a manner that violates the Project Condominium Documents or the Rules and Regulations thereunder.

9.2 <u>Indemnification</u>. From and after the Commencement Date, subject to the provisions of this Lease, (a) Tenant shall assume exclusive control of all areas of the Premises, including all improvements, utilities, equipment, and facilities therein, and (b) Tenant shall be responsible for the Premises and all of Tenant's improvements, equipment, facilities and installations, wherever located on the Property and all liabilities, including without limitation tort liabilities incident thereto. Tenant shall indemnify, save harmless and defend Landlord, and its members, managers, officers, directors, employees, property manager, and mortgagees (collectively, "<u>Indemnitees</u>") from and against any and all claims, damages, losses, penalties, costs, expenses and fees (including reasonable attorneys' fees) arising in whole or in part out of (i) any injury, loss, theft or damage (except to the extent due to the negligence or willful misconduct of any Indemnitee or Landlord Party) to any person or property while in or about the Premises or the Building or (to the extent caused by the negligence or willful misconduct of any Tenant Party) elsewhere on the Building Site, the Project Common Facilities, or other areas of the Project (collectively, the "<u>Appurtenant Exterior Areas</u>"); (ii) any condition in or about the

Premises, the Building, or the Appurtenant Exterior Areas caused by any work or installations by any Tenant Party or the negligence or willful misconduct of any Tenant Party; (iii) the use of the Premises, the Building, or the Appurtenant Exterior Areas by, or any act or omission of, any Tenant Party, (iv) any mechanics lien or similar lien against the Building, the Building Site, the Property, or the Project Land arising from work or services provided for or arranged by any Tenant Party, and (v) any other failure by Tenant to perform its obligations under this Lease. The provisions of this <u>Section 9.2</u> are subject to the limitations set forth elsewhere in this Lease (including, without limitation, <u>Section 16.4(b)</u>) and shall survive the expiration or earlier termination of this Lease.

9.3 Compliance With Legal Requirements. Tenant shall not cause or (by any Tenant Party) permit the Premises or any Appurtenant Exterior Areas to be used in any way that violates any applicable laws, ordinance, orders, codes, rules, regulations, permits and approvals of any court, governmental entity, or governmental agency of competent jurisdiction or any restrictive covenant, encumbrance, or other Project Document applicable to the Premises, the Building, or the Appurtenant Exterior Areas (each a "Legal Requirement" and collectively the "Legal Requirements") or any provision of this Lease, or constitutes a nuisance or waste, and shall comply with all Legal Requirements applicable to the Premises and Tenant's use of the Property. Tenant shall obtain and pay for all permits and shall promptly take all actions necessary to comply with all Legal Requirements applicable to Tenant's use of the Premises, including without limitation the Occupational Safety and Health Act. Notwithstanding the foregoing two sentences to the contrary, Landlord shall be responsible for the compliance of the Base Building Work with all Legal Requirements that are applicable to the Base Building Work as of the Commencement Date, in each case in accordance with and subject to the terms of Exhibit B. Tenant shall maintain in full force and effect all certificates of occupancy, governmental permits, certifications, or approvals required for Tenant's operations at the Premises (collectively, the "Operating Permits"). Tenant shall be solely responsible for procuring and complying at all times with any and all necessary Operating Permits and shall provide Landlord with a copy of all Operating Permits promptly upon issuance. Tenant shall also be solely responsible for complying with all reporting requirements directly relating or incident to the conduct of its activities on the Premises, including without limitation (if applicable) with respect to the transportation, storage, handling, use and disposal of any Hazardous Substances (if any) in connection with Tenant's operations at the Property. Within ten (10) business days after a request by Landlord, which request shall be made not more than once during each period of twelve (12) consecutive months during the Term hereof, unless otherwise requested by any purchaser, investor, or mortgagee of Landlord, Tenant shall furnish Landlord with copies of all Operating Permits that Tenant possesses or has obtained together with a certificate certifying that such Operating Permits are all of the Operating Permits that Tenant possesses or has obtained with respect to the Premises. Tenant shall promptly give notice to Landlord of any written orders, warnings or violations relative to the above received from any federal, state, or municipal agency or by any court of law and shall promptly comply with and cure the conditions causing any such violations in accordance with applicable Legal Requirements. Tenant shall not be deemed to be in default of its obligations under the preceding sentence to promptly cure any condition causing any such violation in the event that, in lieu of such cure, Tenant shall contest the validity of such violation by appellate or other proceedings permitted under applicable law, provided that (i) any such contest is made reasonably and in good faith, (ii) Tenant makes provisions, including, without limitation, posting bond(s) or giving other security, reasonably

acceptable to Landlord to protect Landlord, the Building and the Property from any liability, costs, damages or expenses arising in connection with such alleged violation and failure to cure, (iii) Tenant shall agree to indemnify, defend (with counsel reasonably acceptable to Landlord) and hold Landlord harmless from and against any and all liability, costs, damages, or expenses arising in connection with such alleged condition and/or violation, (iv) Tenant shall promptly cure any violation in the event that it exhausts all available appeals without success, and (v) Tenant shall certify to Landlord's reasonable satisfaction that Tenant's decision to delay such cure shall not result in any actual or threatened bodily injury or property damage to Landlord, any tenant or occupant of the Building or the Project, or any other person or entity.

- 9.4 Hazardous Substances.
- (a) <u>Definitions</u>. As used in this Lease:

"Environmental Law" means all statutes, laws, rules, regulations, codes, ordinances, authorizations and orders of federal, state and local public authorities pertaining to any Hazardous Substances or to environmental compliance, contamination, cleanup or disclosures of any release or threat of release to the environment, of any Hazardous Substances, including, without limitation, the Toxic Substances Control Act, 15 U.S.C. § 2601, et seq.; the Clean Water Act, 33 U.S.C. § 1251, et seq.; the Clean Air Act, 42 U.S.C. § 7401, et seq.; the Safe Drinking Water Act, 42 U.S.C. § 300f-300j, et seq.; the Federal Water Pollution Control Act, 33 U.S.C. § 1321, et seq.; the Solid Waste Disposal Act, 42 U.S.C. § 6901, et seq.; the Comprehensive Environmental Response, Compensation and Liability Act of 1980, 42 U.S.C. Section 9601 et seq.; the Federal Resource Conservation and Recovery Act, 42 U.S.C. Section 6901 et seq.; the Superfund Amendments and Reauthorization Act of 1986, Public Law No. 99-499 (signed into law October 17, 1986); M.G.L. c.21C; and oil and hazardous materials as defined in M.G.L. c.21E, as any of the same are from time to time amended, and the rules and regulations promulgated thereunder, and any judicial or administrative interpretation thereof, including any judicial or administrative orders or judgments, and all other federal, state and local statutes, laws, rules, regulations, codes, ordinances, standards, guidelines, authorizations and orders regulating the generation, storage, containment or disposal of any Hazardous Substances, including but not limited to those relating to lead paint, radon gas, asbestos, storage and disposal of oil, biological, chemical, radioactive and hazardous wastes, substances and materials, and underground and above-ground oil storage tanks; and any amendments, modifications or supplements of any of the foregoing.

"Hazardous Substances" means, but shall not be limited to, any hazardous substances, hazardous waste, environmental, biological, chemical, radioactive substances, oil, petroleum products and any waste or substance, which because of its quantitative concentration, chemical, biological, radioactive, flammable, explosive, infectious or other characteristics, constitutes or may reasonably be expected to constitute or contribute to a danger or hazard to public health, safety or welfare or to the environment, including without limitation any asbestos (whether or not friable) and any asbestos-containing materials, lead paint, waste oils, solvents and chlorinated oils, polychlorinated biphenyls (PCBs), toxic metals, etchants, pickling and plating wastes, explosives, reactive metals and compounds, pesticides, herbicides, radon gas, urea formaldehyde foam insulation and chemical, biological and radioactive wastes, or any other similar materials that are regulated by any Environmental Law.

(b) Tenant Operations. Tenant shall not generate, produce, bring upon, use, store or treat Hazardous Substances in the Premises, the Building, the Appurtenant Exterior Areas or any other portion of the Project, except and unless (x) the same are for customary office uses (e.g., normal office cleaning supplies and printing supplies) for its operations at the Premises, (y) such use is in compliance with all applicable Legal Requirements, including without limitation Environmental Laws, and in compliance with the terms and conditions of this Lease, and (z) as to any Hazardous Substances, processes, or procedures not then subject to Legal Requirements, such activities are conducted in accordance with standard practices for tenants conducting similar office operations in Comparable Properties, and do not endanger or create a hazard to public health, safety or welfare or to the environment, within the Premises, the Building, or any Appurtenant Exterior Areas, or other areas in the vicinity thereof generally. Furthermore, beginning on the Commencement Date, on an annual basis or upon Landlord's request following the occurrence of any environmental incident, or on no more than one additional occasion during any year if reasonably requested by Landlord's mortgagee(s) in connection with any financing or refinancing of the Property, Tenant shall provide Landlord with a list detailing the types and amounts of all Hazardous Substances being generated, produced, brought upon, used, stored, treated or disposed of by or on behalf of Tenant in or about or on the Premises, Building or Property and, upon Landlord's request, copies of any manifests or other federal, state or municipal filings by Tenant with respect to such Hazardous Substances (redacted to protect confidential information to the extent such redactions are permitted by the applicable federal, state or municipal authorities having jurisdiction over such filings). Tenant agrees to pay the reasonable cost of any environmental inspection or assessment requested by any lender that holds a security interest in the Property or this Lease, or by any insurance carrier, to the extent that such inspection or assessment pertains to any release, reasonable threat of release, contamination, or a loss or damage or determination of condition related to the foregoing in the Premises, the Building, the Appurtenant Exterior Areas, or any other portion of the Project caused by Tenant or any other Tenant Party.

(c) <u>Tenant Environmental Incident</u>. If any transportation to or from, or any storage, use or disposal of Hazardous Substances on or about, the Premises, the Building, or the Appurtenant Exterior Areas by any Tenant Party results in any escape, or release, reasonable threat of release, contamination of the soil or surface or ground water or any loss or damage to person or property (any such event, a "<u>Tenant Environmental Incident</u>"), Tenant agrees to: (a) notify Landlord immediately of the occurrence; (b) after consultation with Landlord, clean up the occurrence in full compliance with all applicable Environmental Laws; and (c) indemnify, save harmless and defend the Indemnitees from and against any and all claims, damages, losses, penalties, costs, expenses and fees (including reasonable attorneys' fees) arising in whole or in part out of such occurrence. In the event of such occurrence, Tenant agrees to cooperate fully with Landlord and provide such documents, affidavits, and information and take such actions as may be requested by Landlord from time to time (1) to comply with any Environmental Law or Legal Requirement, (2) to comply with any reasonable request of any mortgagee, insurer, or other affected owner or tenant in the Project, and/or (3) for any other reason deemed reasonably necessary by Landlord in a manner consistent with standard practices for Comparable Properties. In the event of any such occurrence that is required to be reported to a governmental authority under any Environmental Law or Legal Requirement, Tenant shall simultaneously deliver to Landlord copies of any notices given or received by Tenant and shall promptly pay when due any fine or assessment against Landlord, Tenant, or the Premises, the Building, or other portions of the Project relating to such occurrence.

- (d) Environmental Reports. Tenant acknowledges that it has received and reviewed certain environmental reports listed on Exhibit J (the "Environmental Reports") regarding the condition of the Project Land. Landlord represents to Tenant that, as of the Lease Date, to the best of Landlord's knowledge and except as set forth in the Environmental Reports, the Building Site does not contain any material amounts of Hazardous Substances that would constitute a violation of applicable Environmental Laws or adversely affect in any material way Tenant's use and occupancy of the Premises, the Building, or the Building Site under this Lease. Landlord shall indemnify Tenant from and against any breach by Landlord of such representation.
- (e) Environmental Insurance. At no cost to Tenant, Landlord shall cause Tenant to be named as an additional insured on the Pollution Legal Liability Select Policy number PLS 16850034, issued by Lexington Insurance Company, that is currently in effect as of the Lease Date through April 20, 2014, with respect to the following Insured Properties: 360 First Avenue, 410 First Avenue, 66B Street and 37 A Street, all in Needham, MA (the "Policy"). If the Policy is extended beyond its current expiration date, or if Landlord is from time to time named as an insured under any new environmental insurance policy covering all or any part of the Project Land, then at Tenant's request Landlord shall use reasonable efforts to cause Tenant to be named as an additional insured on such extended Policy or new policy, as the case may be, if the same can be done at no cost to Tenant (or, if the same can be done at an additional cost, at Tenant's request and expense), in each case subject to any customary exclusions (if any) arising from Tenant's use of or conduct at the Building, provided that this sentence shall not give rise to any obligation of Landlord or any of its affiliates or any other party to obtain, maintain, extend, or replace any such environmental insurance policy from time to time during the Term.
 - (f) General. The provisions of this Section 9.4 shall survive the expiration or earlier termination of this Lease.
- 9.5 <u>Signage</u>. Landlord shall install, or cause to be installed, standard directional signage within the Project Common Facilities to direct employees, visitors, and other business invitees of Tenant and other Project occupants to the appropriate building within the Project. Such initial standard directional signage for Tenant shall be installed at Landlord's expense. During the Term, the general costs of maintenance and repair of such signage shall be included in Project Common Facilities Expenses, with any changes in such signage requested or required by Tenant being allocated to the Building. Tenant shall be responsible, at its sole cost and expense, for installing (a) Tenant's directional signage in the lobby and other areas of the Building and (b) as provided in <u>Section 2.2(c)</u>, Tenant's exterior Building signage. Except as set forth in <u>Section 2.2(c)</u>, Tenant shall not install any other exterior signage on the Building, the Building Site, or the Project Land.
- 9.6 <u>Landlord's Access</u>. Landlord or its agents may enter the Building and the Premises at all reasonable times (i) to perform the service, maintenance, and repair obligations of Landlord hereunder, (ii) to inspect and monitor Tenant's compliance with the terms of this Lease

and applicable Legal Requirements, (iii) for purposes described in Sections 2.3, 9.4, 10.3 and/or 10.4, as applicable, (iv) to show the Premises to prospective and actual buyers, investors, and lenders at any time or, in the last two years of the Term, to prospective tenants; or (v) for any other purpose Landlord reasonably deems necessary in connection with the exercise of Landlord's rights and obligations under this Lease. Except in cases of regularly scheduled services provided under clause (i) or in cases of emergency, any entry under this Section 9.6 shall require at least two (2) business days' prior notice (which notice may be made orally or by email to Tenant's Designated Representative (as defined below)) and shall be made during normal business hours (unless otherwise scheduled by mutual agreement). Landlord shall cooperate with Tenant to schedule any such entry and activity at a time designed to reduce any inconvenience to Tenant's business operations in the Premises. Tenant shall have the right to have a representative of Tenant accompany Landlord during any such entry, but entry shall not be prohibited if Tenant fails to provide an accompanying representative. However, in case of emergency, Landlord may enter any part of the Premises or the Building with such notice as is reasonably practicable or without prior notice if notice is impracticable and without Tenant's representative, if necessary, and shall, if no notice was provided (Landlord agreeing that it shall endeavor to provide an oral or e-mail notice to Tenant's Designated Representative as provided above), promptly notify Tenant of the nature and extent of such entry. During Landlord's access to the Premises, Landlord shall comply with reasonable security provisions required by Tenant to preserve the confidential nature of information in whatever form maintained within the Premises. For safety, security, confidentiality or compliance with law purposes, Tenant may designate certain limited areas as limited access areas to be shown on plans provided by Tenant to Landlord (as updated by Tenant as reasonably necessary in the future) to which Landlord and related parties shall not have access except in an emergency or as otherwise reasonably necessary and then only in accordance with a mutually agreed-upon plan to protect Tenant's reasonable concerns regarding safety, security and confidentiality, provided that such limited access areas shall be reasonably identified and necessary to protect the health of persons or security of confidential and proprietary information. Landlord and Tenant will develop a protocol limiting and controlling the distribution of Landlord's keys or other access devices to the Premises. "Tenant's Designated Representative" shall mean (a) a person with an office at the Premises identified by Tenant in writing to Landlord from time to time as the primary point of contact for Landlord's access to the Premises and (b) the on-site supervisor of Tenant's private security, if any, that is then on duty. Tenant shall provide Landlord with a phone number for Tenant's Designated Representative with any notice designating such person, and any change in the identification of Tenant's Designated Representative shall take effect two (2) business days following delivery of such notice to Landlord and its property manager.

9.7 <u>Security</u>. Tenant shall be solely responsible, at Tenant's sole cost and expense, to provide any security measures that Tenant deems necessary, appropriate, or desirable for the Building and the Building Site, which may include, without limitation, Tenant's security personnel, lobby attendants, and/or security access card readers, security video cameras, and other security equipment installed by Tenant from time to time (whether in connection with the initial TI Work or subsequent Tenant Work during the Term) in or around entrances, exits, lobbies, and loading docks for the Building, on or within different floors of the Premises, and/or elsewhere on the Building Site. Tenant shall have reasonable access to portions of the Building or the Building Site outside the Premises to install and operate any such security measures, subject to Landlord's reasonable approval in accordance with Exhibit B or Article 10, as

applicable, provided that such security measures shall not restrict or impede access through Project Common Facilities serving other parts of the Project. Tenant shall provide Landlord with a written description of its security plan from time to time, outlining Tenant's security measures to the extent applicable to visitors, guests, and others entitled to access the Premises (Tenant being permitted to redact from such security plan any Confidential Information, as defined in Section 16.14). Tenant's security plan shall include the designation of a person or persons (as a Tenant's Designated Representative under Section 9.6) who shall be on the Premises 24 hours, 7 days a week, but only to the extent required for the purposes of fulfilling municipal fire command obligations (Landlord acknowledging that such person may be a third-party contractor or designee thereof).

The parties acknowledge that the operation of the Project Common Facilities may from time to time (but shall not be required to) involve security personnel, equipment, and/or procedures, the costs of which (to the extent provided and generally applicable to the Project as opposed to a particular building within the Project) shall be included in Project Common Facilities Expenses. Notwithstanding the fact that Landlord, the Project Declarant, or the condominium association may from time to time provide such services during the Term, to the maximum extent permitted by applicable law, neither Landlord nor such other parties shall be deemed to owe Tenant, or any person claiming by, through or under Tenant, any special duty or standard of care as a result of such provision of such security services or be responsible for the efficacy of any such security measures.

ARTICLE 10. CONDITION AND MAINTENANCE OF PREMISES AND PROPERTY

- 10.1 <u>Condition of Premises and Property.</u> Tenant acknowledges that, except for any express representations in this Lease, neither Landlord nor any person acting under Landlord has made any representation as to the condition of the Property or the suitability of the Property for Tenant's intended use. Tenant represents and warrants that Tenant has made its own inspection and inquiry regarding the Property and is not relying on any representations of Landlord or any Broker or persons acting under either of them except for any express representations in this Lease.
- 10.2 <u>Tenant Property.</u> Tenant shall insure its personal property, Tenant's FF&E (as defined in <u>Exhibit B</u>), and all other Tenant Property under a "Special Form" (as defined by the insurance industry) policy in accordance with <u>Section 7.1</u>. Landlord shall not be liable for any damage or injury to such Tenant Property or to other person, property or business (including loss of revenue, profits or data) of any Tenant Party, to the maximum extent permitted by law and subject to the provisions of <u>Section 7.3</u> and limitations on liability set forth in <u>Section 16.4</u>, provided that nothing in this sentence shall be construed to limit Tenant's express remedies pursuant to <u>Sections 6.1(e)</u> and <u>12.1</u> of this Lease. Tenant Property expressly includes all business fixtures and equipment, including without limitation any security or access control systems installed for the Premises, filing cabinets and racks, removable cubicles and partitions, kitchen and cafeteria equipment, computers and related equipment, raised flooring, supplemental cooling equipment, audiovisual and telecommunications equipment, non-building standard signage, and other tenant equipment installations, in each case including related conduits, cabling, and brackets or mounting components therefor and any connectors to base building

systems and in each case whether installed, bolted, or otherwise affixed in or about the Premises, in building core areas, or elsewhere in the Project. Except as otherwise expressly provided in this Lease, the exemption hereunder shall apply whether such damage or injury to Tenant Property is caused by (among other things): (i) fire, steam, electricity, water, gas, sewage, sewer gas or odors, snow, ice, frost or rain; (ii) the breakage, leaking, obstruction or other defects of pipes, faucets, sprinklers, wires, appliances, plumbing, windows, air conditioning or lighting fixtures or any other cause; (iii) any other casualty or any Taking; (iv) theft; (v) conditions in or about the Property or from other sources or places; or (vi) any act or omission of any other Project tenant or occupant or their respective employees, visitors, or business invitees.

10.3 Landlord's Obligations for Building Services, Repair and Maintenance.

- (a) <u>Building Services</u>. During the Term, subject to the provisions of this <u>Section 10.3</u> below, Landlord shall provide or cause to be provided the general office cleaning, trash removal, exterior window washing, general base building systems maintenance, exterior landscaping and snowplowing services, and other general services for the Building, in each case as initially set forth on the preliminary list of services set forth on <u>Exhibit E</u> or as subsequently adjusted from time to time during the Term pursuant to the following provisions. For purposes hereof, the services for the Premises, the Building, and the Building Site are referred to herein as the "<u>Building Services</u>" (which term, for avoidance of doubt, does not include the Project Common Facilities Services), and the services provided under the Project Condominium Documents for the Project Common Facilities (including Limited Common Elements serving the Building but not on the Building Site) are referred to herein as the "<u>Project Common Facilities Services</u>." The costs of all Building Services provided to the Building or its appurtenant areas by or through Landlord shall be included in Operating Expenses in accordance with and subject to <u>Article 8</u> and other provisions of this Lease. The allocable share of the costs of all Project Common Facilities Services provided to the Building or its appurtenant areas by or through Landlord or the condominium association, as the case may be, shall be included in Operating Expenses in accordance with and subject to <u>Article 8</u> and other provisions of this Lease.
 - (i) <u>Selection of Service Contractors</u>. At least ninety (90) days prior to the Term Commencement Date, at the request of either party, the parties (or their designated property management representatives) shall meet to review the scope of initial Building Services set forth on <u>Exhibit E</u>, including any adjustments thereto that may be necessary, appropriate, or desirable in relation to the final design, fit-out, and expected initial Tenant operations in the Premises, and to confirm the adjusted scope of services initially to be provided to Landlord under this <u>Section 10.3</u>. Landlord shall solicit, or cause its designated property manager to solicit, bids from at least three (3) qualified, reputable service contractors for each of the separate Building Services (e.g., cleaning services, trash removal services, HVAC maintenance, elevator maintenance, and the like) for which the annual cost is expected to exceed \$25,000 for the Building. All such service contractors shall be experienced in providing the applicable service in comparable first-class suburban office buildings and provide a level and quality of services consistent with the operation and maintenance of a first-class suburban office building, without creating labor disharmony, without impairing applicable warranties for the Building's base building systems, roof, or exterior shell, and otherwise in compliance with the terms of this Lease (collectively, the "<u>First-Class Service Standard</u>") and mutually approved by

Landlord and Tenant. After soliciting bids from such service contractors, Landlord and Tenant shall review the bids and Tenant shall promptly select from such bidders the service contractor for the applicable service in question. After such a service contractor has been selected for the applicable Building Service pursuant to the foregoing process, Landlord or its property manager may continue to engage such service contractor from year to year, subject to replacement at Landlord's initiative at any time (for cause or otherwise) or (if requested by Tenant at least ninety (90) days prior to the end of a calendar year) to re-bidding and/or replacement, in which event Landlord shall cause its property manager to solicit bids from applicable service contractors in the manner set forth above. For avoidance of doubt, the parties acknowledge that Landlord intends to engage the Management Company (as defined in Section 1.14) or its affiliate to provide the initial property management services for the Building, that any replacement property management company designated by Landlord shall be a reputable third party property management company experienced in the management of comparable first-class suburban office buildings, and that Landlord shall not be obligated to solicit bids for property management services under the foregoing provisions.

(ii) Periodic Review and Adjustment of Tenant's Desired Services. At the request of either party, either on an annual basis or otherwise as from time to time reasonably requested, the parties (or their designated property management representatives) shall meet to evaluate the renewal and/or potential replacement of applicable service contractors under subparagraph (i) above and to review the particular scope of Building Services to be provided by Landlord for the ensuing calendar year of the Term, in order to consider reasonable adjustments to such Building Services as may be necessary, appropriate, or desirable in connection with Tenant's business operations. The implementation of any such replacements of service contractors or any such adjustments to Building Services shall require reasonable prior notice, including without limitation with respect to services being provided under applicable annual service contracts. The resulting costs of any such service contractor replacements or Building Service adjustments (whether increased or decreased) shall be included in Operating Expenses in accordance with Article 8. Tenant may similarly request changes to the level of Project Common Facilities Services from time to time, provided that (A) for any Project Common Facilities that then exclusively serve the Building or are Limited Common Elements of only the Building (e.g., initially the first phase of Garage B and at all times the Amphitheater Area), the resulting costs of any such Project Common Facilities Services adjustments (whether increased or decreased) shall be allocated to the Building and included in Operating Expenses in accordance with Article 8 and (B) for any Project Common Facilities Services that do not then exclusively serve the Building (e.g., shared parking areas, common roadways, or other Project Common Facilities), any increased costs thereof shall be allocated (in proportion to their respective gross square footages or, in the case of parking facilities, the number of allocated parking spaces therein, unless otherwise mutually agreed) to the Building and any other Project building that requested such increased service level for such shared facilities, and any decreased costs of reduced services shall be allocated among the Building and all other Project building(s) served by such Project Common Facilities. Nothing in this subparagraph (ii) shall be construed to require Landlord to reduce any Building Services to a level that Landlord determines would not satisfy the First-Class Service Standard or to cause any

Project Common Facilities Services to be reduced to a level that Project Declarant (or, as applicable, the condominium association then administering the applicable Project Common Facilities) determines would not satisfy the First-Class Service Standard.

- (iii) Selected Tenant-Provided Services to the Premises. In connection with such periodic reviews, on Tenant's reasonable prior notice to Landlord (which may be required to avoid cancellation costs for certain annual service contracts), Tenant may from time elect to use its own personnel or engage its own service contractors, at Tenant's sole cost and expense, to provide certain services that are provided exclusively to the Premises (which term, for avoidance of doubt, does not include base building systems or equipment or the roof or exterior shell of the Building) and do not affect base building systems or equipment, the roof or exterior shell of the Building, or Project Common Facilities, provided that the level and quality of such services is consistent with the First-Class Service Standard. Such selected services to the Premises under this subparagraph (iii) may from time to time include, for example, one or more of the following: interior office cleaning services, day porter service, restroom cleaning and supplies, interior glass cleaning, carpet and tile cleaning, recycling and trash removal, general maintenance and repairs of tenant improvements and other tenant installations, light bulb replacements, touch-up painting, and the like. To the extent that, pursuant to the foregoing procedures, Tenant should elect to provide any such services for the Premises under this subparagraph (iii), then Tenant shall provide such services in accordance with the First-Class Service Standard pursuant to service plans and service contracts reasonably approved by Landlord, provide Landlord with evidence of appropriate insurance for the service contractors engaged by Tenant, coordinate such Tenant-provided service activities with Landlord's property management staff, and provide such regular periodic reports as Landlord may reasonably require to ensure that the Premises are being maintained and serviced by Tenant in accordance with the First-Class Service Standard. Any such Tenant rights to provide services for the Premises shall not apply to the Project Common Facilities (which are operated and maintained under the Project Condominium Documents) or to property management services (which are addressed in subparagraph (v) below). For avoidance of doubt, Tenant shall pay directly to the applicable service contractors from time to time engaged by Tenant for its self-provided services under subparagraph (iii) all amounts due to such parties, and such amounts paid directly by Tenant shall not be included in Operating Expenses.
- (iv) <u>First-Class Service Standard</u>. If Landlord shall fail to provide Building Services (other than those provided by Tenant under subparagraph (iii) above) in a manner consistent with the First-Class Service Standard, then Tenant shall from time to time give Landlord a written notice advising Landlord of such deficiency, the parties (or their designated property management representatives) shall meet to review the nature and extent of the deficiency, and Landlord shall promptly restore the applicable service to the First-Class Service Standard. Such remedial actions by Landlord may include, without limitation, correction of a particular deficiency at issue, replacement of service contractors or personnel in the case of persistent failure to satisfy the First-Class Service Standard, or other appropriate corrective measures, taking into account the nature and extent of the deficiency. For avoidance of doubt, Landlord shall not be responsible under this Section 10.3 for any deficiencies that may arise in whole or in part from any damage

or abuse by Tenant or any Tenant Party, any overloading of base building systems due to Tenant's particular manner of use, any inadequate sizing or capacity of Tenant's HVAC distribution system within the Premises, any supplemental cooling equipment or other equipment or installations installed by Tenant, any defects in any Tenant Work, Tenant's FF&E, or other items installed by Tenant, or any such matters covered by any applicable warranty held by Tenant for any such items installed by Tenant.

(v) Self-Management of Building Services. If, prior to the end of the first three (3) Lease Years, Tenant reasonably determines that the failure by Landlord to provide Building Services (other than those provided by Tenant under subparagraph (iii) above) in accordance with the First-Class Service Standard has continued for more than ninety (90) days after notice from Tenant without Landlord undertaking and implementing appropriate remedial actions that restore such services to the First-Class Service Standard (either at Landlord's own direction through its property management company or by retaining a replacement property management company meeting the First-Class Service Standard), then Tenant may elect to require Landlord to terminate its existing property management company on not less than thirty (30) days' notice to Landlord. In addition, after the first three (3) Lease Years, either Tenant or Landlord may elect, with or without cause, on not less than ninety (90) days' prior written notice to the other party, for Tenant to provide all Building Services hereunder. In the event that Landlord's property manager is terminated under this subparagraph (v), Tenant shall engage, at its sole cost and expense, a reputable third party property management company that is experienced in the management of comparable first-class suburban office buildings, subject to Landlord's reasonable approval, who shall provide all such Building Services in accordance with the First-Class Service Standard. Any such property management company engaged by Tenant shall perform such Building Services pursuant to service plans and service contracts reasonably approved by Landlord, provide Landlord with evidence of appropriate insurance for the property management company engaged by Tenant and for the service contractors engaged thereunder, coordinate such Tenant-provided service activities with Landlord's property management staff, and provide such regular periodic reports as Landlord may reasonably require (such as reports of the type described under Section 10.4(c) for equipment servicing) to ensure that the Premises, the Building's base building systems, and the Building Site are being maintained and serviced by Tenant in accordance with the First-Class Service Standard, together with immediate reporting of any incidents or conditions adversely affecting any base building systems or equipment, structural or building shell elements (including the roof), or any exterior elements of the Building or the Building Site. Notwithstanding the foregoing, Landlord may from time to time, upon reasonable prior notice to Tenant, elect to retain the maintenance responsibility for the Building's roof and exterior shell. Any such Tenant rights to engage the property management company under this subparagraph (v) shall not apply to the Project Common Facilities, which are operated and maintained under the Project Condominium Documents. For any period in which Tenant provides Building Services under this subparagraph (v), Tenant shall pay all amounts due directly to the applicable property management company and service vendors from time to time engaged by Tenant (which amounts shall not be included in Operating Expenses), and the Property Management Fee charged by Landlord as part of the Operating Expenses under Paragraph 1(a)(17) of Exhibit E-1 shall be reduced to three

quarters of one percent (0.75%) of the Base Rent, <u>provided that</u> if any such termination of Landlord's property management services occurs during the period between the end of the third (3rd) Lease Year and the third (3rd) anniversary of the Partial Rent Commencement Date, then the Property Management Fee for the period during the period from such termination until the third (3rd) anniversary of the Partial Rent Commencement Date shall continue to be charged to Tenant during such period at the monthly rate (i.e., 2.5% of the monthly Base Rent) set forth in Paragraph 1(a)(17) of <u>Exhibit E-1</u> and pro-rated on a per diem basis during such period. The parties acknowledge that, notwithstanding any such provision of Building Services by Tenant hereunder, Landlord shall continue to provide services for administering Building insurance, real estate taxes, (if applicable) roof and exterior shell, supervisory Building oversight, accounting, coordination of Project Common Facilities services, and other matters hereunder.

(vi) General. Any such election by Tenant to use its own personnel or service contractors to provide certain Building Services as provided in subparagraph (iii) above, or any such engagement by Tenant of a property management company for Building Services under subparagraph (v) above, as the case may be, in accordance with the foregoing provisions shall continue from year to year unless otherwise agreed by the parties. If Tenant shall fail to provide such Building Services in a manner consistent with the First-Class Service Standard, then Landlord may from time to time (but shall have no obligation to) give Tenant a written notice advising Tenant of such deficiency, the parties (or their designated property management representatives) shall meet to review the nature and extent of the deficiency, and Tenant shall promptly restore the applicable service to the First-Class Service Standard. Such remedial actions by Tenant may include, without limitation, correction of a particular deficiency at issue, replacement of service contractors or personnel in the case of persistent failure to satisfy the First-Class Service Standard, or other appropriate corrective measures, taking into account the nature and extent of the deficiency and its potential adverse impact on the Building or applicable building systems (provided such prior notice shall not be required in cases of emergency or other conditions posing a risk of injury to persons or damage to property), in which event Landlord reserves the right, but shall not be obligated, to implement such curative measures as may be reasonably required to remedy such deficiency in accordance with the procedures set forth in Section 10.4(d). If, notwithstanding such efforts, Landlord reasonably determines that the failure by Tenant to provide such Building Services in accordance with the First-Class Service Standard has continued for more than ninety (90) days after notice from Landlord without Tenant restoring such services to the First Class Service Standard (either at Tenant's own direction through its existing property management company or by retaining a replacement property management company meeting the First-Class Service Standard), then Landlord may require Tenant to terminate its existing property management company on not less than thirty (30) days' notice to Tenant, in which event Landlord shall resume providing the Building Services in accordance with Section 10.3(a) above. If Tenant, having made the election to self-provide or "self-manage" the applicable services hereunder for the Building or the Premises, as the case may be, subsequently requests that Landlord resume providing such services, any such service resumption shall require such reasonable prior notice as Landlord may require to engage appropriate personnel or contractors, and Landlord may

require that any work (if any) required to restore the base building systems and equipment and other elements of Base Building Work to good working order and repair shall be performed, at Tenant's cost, either prior to or pursuant to Landlord's resumption of services.

(b) Repair and Maintenance. Subject to the provisions of Sections 10.3 and 10.4 and Article 12, and except for damage caused by any act or omission of Tenant or any other Tenant Party, Landlord shall make such repairs and replacements to the roof structure and roof membrane; exterior walls; floor slabs, footings, foundations, columns, and other structural components installed as part of the Base Building Work; glass in exterior windows and exterior doors of the Building installed as part of the Base Building Work; and other Base Building Work systems (provided that Tenant shall be responsible for maintaining the utilities distribution and mechanical systems installed by Tenant, whether in the Premises or elsewhere in or about the Building, beyond the applicable Utility Switching Points or (as applicable) beyond the base building equipment and electrical panels installed by Landlord as part of the Base Building Work) as may be necessary to properly maintain them in good repair and condition. Landlord shall have no obligation to repair or maintain any portion of the Premises or perform any service, except as specifically set forth in this Section 10.3. Tenant shall promptly report in writing to Landlord any defective condition known to it that Landlord is required to repair. If, after receiving such report, any failure by Landlord to make necessary repairs unreasonably interferes with Tenant's use or occupancy of the Premises, then Tenant may, upon reasonable prior notice to Landlord (which shall be not less than thirty (30) days, unless the condition poses a substantial risk to person or property, in which event Tenant shall provide such reasonable prior notice to Landlord as is reasonably possible in the circumstances) if Landlord has not then commenced or undertaken to commence such repairs, perform such repairs at Tenant's expense, provided that Tenant shall comply with all requirements under Section 10.3 that would have been applicable to such repair work if Tenant had elected to provide such services or self-manage the Premises as provided in Section 10.3(a). If and to the extent reasonably required for Landlord to carry out its maintenance or service obligations hereunder, Tenant shall assign, make available to Landlord, or use reasonable efforts to enforce (but without any obligation to undertake applicable enforcement proceedings) any applicable warranties held by Tenant or any Tenant Party. Tenant waives the benefit of any present or future law that provides Tenant the right to repair the Premises or Property at Landlord's expense or to terminate this Lease because of the condition of the Property or Premises. Notwithstanding anything to the contrary in this Lease, Landlord shall have no liability or responsibility for the storage, containment or disposal of any Hazardous Substances generated, stored or contained by Tenant or any other Tenant Party in or about the Building or the Building Site, and Tenant hereby agrees to store, contain and dispose of any and all such Hazardous Substances at Tenant's sole cost and expense in accordance with the provisions of Article 9.

10.4 Tenant's Obligations for Repair and Maintenance.

(a) <u>General Obligations</u>. Except for work that <u>Section 10.3</u> or <u>Article 12</u> requires Landlord to perform, Tenant at its sole cost and expense (i) shall keep the Premises, including all TI Work, Tenant Work, and Tenant Property, initially or thereafter in or about the Premises or on the Building Site, in good order, condition and repair, in compliance with all Legal Requirements, and substantially in the condition the same were in upon completion of the TI

Work (or subsequent TI Work), normal wear and tear, casualty and condemnation (to the extent the repair is the responsibility of Landlord pursuant to Article 12 hereof) excepted, (ii) shall keep in a safe, secure and sanitary condition all trash and rubbish temporarily stored at the Premises (prior to its removal by Landlord's trash removal service if applicable under Section 10.3), and (iii) shall make all repairs and replacements and do all other work necessary for the foregoing purposes whether the same may be ordinary or extraordinary, foreseen or unforeseen. The foregoing obligations shall not apply to defects in Base Building Work or items covered under any then applicable warranties held by Landlord for Base Building Work, but shall include without limitation Tenant's obligation to repair, maintain, and replace floors and floor coverings, to paint and repair walls and doors, to replace and repair all glass in windows and doors of the Buildings (except glass in the exterior walls of the Buildings and in exterior doors, unless damage thereto is caused by Tenant or any other Tenant Party), ceiling tiles, lights and light fixtures, pipes, conduits, wires, drains and the like in the Premises and to make as and when needed as a result of misuse by, or neglect or improper conduct of Tenant or any Tenant Party or otherwise, all repairs necessary, which repairs and replacements shall be in quality and class equal to the original work. If and to the extent reasonably required for Tenant to carry out its maintenance or service obligations hereunder, Landlord shall assign, make available to Tenant, or use reasonable efforts to enforce (but without any obligation to undertake applicable enforcement proceedings) any applicable warranties held by Landlord for Base Building Work in the Building. If anything required pursuant to this Section 10.4 to be repaired cannot be fully repaired or restored, Tenant upon prior notice to Landlord shall replace it at Tenant's cost, in accordance with the requirements of Section 10.5.

(b) <u>Cafeteria/Food Service Operations</u>. Without limiting the generality of the foregoing, Tenant shall be solely responsible, at its sole cost and expense, for installing, operating, cleaning, maintaining, and repairing such employee cafeteria, café, and lunch rooms as may from time to time be installed by Tenant in the Premises, in a good and clean condition, in compliance with applicable Legal Requirements, and otherwise in accordance with the terms of this Lease. Such obligations shall include, without limitation, general cleaning services for such areas, proper storage of food and beverage products, pest control measures, and periodic cleaning of grease traps and ventilation equipment serving such facilities (whether such equipment is located in the Premises or elsewhere in or about the Building), in each case in accordance with applicable Legal Requirements, insurance requirements, and prudent operational standards for a first-class suburban office building, as reasonably approved by Landlord from time to time, taking into account the particular nature of Tenant's cooking and food service operations in the Building.

(c) <u>Supplemental HVAC</u> and <u>Other Systems</u>. If Tenant shall from time to time install (or at Tenant's request, cause Landlord to install) supplemental HVAC or similar equipment (or, pursuant to the provisions of <u>Section 10.3</u> above, undertake maintenance of any base building equipment serving the Premises), Tenant shall secure, pay for, and keep in force third-party maintenance and service contracts with appropriate and reputable service companies approved by Landlord (such approval not to be unreasonably withheld, conditioned or delayed) providing for the regular maintenance of all such equipment or systems that landlords of Comparable Properties typically service by use of third-party service companies (collectively, the "<u>Service Contracts</u>"), copies of which shall be provided to Landlord, and Tenant shall provide to Landlord in a timely manner such periodic inspection reports (but no less frequently

than annually) as are prepared by the service providers under the Service Contracts. Without limitation, Tenant shall be responsible for all supplemental heating, ventilating and air-conditioning systems installed in connection with the initial TI Work or as an upgrade to Base Building Work.

- (d) Non-Performance. If Tenant shall fail to perform any of its obligations under Section 10.4, Landlord may provide notice of such failure to Tenant. In the event that Tenant fails to perform or fails to commence and thereafter diligently perform its obligations within ten (10) business days following such notice, then (on not less than a further three (3) days' notice to Tenant of such failure) Landlord may (but shall not be obligated to) perform such maintenance, repair or replacement on Tenant's behalf. Notwithstanding the foregoing, in the case of emergency, Landlord may (but shall not be obligated to) undertake such work without prior notice to Tenant and promptly advise Tenant of such work. Tenant shall reimburse Landlord for all costs reasonably incurred by Landlord under this Section 10.4(d), plus an administrative charge of ten percent (10%) of such costs, within thirty (30) days following invoice from Landlord.
- 10.2. <u>Tenant Work</u>. The terms and provisions of the Work Letter in <u>Exhibit B</u> (and not this <u>Section 10.5</u>) shall apply to Tenant's design and construction of the initial TI Work. The terms and provisions of this <u>Section 10.5</u> below (and not <u>Exhibit B</u>) shall apply to any Tenant Work performed after the initial TI Work is completed under <u>Exhibit B</u>.

As used in this Lease, "<u>Tenant Work</u>" shall mean all tenant improvement work performed by or through Tenant in or at the Building or the Building Site, including without limitation (i) the initial TI Work under <u>Exhibit B</u> and (ii) all future work, including demolition, improvements, additions and alterations, in or to the Premises, but excluding the Landlord's Base Building Work under <u>Exhibit B</u>. Without limitation, Tenant Work includes any penetrations in the walls, partitions, ceilings or floors and all attached carpeting. Tenant Work shall not, however, include Tenant Property, but the provisions of <u>Exhibit B</u> or this <u>Section 10.5</u> below (as the case may be) shall apply to Tenant's design, construction, and installation work for all Tenant Property to be installed, attached, or affixed by Tenant in or about the Premises, the Building, or the Building Site. All Tenant Work shall be subject to Landlord's prior written approval and arranged and paid for by Tenant, all as provided in and subject to the provisions of <u>Exhibit B</u> (with respect to the initial TI Work) or this <u>Section 10.5</u> (as to future work), as the case may be.

(a) General Provisions.

(i) <u>Submission of Construction Documents</u>. Tenant shall submit Construction Documents (as defined below) for the proposed Tenant Work requiring Landlord's approval under this <u>Section 10.5</u>. Landlord shall review, comment upon (if desired), and approve or disapprove such plans by written notice in sufficient detail for Tenant to be able to reply, within ten (10) business days following the delivery of such plans to Landlord after Landlord's actual receipt of such request, provided that such review period shall be reasonably extended as the special nature of the proposed Tenant Work in question may reasonably require (e.g., for structural or exterior work or work affecting base building systems). Landlord shall not unreasonably withhold, condition or

delay Landlord's approval of Tenant Work, but Landlord's disapproval of proposed Tenant Work shall not be unreasonable where, in Landlord's reasonable judgment, such proposed Tenant Work (A) adversely affects any structural component of the Building, (B) would be incompatible with the fire-safety, telecommunications, electrical, mechanical, or plumbing systems of the Building ("Core Building Systems"), (C) affects the exterior or the exterior appearance of the Building or common areas within or around the Building or other property than the Premises (except as expressly permitted under the terms of the Lease), or (D) requires unusual expense to readapt the Premises for general office purposes. Landlord shall cooperate with Tenant, at no cost and liability to Landlord, to execute any permit applications requiring execution by the Building owner in connection with Tenant Work. If Landlord shall timely fail to approve, disapprove, or provide comments on the Construction Documents within the time period set forth above, and if such failure by Landlord continues for five (5) business days after Landlord's receipt of Tenant's notice of such failure, which notice shall bear the legend "NOTICE FOR LANDLORD'S FAILURE TO APPROVE CONSTRUCTION DOCUMENTS UNDER SECTION 10.5(A) OF THE LEASE; FAILURE TO RESPOND WITHIN FIVE (5) BUSINESS DAYS SHALL BE DEEMED AN APPROVAL", then Landlord shall be deemed to have approved the Construction Documents so submitted for the Tenant Work shown thereon, and Tenant may proceed with the performance of such Tenant Work in accordance with the provisions of Section 10.5.

- (ii) <u>Minor Alterations</u>. Notwithstanding the foregoing, any interior, non-structural Tenant Work (including any series of related Tenant Work projects) that (A) costs less than \$250,000 in the aggregate (the "<u>Tenant Work Threshold Amount</u>"), (B) does not affect any Core Building Systems, and (C) does not affect any penetrations in or otherwise affect any structural walls, floors, roofs, or other structural elements of the Building or any signs visible from the exterior of the Premises or any change in the exterior appearance of the windows in the Premises (including shades, curtains and the like), shall not require Landlord's prior approval if Tenant delivers the Construction Documents (as defined in <u>Section 10.5(b)</u>) for such work to Landlord at least five (5) business days' prior to commencing such work.
- (iii) Work Affecting HVAC Systems. Prior to commencing any work affecting air disbursement from ventilation systems serving the Premises or the Building, including without limitation the installation of Tenant's exhaust systems, Tenant shall provide to Landlord, if reasonably requested by Landlord in view of the nature of the proposed work in question, a third party report from a consultant, and in a form, reasonably acceptable to Landlord, showing that such work will not adversely affect the ventilation systems of the Building (or of any other tenant in the Building) and Tenant shall, upon completion of such work, be responsible for ensuring that such work does not adversely affect such systems.
- (iv) <u>Identification of Restoration Items</u>. At the time Landlord grants approval of any Tenant Work under this <u>Section 10.5</u>, Landlord shall specify the items of such Tenant Work (if any) that must be removed by Tenant upon the expiration or earlier termination of the Term (which shall be limited to items that are not readily useable for first class office purposes or would require unusual expense to demolish or readapt for

general office purposes). For any Tenant Work for which Landlord's approval is not required (or is performed without obtaining a required approval) under the terms of this Lease, Landlord may designate for removal, prior to the expiration of the Term or otherwise upon Tenant's reasonable prior request, items of Tenant Work that are not readily useable for first class office purposes or would require unusual expense to demolish or readapt for general office purposes. (Paragraph 3.3(d) of the Work Letter acknowledges certain elements of the initial TI Work that are not required to the removed.) The parties acknowledge that, without limiting the generality of the foregoing provisions, Landlord reserves the right (if so specified by Landlord as provided above) to require Tenant to remove, at the expiration or earlier termination of the Term, any stairwell or floor openings not shown on the Preliminary Plans (if any) and any private or executive restroom facilities (if any) that Tenant may install in the Premises during the Term.

(b) Construction Documents. No Tenant Work shall be effected except in accordance with complete, coordinated construction drawings and specifications ("Construction Documents") prepared in accordance with Exhibit B-6. Before commencing any Tenant Work (other than work that does not typically involve construction drawings or specifications) requiring Landlord's approval hereunder, Tenant shall obtain Landlord's prior written approval of the Construction Documents for such work, which approval shall not be unreasonably withheld, conditioned or delayed as provided in Section 10.5(a) above. The Construction Documents shall be prepared by an architect or, where applicable, a qualified engineer (in either case, "Tenant's Architect") registered in the Commonwealth of Massachusetts, experienced in the construction of tenant space improvements in comparable buildings in the area where the Premises are located and, if the value of such Tenant Work will equal or exceed the Tenant Work Threshold Amount or will affect any Core Building Systems or structural components of the Building, the identity of such Tenant's Architect shall be approved by Landlord in advance, such approval not to be unreasonably withheld, conditioned, or delayed. Tenant shall be solely responsible for the liabilities associated with and expenses of all architectural and engineering services relating to Tenant Work and for the adequacy, accuracy, and completeness of the Construction Documents even if approved by Landlord (and even if Tenant's Architect has been otherwise engaged by Landlord in connection with the Building). The Construction Documents shall set forth in detail the requirements for construction of the Tenant Work and shall show all work necessary to complete the Tenant Work including all cutting, fitting, and patching and all connections to the mechanical, electrical, and plumbing systems and components of the Building. Submission of the Construction Documents to Landlord for approval shall be deemed to constitute Tenant's agreement that, except as is specifically and expressly set forth therein, Tenant shall be and remain solely responsible for ensuring that all Tenant Work described in the Construction Documents (i) complies with all applicable Legal Requirements, building codes, and customary industry design standards for first-class suburban office buildings, (ii) does not materially and adversely affect any structural component of the Building, (iii) is compatible with and does not adversely affect the Core Building Systems, (iv) does not affect any property other than the Premises, and (v) conforms to floor loading limits of the Building as specified for the Base Building Work under Exhibit B-2. The Construction Documents shall comply with Landlord's reasonable requirements for the uniform exterior appearance of the Building. Landlord's approval of Construction Documents shall signify only Landlord's consent to the Tenant Work shown and shall not result in any responsibility of Landlord concerning compliance

of the Tenant Work with applicable Legal Requirements or building codes, or coordination or compatibility with any component or system of the Building, or the feasibility of constructing the Tenant Work without damage or harm to the Building, all of which shall be the sole responsibility of Tenant. Tenant shall reimburse Landlord for all reasonable out-of-pocket third-party expenses incurred by Landlord in reviewing the Construction Documents and coordinating and inspecting Tenant's work hereunder, estimates for which Landlord shall provide upon Tenant's request.

- (c) <u>Performance</u>. The identity of any contractor, subcontractor, or other person or entity (including any employee or agent of Tenant) performing or designing any Tenant Work ("Tenant Contractor") shall, if the cost of such work in any instance is in excess of the Tenant Work Threshold Amount or will affect any Core Building Systems or structural components of the Building or involves any work other than interior, nonstructural alterations, be approved pursuant in advance by Landlord, such approval not to be unreasonably withheld or delayed. Tenant shall procure at Tenant's expense all necessary permits and licenses before undertaking any Tenant Work. Tenant shall perform all Tenant Work at Tenant's risk in compliance with all applicable Legal Requirements and in a good and workmanlike manner employing new materials of good quality and producing a result at least equal in quality to the other parts of the Premises. When any Tenant Work is in progress, Tenant shall cause to be maintained insurance as described in the Tenant Work Insurance Schedule attached as Exhibit G and (as indicated, if applicable, in connection with Landlord's approval of the work in question) such other insurance as may be reasonably required by Landlord covering any additional hazards due to such Tenant Work, and (only if a Material Event of Default then exists, an LC Trigger Event has occurred without cure as provided in Section 14.6 below, or Tenant has suffered within the preceding five-(5)-year period any mechanics or similar lien to be filed with respect to any Tenant Work without a timely discharge under Section 10.5(d) below) also such bonds or other assurances of satisfactory completion and payment as Landlord may reasonably require, in each case for the benefit of Landlord. If the Tenant Work in any instance requires Landlord's approval hereunder, Tenant shall reimburse Landlord for Landlord's reasonable out-of-pocket third-party costs of reviewing the Construction Documents and proposed Tenant Work and inspecting installation of the same (estimates for which Landlord shall provide upon Tenant's request), such reimbursement to be made within thirty (30) days after submission by Landlord of invoices for such costs and expenses. At all times while performing Tenant Work, Tenant shall require any Tenant Contractor to comply with all applicable Legal Requirements and the Rules and Regulations (as defined in Section 9.1) relating to such work. For any work that from time to time affects the roof or any penetrations therein, each Tenant Contractor working on the roof of the Building shall coordinate with Landlord's roofing contractor, shall comply with its reasonable requirements, and shall not violate existing roof warranties. Each Tenant Contractor shall work on the Premises without causing delay to or impairing of any guaranties, warranties, or the work of any other contractor.
- (d) <u>Payment</u>. Tenant shall pay the entire cost of all Tenant Work, including without limitation any services provided to Tenant or those claiming by or through Tenant in connection with Tenant Work giving rise to a lien pursuant to the Massachusetts General Laws, so that the Premises, the Property, and the Project Land, including Tenant's leasehold hereunder, shall always be free of liens for labor, materials, or services, or as otherwise provided under such statutes. If any such lien is filed, then Tenant shall promptly (and always within twenty (20)

days after receiving notice of such lien from any source) discharge the same. If Tenant shall fail to timely discharge any such lien as and when required under this Section 10.5(d), Landlord may, but shall have no obligation to, discharge such lien or obtain a bond to remove such lien, in which event Tenant shall reimburse Landlord for all costs incurred by Landlord in connection with such discharge or bond, within ten (10) days after written demand for such reimbursement. The provisions of this Section 10.5(d) shall survive the expiration or earlier termination of this Lease.

- (e) Work Coordination. Tenant shall schedule and coordinate all aspects of Tenant Work with the Landlord's property manager or designated representative. If an operating engineer is required by any union regulations, Tenant shall pay for such engineer. If shutdown of risers and mains for electrical, mechanical and plumbing work is required, such work shall be supervised by Landlord's representative (the reasonable out-of-pocket third party costs of which shall be reimbursed by Tenant). No work shall be performed to portions of Building systems that serve other tenants without Landlord's approval, which approval shall not be unreasonably withheld, conditioned or delayed, and all such work shall be performed under Landlord's supervision. Except in case of emergency, at least two (2) business days' prior notice must be given to the Building management office prior to the shutdown of fire, sprinkler and other alarm systems, and in case of emergency, prompt notice shall be given. In the event that such work unintentionally alerts the Fire or Police Department or any private alarm monitoring company through an alarm signal, Tenant shall be responsible for any fees or charges levied in connection with such alarm. Tenant shall pay to Landlord such charges as may from time to time be in effect with respect to any such shutdown. All demolition, installations, removals or other work that is reasonably likely to inconvenience other tenants of the Building or interfere with Building services provided by Landlord must be scheduled with the Building manager at least twenty-four (24) hours in advance.
- (f) No Labor Disharmony. With respect to any Tenant Work in or around the Building that Tenant may from time to time perform during the Term, Tenant shall be solely responsible for managing, at its expense, any disruption to Tenant's use and occupy of the Building from labor disputes arising from workers employed by any Tenant Contractor and for ensuring that any such labor disputes do not disrupt the use of the Project Common Facilities or other Project buildings. With respect to any construction work in or around the Building (if any) that Landlord may from time to time perform during the Term, Landlord shall be responsible for ensuring that any such work is carried out without disruption to Tenant's use and occupancy of the Premises arising from workers employed by Landlord's contractor.
- (g) Work Close-out Requirements. Upon completion of any Tenant Work, Tenant shall provide to Landlord (i) copies of the permanent certificate of occupancy (if and to the extent, if any, such certificate is required for occupancy after the applicable Tenant Work in question) and any other final governmental approvals and sign-offs required for completion of such work, (ii) a record set of Construction Documents (compiling all approved change orders and bulletins) prepared by Tenant's architect for the Tenant Work in question (in each case, other than for work that does not typically involve construction drawings or specifications), (iii) proof of payment for all labor and materials, including a reconciliation of total construction-related costs, evidence of payments made, and lien waivers from all Tenant Contractors and

other parties who would be entitled to a mechanics lien or similar lien if not paid in full, and (iv) copies of applicable warranties for such Tenant Work (which, as to any such warranty that extends beyond the expiration of the Term and is assignable in accordance with their terms, shall be assigned to Landlord at the expiration or earlier termination of the Term if so requested by Landlord).

10.6 <u>Condition upon Termination</u>. At the expiration or earlier termination of the Term, Tenant (and all persons claiming under or through Tenant) shall, without the necessity of notice or demand, vacate, surrender, and deliver the Premises to Landlord in a broom-clean condition, in compliance with all Legal Requirements, and substantially in the condition the same were in upon completion of the TI Work (or subsequent Tenant Work) installed and maintained by Tenant in accordance with the terms of this Lease, reasonable wear and tear and (subject to the provisions of <u>Article 12</u>) damage by casualty or taking excepted. As part of such delivery, Tenant shall remove all Tenant Property, provide keys (or lock combinations, codes, or electronic passes) to any locks in and to the Premises to Landlord, provide Landlord with copies of any owners' manuals or software required for the operation of equipment or systems remaining in the Premises, and remove any items of Tenant Work (if any) required to be removed under the provisions of Paragraph 3.3(d) of <u>Exhibit B</u> or <u>Section 10.5(a)(iv)</u>, as the case may be. Tenant shall repair all damage that results from the removal of such items and restore the areas affected by such removal. Any property not so removed shall be deemed abandoned, shall at once become the property of Landlord, and may be disposed of in such manner as Landlord shall from time to time determine in its discretion; and Tenant shall pay the cost of removal and disposal to Landlord upon demand to the extent such cost exceeds the value received, if any, from any sale of such property. The covenants of this Section shall survive the expiration or earlier termination of the Term.

ARTICLE 11. ROOFTOP EQUIPMENT

11.1 Rooftop Installations. Subject to Legal Requirements, Tenant shall have the exclusive right, as appurtenant to its Lease of the Premises and without a separate rental charge, but otherwise subject to the terms and conditions of this Lease, to install (in accordance with Exhibit B or thereafter Section 10.5) and thereafter during the Term to operate, maintain, repair, replace, upgrade and remove, solely for accessory use to Tenant's business operations within the Premises, (a) a satellite dish or other rooftop telecommunications devices and (b) supplemental HVAC equipment (collectively and in each case including associated wires, cabling, brackets, supports, and conduits located on the roof or in other portions of Building to connect such rooftop equipment to the applicable portions of the Premises, the "Rooftop Equipment") in locations on the roof of the Building reasonably approved by Landlord and (as to connecting wiring, cabling and other equipment) in Building shafts and conduits pursuant to plans approved by Landlord under Exhibit B or Section 10.5, as the case may be. Notwithstanding the foregoing provisions, Tenant acknowledges that certain base building equipment or other equipment serving the Building or Appurtenant Exterior Areas (such as security cameras) may also from time to time be located on the roof. Any such rooftop equipment on the Building that Landlord from time to time desires to install to serve Appurtenant Exterior Areas of the Project (if any) shall be subject to Tenant's prior approval, which shall not be unreasonably, conditioned, or delayed. Landlord shall not grant rooftop rights on the Building to any third party.

11.2 <u>Installation and Maintenance of Rooftop Equipment</u>. Tenant shall install the Rooftop Equipment at its sole cost and expense (except as otherwise provided for the TI Allowance in <u>Exhibit B</u>), at such times and in such manner as Landlord may reasonably designate and in accordance with all of the applicable provisions of this Lease regarding Tenant Work. Tenant shall not install or operate the Rooftop Equipment until it receives prior written approval of the Construction Documents in accordance with <u>Exhibit B</u> or <u>Section 10.5</u>, as the case may be, which Construction Documents shall include plans and specifications showing the proposed location, height, dimensions, materials, and other technical specifications, together with such appropriate screening or visual barriers as Landlord may reasonably deem appropriate. Such Construction Documents shall also be sufficient to demonstrate that the proposed installation or operation of the Rooftop Equipment shall not damage the structural integrity of the Building, interfere with other Building operations or systems, or impair any applicable roof warranty.

Tenant shall engage Landlord's roofer (or another roofing contractor reasonably approved by Landlord and approved by Landlord's roof manufacturer) before beginning any rooftop installations or repairs of the Rooftop Equipment, whether under this Article 11 or otherwise, and shall always comply with the roof warranty governing the protection of the roof and modifications to the roof. Tenant shall obtain a letter from Landlord's roof manufacturer following completion of such work stating that the roof warranty remains in effect, if required pursuant to the terms of the roof warranty. Tenant, at its sole cost and expense, shall inspect areas on the rooftop where the Rooftop Equipment is located on a periodic basis as reasonably required by Landlord (taking into account the nature of Tenant's installations and operational history of the equipment in question) and correct any loose bolts, fittings or other appurtenances and shall repair any damage to the roof caused by the installation or operation of the Rooftop Equipment. Tenant covenants that the installation, existence, maintenance and operation of the Rooftop Equipment shall not violate any Legal Requirements or constitute a nuisance under law. Tenant shall pay Landlord on demand (i) all applicable taxes or governmental charges, fees, or impositions imposed on Landlord because of Tenant's use of the Rooftop Equipment under this Article 11 and (ii) the amount of any increase in Landlord's insurance premiums as a result of the installation or existence of the Rooftop Equipment.

11.3 Operation of Rooftop Equipment. Landlord shall have no responsibility for ensuring the proper operation of the Rooftop Equipment, but shall reasonably cooperate with Tenant (at no cost to Landlord) to address Tenant's reasonable requests regarding any such operational issues in a manner consistent with comparable rooftop usage in Comparable Properties. If Tenant's Rooftop Equipment (i) causes physical damage to the roof or the structural integrity of the Building, or (ii) materially, adversely interferes with any of the Building's mechanical or other systems, Tenant shall within five (5) business days after notice of a claim of damage or interference reasonably cooperate with Landlord to determine the source of the damage or interference and effect a prompt solution at Tenant's expense (if Rooftop Equipment caused such interference or damage). In the event that, after Tenant's installation of the Rooftop Equipment, Landlord requires any particular elements of the Rooftop Equipment (including, without limitation, any pipes, ducts, conduits, wires and appurtenant equipment) to be relocated on account of other Building-related equipment, the same shall be relocated to comparably functional space on the roof or in other portions of the Building, as applicable, provided that Landlord shall pay or reimburse Tenant for the reasonable costs of moving such

equipment to such other space, taking such other steps necessary to ensure comparable functionality of equipment, and installing connecting equipment to a condition comparable to the then condition of the current location of such equipment. Tenant shall arrange for the relocation of the affected equipment within ninety (90) days after a comparable space is agreed upon or selected by Landlord. Any actions by Landlord in connection with a relocation hereunder shall be performed in a manner designed to minimize interference with Tenant's business.

ARTICLE 12. DAMAGE OR DESTRUCTION; CONDEMNATION

12.1 Damage or Destruction of Premises.

- (a) Repairs and Restoration. If the Building or the Premises or any part thereof shall be damaged by fire or other insured casualty, then, subject to the provisions below, Landlord shall proceed with diligence, subject to then applicable Legal Requirements and the Project Condominium Documents, to repair or cause to be repaired such damage, including the Base Building Work and the Leasehold Improvements, but excluding Tenant's FF&E, other Tenant Property, and any Tenant-Insured Work. Any restoration of Leasehold Improvements may include such changes or re-designs to the prior Tenant Work that Tenant may desire and are approved by Landlord in accordance with Section 10.5, provided that Tenant shall be responsible for all costs and delays arising from such Tenant-requested changes. To the extent that the cost of repairs is less than the deductible amount in Landlord's property insurance policy maintained in accordance with Section 7.4, such repairs shall be made by Landlord at Tenant's expense (including all hard and soft costs incurred by Landlord in making such repairs, which shall be paid by Tenant as Additional Rent hereunder), provided that (i) such repairs shall be made at Landlord's expense, to the extent the damage was due to the negligence or willful misconduct of Landlord, and (ii) in the case of such a casualty (for which the repair costs are less than such deductible amount) that primarily involves damage to Tenant Work, Landlord reserves the right to require Tenant to perform such repairs at Tenant's expense.
- (b) <u>Tenant Restoration Items</u>. In the case of fire or other casualty, Tenant shall be solely responsible for making, at its expense, all repairs to and replacements of Tenant's FF&E, other Tenant Property, or Tenant-Insured Work. Tenant shall promptly and diligently restore, replace, or remove any such damaged items as Tenant may determine are then necessary for the conduct of its business in the Premises, with all such work being performed by Tenant at its sole expense in accordance with <u>Section 10.5</u> and the provisions of the Lease.
- (c) <u>Abatement of Rent Due to Casualty</u>. If the Building or the Premises or any part thereof shall have been rendered unfit for use and occupation for the Permitted Use hereunder by reason of such damage, the Base Rent and Tenant's Pro Rata Share of Total Operating Costs, or a just and proportionate part thereof according to the nature and extent to which the Building or the Premises or applicable portion thereof shall have been so rendered unfit, shall be abated until the Premises (except as to Tenant's FF&E, other Tenant Property, or Tenant-Insured Work) shall have been restored as nearly as practicable to the condition in which they were immediately prior to such fire or other casualty (excluding the period of any delay due to Tenant's changes or re-designs to the prior Tenant Work as provided above, or any period

beyond the time reasonably required to restore the Premises in the case of repairs to Tenant Work performed by Tenant as provided above) and for such further period (not to exceed thirty (30) days) reasonably required for Tenant to recommence occupancy of the restored portion of the Premises.

- (d) <u>Restoration Work</u>. Landlord shall, from time to time upon Tenant's request, advise Tenant of the then expected date on which Landlord's repairs will be substantially completed. Landlord shall not be liable for delays in the making of any such repairs that are due to Force Majeure, nor shall Landlord be liable for any inconvenience or annoyance to Tenant or injury to the business of Tenant resulting from delays in repairing such damage, subject to the abatement of Rent as provided above, but Landlord shall use diligent efforts to pursue such repairs promptly in view of the nature and extent of such Force Majeure. Landlord shall have the right, but not the obligation, to suspend or delay any repair or replacement work hereunder for so long as a Material Event of Default by Tenant then exists.
- (e) Termination Due to Casualty. If (i) the Building or the Premises or any substantial portion thereof are so damaged by fire or other casualty (whether or not insured) at any time during the last three (3) years of the Term (as the same may have been extended, provided that Tenant may irrevocably exercise any then remaining Extension Option under Exhibit D-1 by notice to Landlord within thirty (30) days after such fire or other casualty in order to avoid the application of this clause (i), notwithstanding the fact that the extension procedures set forth in Exhibit D-1 may not have then been commenced) that the cost to repair such damage is reasonably estimated to exceed one-third of the total Base Rent payable hereunder for the period from the estimated completion date of repair until the end of the Term, (ii) the Building or the Premises or any substantial portion thereof are so substantially damaged that Legal Requirements prohibit Landlord from restoring the Building to substantially the same condition and at least the same size existing prior to such casualty, or (iii) the Building or the Premises or any substantial portion thereof are damaged by an uninsured casualty that was not covered (and was not required to be covered) under Landlord's property policy under the terms of Section 7.4 and Landlord determines not to repair such damage (and Tenant does not elect, by notice given within thirty (30) days after Landlord's notice of such determination, to pay for the costs of performing such repair work), then and in any of such events, this Lease and the Term hereof may be terminated at the election of Landlord by a notice from Landlord to Tenant within sixty (60) days after such damage, or such longer period as is required to complete adjustment of such casualty loss with the insurer, and the effective termination date pursuant to such notice shall be not less than sixty (60) days after the day on which such termination notice is received by Tenant. Notwithstanding the foregoing, in the event that, in the case of a fire or other casualty event during the last three (3) years of the Term under clause (i) above, at least half of the Premises remains tenantable (without abatement of rent for such portion of the Premises), then Tenant may elect, by notice given to Landlord within thirty (30) days after delivery of Landlord's termination notice thereunder, to postpone the effective date of Landlord's termination notice to a date specified in such Tenant's notice that is not later than eighteen (18) months after the date of the fire or other damage (but not later than the date on which Term would have expired in the absence of such termination), in which event Tenant shall continue to pay rent on the portion of the Premises that is tenantable and Landlord shall have the right, but not the obligation, to restore any portion of the damage hereunder.

If, following any fire or other casualty, Landlord has failed to commence such repair or replacement work as required under this <u>Section 12.1</u> within three (3) months following adjustment of such casualty loss with the insurer (subject to any delays caused by Tenant, and to events of Force Majeure not exceeding ninety (90) days), then Tenant may, until any such repair or replacement work commences, terminate this Lease by giving at least thirty (30) days prior written notice thereof to Landlord and such termination shall be effective on the date specified in such notice if such repair or replacement work has not then commenced.

If less than twelve (12) months remain in the Term at the time of a fire or other casualty that requires Tenant to cease operations in all or any substantial portion (for such purposes, at least half) of the Premises for a period of at least thirty (30) days and in Landlord's reasonable estimate the time to restore the Premises will take more than one-half of the then remaining Term, then either party may upon thirty (30) days' prior written notice terminate this Lease provided that such termination election by Tenant shall be null and void if Landlord completes such restoration within thirty (30) days after such notice; provided that if Landlord so elects to terminate, Tenant may elect, by notice given to Landlord within thirty (30) days after delivery of Landlord's termination notice thereunder, to defer the effective date of Landlord's termination notice to a date specified in such notice (but not later than the date on which Term would have expired in the absence of such termination), in which event Tenant shall continue to pay rent on the portion of the Premises that is tenantable and Landlord shall have the right, but not the obligation, to restore any portion of the damage hereunder.

In the event of any termination under this <u>Section 12.1</u>, the Term shall expire as though such effective termination date were the date originally stipulated in <u>Section 1.5</u> for the end of the Term (with all remaining Extension Options being deemed terminated) and the Base Rent and Additional Rent (to the extent not abated as set forth above) shall be apportioned as of such date.

12.2 Eminent Domain. In the event that all or any substantial part of the Premises or the Building (or the Project Common Facilities necessary for access and use of the Premises or the Building) for Tenant's business operations as then conducted are taken (other than for temporary use hereafter described) by public authority under power of eminent domain (or by conveyance in lieu thereof), then by notice given within three months following the recording of such taking (or conveyance) in the appropriate registry of deeds, this Lease may be terminated at either party's election thirty (30) days after such notice, and Rent shall be apportioned as of the date of termination. If this Lease is not terminated as aforesaid, subject to the rights of mortgagees and applicable Legal Requirements, Landlord shall within a reasonable time thereafter diligently restore what may remain of the Premises (excluding any Tenant's FF&E, other Tenant Property, or Tenant-Insured Work, or other items installed or paid for by Tenant that Tenant is permitted or may be required to remove upon expiration) to a tenantable condition for occupancy by Tenant for the Permitted Uses. In the event some portion of useable floor area of the Premises is taken (other than for temporary use) and this Lease is not terminated, Base Rent shall be proportionally abated for the remainder of the Term. In the event of any taking of the Premises or any part thereof for temporary use, (i) this Lease shall be and remain unaffected thereby and rent shall not abate, and (ii) Tenant shall be entitled to receive for itself such portion or portions of any award made for such use with respect to the period of the taking that is within the Term, provided that if such taking shall remain in force at the expiration or earlier

termination of this Lease, then Tenant shall pay to Landlord a sum equal to the reasonable cost of performing Tenant's obligations hereunder with respect to surrender of the Premises and upon such payment shall be excused from such obligations.

Any damages that are expressly awarded to Tenant on account of Tenant's FF&E or other Tenant Property and Tenant's relocation expenses, and specifically so designated, shall belong to Tenant. Except as provided in the preceding sentence of this paragraph, Landlord reserves to itself, and Tenant releases and assigns to Landlord, all rights to damages accruing on account of any taking or by reason of any act of any public authority for which damages are payable. Subject to its rights hereunder, Tenant agrees to execute such further instruments of assignment as may be reasonably requested by Landlord, and to turn over to Landlord any damages that may be recovered in any proceeding or otherwise.

ARTICLE 13. ASSIGNMENT AND SUBLETTING

- 13.1 Landlord's Consent Required. Except for a Permitted Transfer (as defined below), Tenant shall not transfer, or permit the transfer of, all or any part of the Premises or of its interest in this Lease to any other entity, whether by sale, assignment, mortgage, sublease, license, transfer, operation of law (including, without limitation by merger, consolidation, sale or other transfer of all or substantially all of the stock or business and assets of Tenant, or otherwise) or act of Tenant, subtenant, or any other party acting under or through Tenant (each a "Transfer" to a "Transferee") without Landlord's prior written consent as provided in Section 13.3 below. Consent to one Transfer shall not imply consent to any other Transfer or waive the consent requirement as to any other Transfer. As used herein, the term "Third Party" shall mean any party other than a Permitted Transferee (as defined below). Any attempted Transfer without having received the required Landlord consent hereunder shall be void at the election of Landlord. Any entity to which a Transfer is made is a "Transferee". Notwithstanding anything to the contrary herein, so long as Tenant's shares are and remain traded on a nationally recognized stock exchange, any sale of Tenant's shares on such exchange shall not be deemed a Transfer for purposes of this Article 13. Tenant acknowledges that the covenants contained in this Article 13 are material to the transaction contained herein and that Landlord shall have, in addition to any other rights and remedies available under this Lease or at law, the right to seek injunctive relief and/or specific performance in order to enforce such covenants.
- 13.2 Permitted Transfers. The following transactions set forth in this Section 13.2 (each, a "Permitted Transfer" to a "Permitted Transferee") shall not require the consent of Landlord provided that (i) Landlord shall receive prior notice (except to the extent prohibited by applicable securities or other laws or regulations, in which event such notice shall be given when such notice is permissible under such laws or regulations) thereof plus reasonable evidence upon closing that the transaction is in fact a Permitted Transfer, and (ii) the Permitted Transfer complies with all other provisions of this Lease (including, without limitation, this Article 13), does not alter Landlord's rights under this Lease, does not impose any additional obligation on Landlord, and the transaction is not part of a series of transfers that, when taken together, avoids the restrictions on Transfers to a Third Party under this Article 13 (such as a sublease or assignment to a Tenant affiliate that is thereafter acquired by a Third Party):
 - (a) An assignment of the Lease to an entity acquiring all or substantially all of the stock or business and assets of the Tenant originally named herein (or of any Successor Entity or any Third Party assignee approved by Landlord under Section 13.3), whether by way of merger, consolidation, acquisition, sale of stock or business and assets, or otherwise (any such entity, a "Successor Entity"); or

- (b) Any sublease, license, or other Transfer (other than an assignment of the Lease) of all or any portion of the Premises to a Related Entity (as defined below); provided that the Permitted Transferee shall remain a Related Entity of Tenant and Guarantor; or
- (c) An assignment of the Lease to a Related Entity, provided that the Related Entity shall have a financial condition sufficient to satisfy Tenant's then remaining obligations under the Lease and remain a Related Entity of Tenant and Guarantor.

As used herein, "Related Entity" shall mean an entity directly or indirectly controlled, controlling, or under common control with Tenant, and "control" (and its cognates) shall mean possession of more than fifty percent (50%) ownership of the shares, membership interests, or other beneficial ownership interest of the entity in question, together with the power to control and manage the affairs thereof either directly or by election of directors and/or officers. For avoidance of doubt, the parties acknowledge that Tenant is the operating company that owns and operates the TripAdvisor on-line travel services business and that, as recited in the Guaranty, Tenant is a wholly owned subsidiary of TripAdvisor Holdings LLC, a Delaware limited liability company, and TripAdvisor Holdings LLC is a wholly owned subsidiary of Guarantor, a publicly traded company. Accordingly, the parties acknowledge that clause (a) above refers to an assignment of the Lease to a Successor Entity that acquires all or substantially all of the stock or business and assets of Tenant, whether such transaction is structured by way of merger, consolidation, acquisition, or sale of stock or business and assets of Tenant, TripAdvisor Holdings LLC, or Guarantor, or otherwise.

13.3 <u>Consent Procedures</u>. Tenant's request for Landlord's consent to any Transfer for which Landlord's consent is required under this <u>Article 13</u> shall be made at least thirty (30) days prior to the effective date of the proposed Transfer, describe the details of the proposed Transfer, including the name, business and financial condition of the prospective Transferee, and the financial terms of the proposed Transfer (e.g., payments in consideration of the proposed Transfer, term, rent, construction, and security deposit), and Tenant shall also provide any other information in Tenant's possession (or reasonably obtainable from the proposed Transferee) Landlord reasonably deems relevant, including without limitation the proposed form of Transfer documentation. Landlord shall not unreasonably withhold, condition or delay (more than twenty (20) days following receipt of Tenant's request for consent with all information required herein) its consent to any assignment or subletting of the Premises, provided that Tenant is not then in default under this Lease (following the giving of notice of such default, where applicable). It shall not be deemed unreasonable for Landlord to deny consent for the following reasons, among others: (i) the business of the proposed Transferee or the proposed use of the Premises are inconsistent with the Permitted Uses; (ii) the net worth and financial condition of any proposed assignee or subtenant of a substantial portion of the Premises is not reasonably satisfactory to Landlord, taking into account the then remaining obligations under this Lease (in the case of an assignment) or under the sublease (in the case of a sublease) and Tenant's then net

worth and financial condition; (iii) any Material Event of Default shall then exist, (iv) the Transferee's proposed particular use of the Premises is reasonably incompatible with a first class suburban office building, taking into account the existing tenants and use mix at the Project, or the Transferee's business reputation is reasonably unsatisfactory to Landlord (e.g., a current or recent criminal conviction or investigation), or the Transferee's proposed occupancy would result in a use of the Building that is open to the general public, (v) the Transferee or its affiliate has filed a petition for insolvency or for appointment of a receiver, trustee or assignee or for adjudication, reorganization or arrangement under any bankruptcy act, or if any similar petition has been filed against such Transferee or affiliates, or (vi) the Transfer in question would result in a violation of the Project Condominium Documents or any Legal Requirement, including without limitation any Legal Requirement governing contracts or agreements with so-called "prohibited persons" under the laws, rules and regulations promulgated by the Office of Foreign Asset Control in the United States Department of the Treasury or any Legal Requirement under the Employee Retirement Income Security Act of 1974, as amended.

13.4 Sharing of Transfer Profits. Tenant shall pay to Landlord, as Additional Rent, fifty percent (50%) of the Profits (as defined below) arising from any Transfer (other than a Permitted Transfer) as and when received by Tenant, unless Landlord notifies Tenant and the Transferee that the Transferee shall pay Landlord's share of the Profits directly to Landlord. "Profits" means (a) all rent, fees and other consideration paid for or in respect of the Transfer (excluding the fair market value of personal property purchased from Tenant by such Transferee, but including consideration in excess of the fair market value of personal property and fees in excess of reasonable amounts under collateral agreements, the intent being to prohibit Tenant from shifting occupancy costs to collateral agreements), less (b) the sum of (i) the Rent and other sums payable under this Lease (or if the Transfer is a sublease of part of the Premises, allocable to the subleased premises) and (ii) all reasonable costs and expenses directly incurred by Tenant for all reasonable real estate broker's commissions, legal fees, and costs of renovation or construction of tenant improvements in the applicable space required by the Transfer for the applicable space, with the costs of such improvements to be amortized on a straight-line basis over the applicable term of the Transfer transaction. Tenant shall give Landlord a written statement certifying all amounts to be paid from any Transfer (including any collateral agreements) within thirty (30) days after the transfer agreement is signed and from time to time thereafter on Landlord's request, and Landlord may inspect Tenant's books and records to verify the accuracy of such statements. On written request, Tenant shall promptly furnish to Landlord copies of all Transfer documents, certified by Tenant to be complete, true and correct. The provisions of this Section shall survive the expiration or earlier termination of the Lease.

13.5 No Release. Notwithstanding any Transfer (whether a Permitted Transfer or a Transfer to a Third Party, and whether or not the same is consented to), the liability of Tenant to Landlord for all obligations under this Lease shall remain direct and primary. Any Transferee shall be jointly and severally liable with Tenant to Landlord for the performance of all of Tenant's covenants under this Lease, but only to the extent undertaken in the Transfer instrument; and such Transferee shall, upon Landlord's request, execute and deliver such instruments as Landlord reasonably requests in confirmation thereof. Tenant hereby irrevocably authorizes Landlord, upon the occurrence of a default (following the giving of notice of such default, where applicable) to collect rent directly from any Transferee (and upon notice any

Transferee shall pay directly to Landlord) and apply the net amount collected to the Rent and other charges reserved under this Lease. No Transfer (whether or not consented to by Landlord, and whether or not such consent is required) shall be deemed a waiver of the provisions of this Section, or the acceptance of the Transferee as a tenant, or a release of Tenant from direct and primary liability for the performance of all of the covenants of this Lease. The consent by Landlord to any Transfer shall not relieve Tenant or any Transferee from the obligation of obtaining the express consent of Landlord to any modification of such Transfer or a further Transfer by Tenant or such Transferee. Notwithstanding anything to the contrary in the documents effecting the Transfer, Landlord's consent shall not alter in any manner whatsoever the terms of this Lease, to which any Transfer at all times shall be subject and subordinate.

13.6 Additional Provisions.

- (a) <u>Delivery of Documents, Etc.</u> Within five (5) business days after the final execution of the documents for any Transfer for which Tenant has received Landlord's consent hereunder or for any Permitted Transfer, Tenant shall deliver to Landlord (i) a true and complete copy of the fully executed instrument or instruments evidencing such Transfer and (ii) a written agreement of the Transferee agreeing with Landlord to perform and observe all of the terms, covenants, and conditions of this Lease undertaken by such Transferee under such Transfer documents. Tenant shall pay to Landlord, as Additional Rent, all reasonable third party costs incurred by Landlord (including, without limitation, reasonable attorneys' fees) in reviewing any proposed Transfer, whether or not any such proposed Transfer is consummated by Tenant pursuant to the provisions of this <u>Article 13</u>.
- (b) <u>Prohibition on Rents Based on Net Profits</u>. Anything contained in the foregoing provisions of this <u>Article 13</u> to the contrary notwithstanding, neither Tenant nor any Transferee nor any other person having an interest in the possession, use, occupancy or utilization of the Premises shall enter into any lease, sublease, assignment, license, concession or other agreement for use, occupancy or utilization of space in the Premises that provides for rental or other payment for such use, occupancy or utilization based, in whole or in part, on the net income or profits derived by any person from the Premises leased, used, occupied or utilized (other than an amount based on a fixed percentage or percentages of receipts or sales), and any such purported lease, sublease, assignment, license, concession or other agreement shall be absolutely void and ineffective as a conveyance of any right or interest in the possession, use, occupancy or utilization of any part of the Premises.

ARTICLE 14. EVENTS OF DEFAULT AND REMEDIES

- 14.1 Events of Default. As used herein, an "Event of Default" shall mean and be deemed to have occurred if any of the following occurs:
- (a) if Tenant fails to pay the Base Rent or any portion thereof under <u>Section 4.1</u> or any regular monthly installment of Additional Rent under <u>Section 4.2</u> when due and such default continues for five (5) days after written notice from Landlord, provided that such notice and cure period shall not apply after the second occasion during any twelve-(12)-month period in which a default notice for such a failure is given to Tenant,

- (b) if Tenant, with respect to any non-recurring payment of Additional Rent, fails to pay any such Additional Rent when due and such default continues for ten (10) days following notice from Landlord,
 - (c) if Tenant (or any Transferee) makes any Transfer (each as defined in Article 13) in violation of this Lease,
- (d) if a petition is filed by Tenant (or any assignee of Tenant's interest in this Lease) or any Guarantor for insolvency or for appointment of a receiver, trustee or assignee or for adjudication, reorganization or arrangement under any bankruptcy act, or if any similar petition is filed against Tenant (or such assignee) or any Guarantor and such petition is not dismissed within sixty (60) days thereafter,
- (e) if any representation or warranty made by Tenant in this Lease is untrue in any material respect and is not cured within ten (10) days after notice thereof, or
- (f) if Tenant fails to perform any other covenant or condition hereunder and such default continues longer than any period (following notice, if expressly required) expressly provided in this Lease for the correction thereof (and if no period is expressly provided, then for thirty (30) days after notice is given, provided, however, that such period shall be reasonably extended in the case of any such non-monetary default that cannot be cured within such period (but in any event shall not exceed ninety (90) days in the aggregate) only if the matter complained of can be cured, Tenant begins promptly and thereafter diligently completes the cure, and Tenant gives Landlord notice of such intent to cure within ten (10) days after notice of such default).

Time is of the essence in performance of all covenants and conditions set forth herein.

14.2 Right to Terminate. If any Event of Default occurs, then, and in any such case, Landlord and its agents lawfully may, in addition to any remedies for any preceding breach, immediately or at any time thereafter without further demand or notice, enter upon any part of the Premises in the name of the whole, or mail or deliver a notice of termination of the Term of this Lease addressed to Tenant at the Premises or any other address herein, and thereby terminate the Term and repossess the Premises in accordance with process of law. At Landlord's election such notice of termination may be included in any notice of default, subject to any applicable cure period. Upon such entry or mailing of such notice of termination, the Term shall terminate, all executory rights of Tenant and all obligations of Landlord will immediately cease, and Landlord may expel Tenant and all persons claiming under Tenant and remove their effects without any trespass and without prejudice to any remedies for arrears of Rent or prior breach; and Tenant waives all statutory and equitable rights to its leasehold (including rights in the nature of further cure or redemption, if any, to the extent such rights may be waived). If Landlord engages attorneys in connection with any failure by Tenant to perform its obligations hereunder, any collection of amounts not paid when due hereunder, or any enforcement of Tenant's obligations hereunder that Tenant has failed to perform, Tenant shall reimburse Landlord, as Additional Rent, for the reasonable fees of such attorneys on demand from time to time. Without implying that other provisions do not survive, the provisions of this Article 14 shall survive the expiration of the Term or any earlier termination of this Lease.

14.3 Remedies for Default.

- (a) Reletting Expenses Damages. If the Term of this Lease is terminated for default, Tenant covenants, as an additional cumulative obligation after such termination, to pay all of Landlord's reasonable costs and expenses, including without limitation reasonable attorneys' fees, related to Tenant's default and in collecting amounts due and all reasonable expenses in connection with reletting, including tenant inducements to new tenants, brokerage commissions, fees for legal services, expenses of preparing the Premises or part of parts thereof for reletting and the like (collectively, "Reletting Expenses"). It is agreed that Landlord may (i) relet the Premises or part or parts thereof for a term or terms that may be equal to, less than, or exceed the period that would otherwise have constituted the balance of the Term, and may grant such tenant inducements, including free rent and other concessions, as Landlord in its sole good faith discretion considers advisable, and (ii) make such alterations to the Premises or part or parts thereof as Landlord in its sole good faith discretion considers advisable, and no failure to relet or to collect rent under any reletting shall operate to reduce Tenant's liability hereunder. Any obligation imposed under applicable law on Landlord to use reasonable efforts to relet the Premises shall be subject to Landlord's reasonable objectives of developing and leasing its property and the Project to reputable first-class tenants in a harmonious manner with appropriate mixes of tenants, uses, floor areas, terms and the like, and in no event shall Landlord shall be obligated to relet the Premises or any portion thereof to any party to whom Landlord or its affiliate may desire to lease other available space in the Project. Landlord's Reletting Expenses, together with all other sums provided for whether incurred prior to or after such termination, shall be due from Tenant to Landlord upon Landlord's demand from time to time.
- (b) <u>Termination Damages</u>. If the Term of this Lease is terminated for default, unless and until Landlord elects lump sum liquidated damages described in the next paragraph, Tenant covenants, as an additional, cumulative obligation after any such termination, to pay punctually to Landlord all the sums and perform all of its obligations in the same manner as if the Term had not been terminated. In calculating such amounts Tenant will be credited with the net proceeds of any rent (if any) then actually received by Landlord from a reletting of the Premises after deducting all Reletting Expenses that have not then been paid by Tenant to Landlord, provided that in no event shall Tenant be entitled to receive any portion of the re-letting proceeds, even if the same exceed the Rent originally due hereunder.
- (c) <u>Lump Sum Liquidated Damages</u>. If the Term of this Lease is terminated for default, Tenant covenants, as an additional, cumulative obligation after any such termination, to pay forthwith to Landlord at Landlord's election made by written notice at any time after termination, as liquidated damages, a single lump sum payment equal to the sum of (i) all sums to be paid by Tenant and not then paid at the time of such election, plus (ii) the excess of (x) the present value of all of the Rent reserved for the residue of the Term in the absence of such termination (with Additional Rent deemed to increase 5% in each year on a compounding basis) over (y) the present value of the aggregate fair market rent and Additional Rent (with Additional Rent deemed to increase 5% in each year on a compounding basis) payable (if less than the Rent payable hereunder) on account of the Premises during such period, which fair market rent shall be reduced by reasonable projections of vacancies and by Landlord's Reletting Expenses described above to the extent not theretofore paid to Landlord), plus (iii) the amount of free rent provided under <u>Section 4.1(a)</u> of the Lease (less a pro-rata portion thereof, in proportion that the

number of months of the Term after the Rent Commencement Date for which Tenant made payment of all Base Rent due to Landlord bears to the total number of months in the Initial Term following the Rent Commencement Date in the absence of such termination). The Federal Reserve discount rate (or equivalent) shall be used in calculating such present values under clause (ii), and in the event the parties are unable to agree on such fair market rent, the matter shall be submitted, upon the demand of either party, to the office of the American Arbitration Association in Boston, with a request for arbitration in accordance with the expedited commercial arbitration rules of the Association by a single arbitrator who shall be a licensed real estate broker with at least ten (10) years' experience in the leasing of 1,000,000 or more square feet of floor area of buildings similar in character and location to the Premises, and who shall not be affiliated with either Landlord or Tenant and has not worked for either party or its affiliates at any time during the prior five (5) years, whose decision shall be conclusive and binding on the parties.

- (d) Remedies Cumulative; Jury Waiver; Late Performance. The remedies to which Landlord may resort under this Lease, and all other rights and remedies of Landlord are cumulative, and any two or more may be exercised concurrently or in such order as Landlord may from time to time elect, except where this Lease specifically provides otherwise. In addition to the other remedies expressly provided in this Lease, Landlord shall be entitled to the restraint by injunction of the violation or attempted or threatened violation of any of the covenants, conditions or provisions of this Lease by Tenant or to a decree compelling specific performance of any such covenants, conditions or provisions. Nothing in this Lease shall limit the right of Landlord to prove and obtain in proceedings for bankruptcy or insolvency an amount equal to the maximum allowed by any statute or rule of law in effect at the time, but not to exceed the limitations set forth in this Section 14.3; provided that Tenant agrees that the fair value for occupancy of all or any part of the Premises occupied by Tenant under such proceedings at all times shall never be less than the Base Rent and all Additional Rent payable from time to time. Tenant shall also indemnify and hold Landlord harmless, in the manner provided in Section 9.2, if Landlord shall become or be made a party to any claim or action necessary to protect Landlord's interest under this Lease in a bankruptcy proceeding, or other proceeding under Title 11 of the United States Code, as amended. LANDLORD AND TENANT WAIVE TRIAL BY JURY IN ANY ACTION TO WHICH THEY ARE PARTIES, and further agree that any action arising out of this Lease (except an action for possession by Landlord, which may be brought in whatever manner or place provided by law) shall be brought in the Trial Court, Superior Court Department, in the county where the Premises are located.
- (e) Waivers; Accord and Satisfaction. No consent by Landlord or Tenant to any act or omission that otherwise would be a default shall be construed to permit other similar acts or omissions. Neither party's failure to seek redress for violation or to insist upon the strict performance of any covenant, nor the receipt by Landlord of Rent with knowledge of any breach of covenant, shall be deemed a consent to or waiver of such breach. No breach of covenant shall be implied to have been waived unless such waiver is in writing, signed by the party benefiting from such covenant and delivered to the other party; and no acceptance by Landlord of a lesser sum than the Rent due shall be deemed to be other than on account of the earliest installment of such Rent. No endorsement or statement on any check or in any letter accompanying any check or payment shall be deemed an accord and satisfaction; and Landlord may accept such check or payment without prejudice to Landlord's right to recover the balance of such installment or

pursue any other right or remedy. The acceptance by Landlord of any Rent following the giving of any default and/or termination notice shall not be deemed a waiver of such notice. Tenant shall not interpose any counterclaim or counterclaims (other than compulsory counterclaims that would be lost if not interposed) in a summary proceeding or in any action based on non-payment of Rent, it being agreed that Tenant shall not be deemed to have waived any such claims (other than compulsory counterclaims) that are not so interposed.

- (f) <u>Landlord's Right to Cure</u>. If Tenant fails to perform any covenant within any applicable notice and/or cure period, then Landlord at its option may (without waiving any right or remedy for Tenant's non-performance), after giving Tenant not less than ten (10) days' prior notice of Landlord's intent to exercise its right to undertake such curative work under this <u>Section 14.3(f)</u> at any time thereafter, perform the covenant for the account of Tenant, provided that Tenant has not then effected such cure. Tenant shall, within thirty (30) days after demand therefor accompanied by reasonable supporting documentation, reimburse Landlord's cost (including reasonable attorneys' fees) of so performing such work as Additional Rent. Notwithstanding any other provision concerning cure periods, Landlord may cure any non-performance for the account of Tenant after such notice to Tenant, if any, as is reasonable under the circumstances if curing prior to the expiration of the applicable cure period is reasonably necessary to prevent likely damage to the Premises or Building or possible injury to persons, or to protect Landlord's interest in the Premises or Building.
- 14.4. <u>Late Charge</u>. Tenant acknowledges that, if Rent is not paid when due under the provisions of this Lease, Landlord shall incur unanticipated costs, which shall be extremely difficult to ascertain exactly. Such costs include processing and accounting charges, together with late charges that may be imposed on Landlord by any mortgage on the Property. Accordingly, if Landlord does not receive any Rent payment within five (5) days following its due date, Tenant shall pay Landlord a late charge equal to two percent (2%) of the overdue amount. The parties agree that this late charge represents a fair and reasonable estimate of the costs Landlord shall incur by reason of Tenant's delay or failure to pay the Rent when due. Payment of the late charge shall not cure Tenant's payment default or prevent Landlord from exercising other rights and remedies. Notwithstanding the foregoing, no late charges under this <u>Section 14.4</u> shall accrue until Landlord provides notice of such late payment to Tenant and five (5) business days elapse from such notice without Tenant having made such payment; provided, however, that Landlord shall not be required to give such notice more than once during any twelve-(12)-month period.
- 14.5. Interest. Any Rent not paid when due under the provisions of this Lease shall bear interest from the date due until the date paid at the annual rate (the "Default Rate") equal to the greater of (a) twelve percent (12%) per annum or (b) the prime rate of interest from time to time published by the Wall Street Journal (or its successor) plus four percent (4%) per annum, provided that, if such Default Rate should exceed the rate permitted by applicable law, the Default Rate shall be reduced to the rate legally permitted. Payment of such interest shall not cure Tenant's payment default or prevent Landlord from exercising other rights and remedies. Notwithstanding the foregoing, no interest under this Section 14.5 shall accrue until Landlord provides notice of such late payment to Tenant and five (5) business days elapse from such notice without Tenant having made such payment; provided, however, that Landlord shall not be required to give such notice more than once during any twelve-(12)-month period.

14.6. Letter of Credit.

- (a) <u>LC Amount</u>. Tenant shall provide to Landlord a clean, irrevocable letter of credit as security for the performance of the obligations of Tenant hereunder, subject to the terms and conditions set forth in this <u>Section 14.6</u> (together with any renewal, replacement, or increase thereof or supplement thereto in accordance herewith, the "<u>Letter of Credit</u>"). Simultaneously with Tenant's execution of this Lease, Tenant shall provide the Letter of Credit in the initial amount set forth for the Letter of Credit in <u>Section 1.7</u> (the "<u>Initial LC Amount</u>"). Any Letter of Credit delivered under this <u>Section 14.6</u> shall comply with and be subject to the requirements of <u>Exhibit I</u>. The amount of the Letter of Credit that is from time to time required to be maintained under the terms of this <u>Section 14.6</u> is referred to herein as the "<u>LC Amount</u>."
- (b) <u>Increased LC Amount</u>. Notwithstanding the foregoing, if an LC Increase Trigger Event (as defined below) occurs, Tenant shall (i) give Landlord notice of the LC Increase Trigger Event within ten (10) days after its occurrence and (ii) increase the Letter of Credit to an amount equal to six (6) times the Initial LC Amount (the "<u>Increased LC Amount</u>") as required herein. Tenant shall cause such increase in the LC Amount to be effected, within thirty (30) days after the occurrence of the LC Increase Trigger Event, by delivering to Landlord either an amendment to the existing Letter of Credit or (if so desired by Tenant) a replacement or supplemental Letter of Credit for such Increased LC Amount, in either case in compliance with the requirements of <u>Exhibit I</u>. Tenant's obligations under this <u>Section 14.6(b)</u> shall be automatic upon the occurrence of an LC Increase Trigger Event, without the requirement for any notice or demand by Landlord. The failure by Tenant to timely increase the Letter of Credit as and when required under this <u>Section 14.6(b)</u> shall constitute an Event of Default, without any notice or cure period under <u>Section 14.1</u>. After the occurrence of an LC Increase Trigger Event, Tenant shall maintain the Letter of Credit in the Increased LC Amount, except as provided in <u>Section 14.6(e)</u> below.

(c) <u>Definitions</u>. As used herein:

- "LC Increase Trigger Event" shall mean the occurrence of either (x) a Credit Agreement Default (as defined below) or (y) a Net Worth Shortfall (as defined below).
- "Credit Agreement Default" shall mean a default under the terms of any Credit Agreement (as defined below) beyond (i) any applicable notice and cure period under such Credit Agreement and (ii) any forbearance period or waiver of a covenant violation, if any, that such Credit Agreement lender may have granted prior to the expiration of the such notice and cure period.
- "Credit Agreement" shall mean the \$400,000,000 credit line agreement with TripAdvisor Inc. (or any subsidiary thereof) existing on the Lease Date, as the same may hereafter be extended, supplemented, or modified from time to time, and any replacement or new credit agreement(s) existing from time to time under which Tenant or any Related Entity is a borrower, including without limitation any credit agreement from time to time maintained by TripAdvisor Holdings LLC (which currently owns 100% of the ownership interests in TripAdvisor, Inc. (which currently owns 100% of the ownership interests in TripAdvisor Holdings LLC).

- "Net Worth Shortfall" shall mean that the Net Worth (as defined below) of Tenant is less than Four Hundred Million Dollars (\$400,000,000).
- "<u>Net Worth</u>" shall mean the Total Stockholder's Equity of Tenant as defined by generally accepted accounting standards, in a manner consistent with TripAdvisor Inc.'s financial statements as reported in its quarterly and annual filings with the U.S. Securities and Exchange Commission ("<u>SEC</u>").
- (d) <u>Confirmations</u>. Tenant represents and warrants that, as of the Lease Date, Tenant has disclosed to Landlord the existence of all existing Credit Agreements, no Credit Agreement Default exists, and there has been no material adverse change in Tenant's Net Worth since the date of the audited Tenant financial statements provided to Landlord prior to the execution of this Lease. Within ten (10) business days after Landlord's request from time to time, Tenant shall provide Landlord with (i) a certification, signed by Tenant's chief financial officer, that no Credit Agreement Default or Net Worth Shortfall then exists or has occurred during the preceding twelve-(12)-month period (or, if such an event has occurred, stating that such event has occurred and whether such event continues to exist), (ii) a list of all then existing Credit Agreements, and (iii) a copy of Tenant's most recent annual and quarterly financial statements, whether then audited or unaudited, evidencing Tenant's Net Worth as of the date of such financial statements. If requested from time to time by Landlord, on an annual basis, from the Lease Date until the end of the Term, Tenant shall also provide Landlord with a copy of its annual financial statements for the fiscal year just ended (which, if Tenant does not have audited statements, may be unaudited so long as Guarantor remains a publicly traded company and a Related Entity of the then Tenant under the Lease and which shall be audited if Guarantor is no longer a publicly traded company or no longer a Related Entity of the then Tenant under the Lease), within ten (10) business days after the issuance of such financial statements.
- (e) <u>Restoration of Initial LC Amount</u>. In the event that the Letter of Credit is increased due to an LC Increase Trigger Event as set forth above, Landlord shall thereafter permit Tenant to reduce the Letter of Credit to the Initial LC Amount, if and only if Tenant, at any time after the date that is six (6) months after the delivery of the Letter of Credit for the Increased LC Amount, submits to Landlord evidence reasonably satisfactory to Landlord that Tenant then satisfies the following conditions (any such submission being referred to herein as a "<u>Reduction Submission</u>"):
 - (i) in the case of an LC Increase Trigger Event that arose from a Credit Agreement Default, that no Credit Agreement Default then exists or has existed at any time during the six-(6)-month period immediately preceding the Reduction Submission;
 - (ii) in the case of an LC Increase Trigger Event that arose from a Net Worth Shortfall, that (x) for the two consecutive quarterly reporting periods immediately preceding the Reduction Submission, Tenant's financial statements (which shall be prepared in a manner consistent with the financial statements filed by Guarantor with the SEC) show that the reported Net Worth of Tenant equals or exceeds \$400,000,000 and (y) Tenant's audited financial statements for the most recent calendar year (which shall be prepared in a manner consistent with the financial statements filed by Guarantor with the SEC) subsequent to the occurrence of the Net Worth Shortfall reflect a Tenant's Net Worth balance of at least \$400,000,000; and

(iii) in the case of either clause (i) or (ii) above, that (x) no draw on the Letter of Credit has occurred within the six-(6)-month-period immediately preceding the Reduction Submission and (ii) no Material Event of Default then exists and no event has occurred or condition then exists which with notice and the passage of time would constitute such a Material Event of Default.

If the LC Amount is reduced to the Initial LC Amount under this <u>Section 14.6(e)</u>, the provisions of <u>Section 14.6(b)</u> shall again apply in the event of any subsequent LC Increase Trigger Event.

- 14.7. <u>Guaranty</u>. Simultaneously with Tenant's execution of this Lease, Tenant shall cause Guarantor to execute and deliver to Landlord the Guaranty in the form attached hereto as <u>Exhibit M</u>.
- 14.8. Holdover. If Tenant (or anyone claiming through Tenant) shall remain in occupancy of the Premises or any part thereof after the expiration or earlier termination of the Term (without a written agreement therefor executed and delivered by Landlord), then, without limiting Landlord's other rights and remedies, the person remaining in possession shall be deemed a tenant at sufferance, and Tenant shall thereafter pay to Landlord monthly rent (pro-rated for such portion of any partial month as Tenant shall remain in possession) at a rate equal to 125% for up to the first thirty (30) days of such holdover (and 150% for any holdover period thereafter) of the greater of (a) the Base Rent applicable to the Premises during the last monthly period immediately preceding such expiration or termination or (b) the fair market rent for the Premises (on a so-called "triple net" basis), in each case with all Additional Rent also payable as provided in this Lease. The foregoing provisions shall not serve as permission for Tenant or anyone claiming by, through, or under Tenant to holdover, nor serve to extend the Term (although Tenant shall remain bound to comply with all provisions of this Lease until Tenant vacates the Premises), and Landlord shall have the right at any time after the expiration or earlier termination of this Lease to enter and possess the Premises and remove all property and persons thereform or to require Tenant to surrender possession of the Premises as provided in this Lease upon the expiration or earlier termination of the Term. If, within forty (45) days after the expiration or earlier termination of this Lease, Tenant fails to vacate and surrender the Premises as required under this Lease, Tenant shall indemnify, defend and hold harmless Landlord from all costs, loss, expense or liability arising from such failure, including, without limitation, claims made by any succeeding tenant and real estate brokers' claims and attorneys' fees. No acceptance by Landlord of any Rent during or for any period fol

ARTICLE 15. PROTECTION OF LENDERS

15.1 <u>Subordination and Superiority of Lease</u>. Landlord represents and warrants that, as of the Lease Date, there is no mortgage on the Property or the Project Land to which this Lease is subject. Tenant agrees that this Lease and the rights of Tenant hereunder will be subject and subordinate to any lien of the holder of any future mortgage, and to the rights of any lessor under any ground or improvements lease, of the Building or the Property (all mortgages and ground or improvements leases of any priority are collectively referred to in this Lease as a "mortgage," and the holder or lessor thereof from time to time as a "mortgagee"), and to all advances and interest thereunder and all modifications, renewals, extensions and consolidations thereof; provided that any subordination of this Lease shall, if no Event of Default then exists, be conditioned upon Landlord delivering to Tenant a written, recordable subordination, non-disturbance and attornment agreement from the mortgagee seeking to have this Lease subordinated to its interest in the form attached as Exhibit H or in such other substantially equivalent form as such mortgage may reasonably request. Such subordination, non-disturbance and attornment agreement shall include, among other things, provisions for the application of casualty insurance proceeds for the restoration of the Building substantially in the form set forth in Section 4.3 of Exhibit H. Tenant shall not be required to execute any subordination, non-disturbance and attornment agreement and this Lease shall not be subordinate to any junior mortgage where a mortgagee having priority over such junior mortgage has prohibited execution of a further subordination, nondisturbance and attornment agreement with Tenant and has not consented to Tenant so executing a subordination, nondisturbance and attornment agreement with respect to such junior mortgage.

Tenant agrees that this Lease shall survive the merger of estates of any ground or improvements lessor and lessee, if any. Until a mortgagee (either superior or subordinate to this Lease) forecloses Landlord's equity of redemption (or terminates or succeeds to a new lease in the case of a ground or improvements lease) no mortgagee shall be liable for failure to perform any of Landlord's obligations (and such mortgagee shall thereafter be liable only after it succeeds to and holds Landlord's interest and then only as limited herein). Tenant shall, if requested by Landlord or any mortgagee, give notice of any alleged non-performance on the part of Landlord to any such mortgagee, provided that an address for such mortgagee has been designated to Tenant in writing, and Tenant agrees that such mortgagee shall have a separate, consecutive reasonable cure period of no less than thirty (30) days (to be reasonably extended in the same manner Landlord's cure period is to be extended and for such additional periods as is necessary to allow such Mortgagee to take possession of the Property) following Landlord's cure period during which such mortgagee may, but need not, cure any non-performance by Landlord. The agreements in this Lease with respect to the rights and powers of a mortgagee constitute a continuing offer to any person that may be accepted by taking a mortgage (or entering into a ground or improvements lease) of the Premises. This Section 15.1 shall be self-operative, but in confirmation thereof, Tenant shall execute and deliver the subordination, nondisturbance and attornment agreement in the form of Exhibit H or in such other substantially equivalent form as such mortgagee may reasonably request.

15.2 Attornment. If Landlord's interest in the Property is acquired by mortgagee or purchaser at a foreclosure sale, Tenant shall, at the election of such mortgagee or purchaser (except as may be required pursuant to any applicable subordination, non-disturbance and attornment agreement then in effect), attorn to the transferee of or successor to Landlord's interest in the Property and recognize it as Landlord under this Lease. Tenant waives the protection of any statute or rule of law which gives Tenant any right to terminate this Lease or surrender possession of the Premises upon the transfer of Landlord's interest. Upon such

attornment, this Lease shall continue in full force and effect as a direct lease between the mortgagee and Tenant upon all of the terms, conditions and covenants as are set forth in this Lease, except that the mortgagee shall not (except as may be required pursuant to any applicable subordination, non-disturbance and attornment agreement then in effect) be (i) liable in any way to Tenant for any act or omission, neglect or default on the part of Landlord under this Lease (nothing in this clause (i) being deemed to relieve any mortgagee succeeding to the interest of Landlord hereunder of its continuing obligations as landlord under this Lease from and after the date of such succession), (ii) responsible for any monies owing by or on deposit with Landlord to the credit of Tenant (except to the extent any such deposit is actually received by such mortgagee), (iii) subject to any counterclaim or setoff which theretofore accrued to Tenant against Landlord, (iv) bound by any amendment or modification of this Lease subsequent to such mortgage, or by any previous prepayment of Rent for more than one (1) month, which was not approved in writing by the mortgagee, or bound by the representations set forth in Section 9.4, (v) liable beyond mortgagee's interest in the Property, (vi) responsible for the performance of any work to be done by the Landlord under this Lease to render the Premises ready for occupancy by the Tenant or the payment of the TI Allowance, or (vii) required to remove any person occupying the Premises or any part thereof, except if such person claims under the mortgagee. Notwithstanding the foregoing, for any subordination, non-disturbance and attornment agreement delivered prior to completion of Landlord's construction-related obligations for the Building under Exhibit B, any such agreement shall include provisions, substantially in the form set forth in Section 4.4(a) of Exhibit H, that recognize Tenant's rent abatement remedies, term commencement delay remedies, off-set rights, and termination rights under Paragraphs 11.2(c), 12.2, and 12.3 of Exhibit B. Tenant agrees that any present or future mortgagee may at its option unilaterally elect to subordinate, in whole or in part and by instrument in form and substance satisfactory to such mortgagee alone, the lien of its mortgagee (or the priority of its ground lease) to some or all provisions of this Lease.

15.3 <u>Rent Assignment</u>. If from time to time Landlord assigns this Lease or the rents payable hereunder to any mortgagee, whether such assignment is conditional in nature or otherwise, such assignment shall not be deemed an assumption by the assignee of any obligations of Landlord; but, subject to the limitations herein including <u>Section 15.1</u> and <u>16.4</u>, the assignee shall be responsible only for non-performance of Landlord's obligations that occur after it succeeds to, and only during the period it holds possession of, Landlord's interest in the Property after foreclosure or voluntary deed in lieu of foreclosure, subject to the provisions of <u>Section 15.2</u> above.

15.4 Other Instruments. The provisions of this Article 15 shall be self-operative; nevertheless, Tenant agrees to execute, acknowledge and deliver any subordination, attornment or priority agreements conforming to the provisions of this Article 15 from time to time requested by Landlord or any mortgagee, and further agrees that its failure to do so within ten (10) business days after written request shall be an Event of Default. Without limitation, where Tenant in this Lease has agreed to indemnify or otherwise make covenants for the benefit of Landlord's mortgagee, such agreements are for the benefit of such mortgagee as third party beneficiaries; and at the request of Landlord or such mortgagee, Tenant from time to time will confirm such matters in a confirmatory instrument, in a commercially reasonable form consistent with the provisions of this Lease, directly with such mortgagee.

15.5 Estoppel Certificates. Within ten (10) business days after request by a party to this Lease, the other party shall execute, acknowledge and deliver a written statement certifying: (i) that none of the terms or provisions of this Lease has been changed (or if they have been changed, stating how); (ii) that this Lease has not been canceled or terminated; (iii) the last date of payment of Base Rent and other charges and the time period covered; (iv) that to the knowledge of the party executing the certificate, the party requesting such certificate is not in default under this Lease (or, if in default, describing it in reasonable detail); and (v) such other information with respect this Lease as may be reasonably requested or which any prospective purchaser or encumbrancer of the Property may require. Any certificate delivered under this Section 15.5 shall be for the benefit of specified third parties and their respective successors and assigns, and neither party to this Lease shall have the right to rely on a certification delivered under this Section 15.5 by the other party to this Lease, except to the extent that Landlord relies on such statements when making any certifications, representations or warranties to such third parties. The party receiving any such statement may deliver the statement to any such prospective purchaser or encumbrancer, which may rely conclusively upon such statement as true and correct.

ARTICLE 16. MISCELLANEOUS PROVISIONS

- 16.1 <u>Landlord's and Tenant's Consent</u>. Tenant shall pay to Landlord, as Additional Rent, all reasonable out-of-pocket third-party fees and expenses incurred in connection with any act or request by Tenant that requires Landlord's consent or approval under this Lease during the Term. Landlord shall reimburse Tenant for all reasonable out-of-pocket third-party fees and expenses incurred in connection with any act or request by Landlord that requires Tenant's consent or approval under this Lease during the Term. The provisions of this <u>Section 16.1</u> shall not apply to any matters arising under the Work Letter attached hereto as <u>Exhibit B</u>.
- 16.2 Notice of Landlord's Default. Tenant shall give notice of Landlord's failure to perform any of its obligations under this Lease to Landlord and any mortgagee whose name and address have been given to Tenant. Landlord shall not be in default under this Lease unless Landlord (or such mortgagee or beneficiary) fails to cure such non-performance within thirty (30) days after receipt of Tenant's notice, provided that, if such non-performance requires more than thirty (30) days to cure, such period shall be reasonably extended in the case of any such non-performance that cannot be cured by the payment of money where such non-performance can be cured (but in any event shall not exceed 180 days in the aggregate) and Landlord begins promptly within said thirty (30) day period and thereafter diligently completes the cure. In addition to the other remedies expressly provided in this Lease, Tenant shall be entitled to the restraint by injunction of the violation or attempted or threatened violation of any of the covenants, conditions or provisions of this Lease by Landlord or to a decree compelling specific performance of any such covenants, conditions or provisions. No termination remedy that is not expressly set forth in this Lease for any breach or failure by Landlord to perform any obligation under this Lease shall be implied or applicable as a matter of law.
- 16.3 Quiet Enjoyment. Landlord agrees that, so long as Tenant performs all of its obligations under the terms of this Lease within any applicable notice and cure periods, Tenant shall lawfully and quietly hold, occupy and enjoy the Premises during the Term of this Lease without disturbance by Landlord or by any person claiming through or under Landlord, subject to the terms of this Lease.

16.4. Limitations on Liability.

- (a) Landlord's Interest in the Property. Tenant agrees that Landlord shall be liable only for breaches of its covenants occurring while it is owner of the Property (provided, however, that if Landlord from time to time is lessee of the ground or improvements constituting the Building, then Landlord's period of ownership of the Property shall be deemed to mean only that period while Landlord holds such leasehold interest). Upon any sale or transfer of the Property (or Landlord's interest as ground lessee, as applicable), the transferor Landlord (including any mortgagee) shall be freed of any liability or obligation thereafter arising to the extent that such liabilities and obligations are assumed by such transferee and, thereafter, Tenant shall look solely to the transferee Landlord as aforesaid for satisfaction of such liability or obligation. Tenant (and any person acting under or through Tenant) agrees to look solely to Landlord's interest from time to time in the Property, including the rents, insurance proceeds and condemnation proceeds therefrom, for satisfaction of any claim against Landlord. No owner, trustee, beneficiary, partner, member, manager, agent, or employee of Landlord (or of any mortgagee or any lender or ground or improvements lessor) nor any person acting under any of them shall ever be personally or individually liable to Tenant or any person claiming under or through Tenant for or on account of any default by Landlord or failure by Landlord to perform any of its obligations hereunder, or for or on account of any amount or obligations that may be or become due under or in connection with this Lease or the Premises; nor shall it or they ever be answerable or liable in any equitable judicial proceeding or order beyond the extent of their interest in the Property. No deficit capital account of any member or partner of Landlord shall be deemed to be a liability of such member or partner or an asset of Landlord.
- (b) <u>No Indirect or Consequential Damages</u>. In no event shall Landlord or Tenant ever be liable to the other for indirect or consequential damages (including loss of revenue, profits, or data); provided, however, that the provisions of this <u>Section 16.4</u> shall not apply to <u>Section 9.4</u> or to the remedies expressly set forth in <u>Article 14</u> of this Lease.
- 16.5. Notices. All notices, requests and other communications required under this Lease shall be in writing, addressed as specified in Section 1.12, and shall be (i) personally delivered, (ii) sent by certified mail, return receipt requested, postage prepaid, or (iii) delivered by a national overnight delivery service that maintains delivery records. All notices shall be effective upon delivery (or refusal to accept delivery). Either party may change its notice addresses under Section 1.12 upon written notice to the other party. Notices under this Lease may be given by counsel for either party.
- 16.6. Notice of Lease. Tenant shall not record this Lease. Either Landlord or Tenant may require that a statutory notice of lease executed by both parties be recorded with respect to this Lease, which will include (as provided in Section 2.4) notice of Tenant's Extension Option rights and Expansion/Purchase Rights. In the event that any rights under this Lease (including, without limitation, any Extension Option rights or Expansion/Purchase Rights) shall expire, lapse, or be terminated in accordance with the applicable terms set forth herein, upon the request of Landlord (or, as applicable, the Project Declarant or its respective successors or assigns), Tenant shall execute and deliver a recordable instrument evidencing the expiration, lapse, or other applicable termination of such rights.

16.7. Corporate Authority. Tenant warrants and represents that (a) Tenant is duly organized, validly existing and in good standing under the laws of the jurisdiction in which such entity was organized; (b) Tenant has the authority to own its property and to carry on its business as contemplated under this Lease; (c) Tenant has duly executed and delivered this Lease; (d) the execution, delivery and performance by Tenant of this Lease (i) are within the powers of Tenant, (ii) have been duly authorized by all requisite action, (iii) will not violate any provision of law or any order of any court or agency of government, or any agreement or other instrument to which Tenant is a party or by which it or any of its property is bound, and (iv) will not result in the imposition of any lien or charge on any of Tenant's property, except by the provisions of this Lease; and (e) this Lease is a valid and binding obligation of Tenant in accordance with its terms. This warranty and representation shall survive the termination of the Term.

Landlord represents and warrants that (a) Landlord is duly organized, validly existing and in good standing under the laws of the jurisdiction in which such entity was organized; (b) Landlord has the authority to own its property and to carry on its business as contemplated under this Lease; (c) Landlord has duly executed and delivered this Lease; (d) the execution, delivery and performance by Landlord of this Lease (i) are within the powers of Landlord, (ii) have been duly authorized by all requisite action, (iii) will not violate any provisions of law or any order of any court or agency of government, or any agreement or other instrument to which Landlord is a party or by which it or any of its property is bound, and (iv) will not result in the imposition of any lien or charge on any of Landlord's property, except by the provisions of this Lease; and (e) this Lease is a valid and binding obligation of Landlord in accordance with its terms. This warranty and representation shall survive the termination of the Term.

- 16.8. Joint and Several Liability. If more than one party signs this Lease as Tenant, they shall be jointly and severally liable for all obligations of Tenant.
- 16.9. Force Majeure. Except where Force Majeure (as defined below) is expressly excluded elsewhere in this Lease, if a party cannot perform any of its obligations due to an event of Force Majeure (other than the inability to make payments when due), the time provided for performing such obligations shall be extended by a period of time equal to the duration of the events. "Force Majeure" shall mean any cause that is beyond the reasonable control of a party and beyond the reasonable control of such party's contractors, consultants or other agents or representatives, including acts of God; fire, explosion or other casualty; strikes, lockouts, labor disputes, labor shortages or material shortages; acts of terrorism or war; acts of public enemy; riots or civil disobedience; governmental acts or orders (as distinguished from inability to obtain permits in the ordinary course); blackouts or power shortages; epidemics; landslides; earthquakes; hurricanes and/or tornadoes; and floods.
- 16.10. <u>Limitation of Warranties</u>. Landlord and Tenant expressly agree that, other than those warranties expressly set forth in this Lease, there are and shall be no implied warranties of merchantability, habitability, suitability, fitness for a particular purpose or of any other kind arising out of this Lease.

16.11. <u>Brokers.</u> Landlord and Tenant represent and warrant to each other that the Broker(s) named in <u>Section 1.13</u> are the only agents, brokers, finders or other parties with whom such party has dealt who may be entitled to any commission or fee with respect to this Lease or the Premises or the Property. Each of Landlord and Tenant agrees to indemnify and hold the other harmless from any claim, demand, cost or liability, including without limitation attorneys' fees and expenses, asserted by any party other than the Brokers named in <u>Section 1.13</u> based upon dealings of that party with the indemnifying party. Landlord shall be responsible for the payment of any brokerage fees to the Brokers named in <u>Section 1.13</u> pursuant to separate agreements with Landlord. The provisions of this Section shall survive the expiration of the Term or any earlier termination of this Lease.

16.12. Applicable Law and Construction. This Lease may be executed in one or more counterpart signature pages (all of which when signed shall constitute a single agreement), shall be construed as a sealed instrument, and shall be governed exclusively by the provisions hereof and by the laws of The Commonwealth of Massachusetts without regard to principles of choice of law or conflicts of law. A facsimile signature to this Lease shall be sufficient to prove the execution by a party. If any provisions shall to any extent be invalid, the remainder shall not be affected. Other than contemporaneous instruments executed and delivered of even date, if any, this Lease contains all of the agreements between Landlord and Tenant relating in any way to the Premises and supersedes all prior agreements and dealings between them. There are no oral agreements between Landlord and Tenant relating to this Lease or the Premises. This Lease may be amended only by instrument in writing executed and delivered by both Landlord and Tenant (and, as to any matters set forth in Section 2.4, Project Declarant or its respective successors and assigns as set forth therein). The provisions of this Lease shall bind Landlord and Tenant and their respective successors and assigns, and shall inure to the benefit of Landlord and its successors and assigns and of Tenant and its permitted successors and assigns, subject to Article 13. The titles are for convenience only and shall not be considered a part of this Lease. This Lease shall not be construed more strictly against one party than against the other merely by virtue of the fact that it may have been prepared primarily by counsel for one of the parties, it being recognized that both Landlord and Tenant have contributed substantially and materially to the preparation of this Lease. If Tenant is granted any extension or other option, to be effective the exercise (and notice thereof) shall be unconditional; and if Tenant purports to condition the exercise of any option or to vary its terms in any manner, then the purported exercise shall be ineffective. The enumeration of specific examples of a general provision shall not be construed as a limitation of the general provision, and the use of "including" shall be deemed to mean "including without limitation". Unless a party's approval or consent is required by the express terms of this Lease not to be unreasonably withheld, such approval or consent may be withheld in the party's sole discretion. The submission of a form of this Lease or any summary of its terms shall not constitute an offer by Landlord to Tenant; but a leasehold shall only be created and the parties bound when this Lease is executed and delivered by both Landlord and Tenant, together with the Project Declarant. Nothing herein shall be construed as creating the relationship between Landlord and Tenant of principal and agent, or of partners or joint venturers, or any relationship other than landlord and tenant. This Lease and all consents, notices, approvals and all other related documents may be reproduced by any party by any electronic means or by facsimile, photographic, microfilm, microfiche or other reproduction process and the originals may be destroyed; and each party agrees that any reproductions shall be as admissible in evidence in any judicial or administrative proceeding as the original itself

(whether or not the original is in existence and whether or not reproduction was made in the regular course of business), and that any further reproduction of such reproduction shall likewise be admissible. If any payment in the nature of interest provided for in this Lease shall exceed the maximum interest permitted under controlling law, as established by final judgment of a court, then such interest shall instead be at the maximum permitted interest rate as established by such judgment.

16.13. Construction on the Property or Adjacent Property.

Tenant acknowledges that Project Declarant, Landlord, and/or its affiliates is or are undertaking or may hereafter undertake major renovations and/or construction at the Project pursuant to the Project Condominium Documents. For avoidance of doubt, in connection therewith, Project Declarant expressly reserves the right from time to time to amend the Project Site Plan, any Master Plan Permits, or any other Project Documents in accordance with the rights reserved to Project Declarant under the Project Condominium Documents with respect to any future development, redevelopment, alteration, improvement, operation, maintenance of the Project, provided that such amendment does not (1) materially adversely affect the use of the Premises for their intended purposes (or with access thereto or the appurtenant parking rights for the Building under Section 2.2(b) for their intended purposes) in each case as more particularly set forth under the terms and conditions of this Lease or (2) conflict with the Building 2 Rights (including, without limitation, any amendment to the Special Permit Amendment that conflicts with the rights for Building 2 granted to Tenant under Exhibit C-1 and Exhibit C-2) and Building 4 Rights under their respective express terms for so long as such respective rights have not lapsed or been terminated in accordance with their respective terms. Project Declarant and Landlord shall have the right, in connection with Project Declarant's development, redevelopment, alteration, improvement, operation, maintenance, or repair of the Project, to subject the Property and its appurtenant rights to easements for the construction, reconstruction, alteration, improvement, operation, repair or maintenance of elements thereof, for access and egress, for parking, for the installation, maintenance, repair, replacement or relocation of utilities serving the Project and to subject the Property to such other rights, agreements, and covenants for such purposes as Landlord may determine; provided that such rights, agreements, and covenants do not materially adversely affect the use of the Premises for their intended purposes (or with access thereto or parking rights granted hereunder for their intended purposes) in each case as more particularly set forth under the terms and conditions of this Lease. Tenant hereby agrees that this Lease shall be subject and subordinate to any such matters that do not materially adversely affect the use of the Premises, access thereto, or parking rights granted hereunder for their intended purposes (in each case as more particularly set forth under the terms and conditions of this Lease). Neither Tenant nor any Related Entity shall take any action, directly or indirectly, to oppose the Project or any phase thereof, to the extent the same are not in violation of this Section 16.13. In the event of any municipal action under the Special Permit Amendment that seeks to revoke occupancy rights for the Building due to a failure to comply with applicable conditions of the Special Permit Amendment (other than as may have arisen in whole or in part from an act or omission by Tenant or any Tenant Party), Landlord will use diligent efforts to avert the application to the Building of any such revocation action, including if appropriate taking measures to remedy any such conditions that are within Landlord's reasonable control on the Building Site or the seeking of appropriate injunctive relief against the party responsible for causing or curing such failure in other portions of the Project.

Project Declarant and Landlord and their respective affiliates and their respective agents, employees, licensees and contractors shall also have the right to enter on the Property or Building (or any Appurtenant Exterior Areas serving the Building) to undertake work pursuant to any easement granted pursuant to the above paragraph; to shore up the foundations and/or walls of the Building or any Appurtenant Exterior Area; to erect scaffolding and protective barricades around, within or adjacent to the Building or any Appurtenant Exterior Area; and to do any other act necessary for the safety of the Building or any Appurtenant Exterior Area or the expeditious completion of such work. Landlord shall not be liable to Tenant for any compensation or reduction of Rent by reason of inconvenience or annoyance or for loss of business resulting from any act by Landlord pursuant to this Section 16.13 provided that Landlord complies with this Section 16.13. Landlord shall use reasonable efforts to minimize the extent and duration of any inconvenience, annoyance or disturbance to Tenant resulting from any work pursuant to this Section 16.13 in or about the Building or any Appurtenant Exterior Area, consistent with accepted construction practice. Landlord shall inform Tenant and work reasonably with Tenant to create an operating plan to ensure the minimization of inconvenience, annoyance and disturbance to Tenant and shall continue such consultative process throughout the course of completion of the work.

16.14. Confidentiality of Information.

Landlord agrees to hold any proprietary information identified as confidential by Tenant in writing and supplied to Landlord pursuant to this Lease, excluding any information required to be filed with a governmental agency ("Confidential Information") in confidence. Notwithstanding the foregoing, Landlord may disclose such Confidential Information to its attorneys, accountants, property managers, real estate brokers, current or prospective purchasers, investors, or lenders, or their respective attorneys and consultants in connection with the financing or sale of the Property or Landlord's review of such information to the extent such parties need to know the Confidential Information for the purpose of evaluating the proposed transaction and Landlord informs such parties of the confidential nature of the Confidential Information. The term "Confidential Information" does not include information that (i) is publicly known at the time of delivery, (ii) subsequently becomes publicly known through no breach of this Section 16.14 by Landlord or its representatives, (iii) Landlord can demonstrate was in its possession at the time of disclosure and was not acquired by it directly or indirectly from Tenant on a confidential basis, (iv) becomes available to Landlord on a non-confidential basis from a source other than the Tenant and which source, to the best of Landlord's knowledge, is not under an obligation of confidence to Tenant or (v) is disclosed in the course of litigation between Landlord and Tenant or Landlord and any other third party.

[BALANCE OF PAGE HAS BEEN INTENTIONALLY LEFT BLANK.]

IN WITNESS WHEREOF, the undersigned have caused this Lease to be executed as of the Lease Date set forth above.

LANDLORD:

NORMANDY GAP-V NEEDHAM BUILDING 3, LLC, a Delaware limited liability company

By: /s/ David T. Welsh
Name: David T. Welsh
Title: Authorized Person

TENANT:

TRIPADVISOR LLC,

a Delaware limited liability company

By: /s/ Julie M.B. Bradley
Name: Julie M.B. Bradley
Title: Authorized Person

JOINDER BY PROJECT DECLARANT

The undersigned, in its capacity as the Project Declarant under the Project Condominium Documents, hereby joins in this Lease solely for the purposes set forth in Section 2.4(b) and Exhibits C-1, C-2, C-3, C-4, and \underline{N} .

PROJECT DECLARANT:

NORMANDY GAP-V DEVELOPMENT NEEDHAM, LLC, a Delaware limited liability company

By: /s/ David T. Welsh
Name: David T. Welsh

Title: Authorized Person

GUARANTY

This Guaranty is executed as of June 20, 2013 by the undersigned TripAdvisor, Inc., a Delaware corporation (the "<u>Guarantor</u>"), for the benefit of Normandy GAP-V Needham Building 3, LLC, a Delaware limited liability company, and its successors and assigns as set forth below.

- A. TripAdvisor LLC, a Delaware limited liability company, as Tenant, and Normandy GAP-V Needham Building 3, LLC, a Delaware limited liability company, as Landlord, are entering into that certain Lease of even date herewith (as the same may from time to time hereafter be amended or assigned, the "Lease") for certain premises in the office park known as Center 128, Needham, Massachusetts. Capitalized terms used and not defined in this Guaranty shall have the meanings set forth in the Lease.
- B. Guarantor has advised Landlord that Tenant is a wholly owned subsidiary of TripAdvisor Holdings LLC, a Delaware limited liability company, and that TripAdvisor Holdings LLC is a wholly owned subsidiary of Guarantor, a publicly traded company.
- C. Landlord has required the execution and delivery of this Guaranty as a condition of Landlord entering into the Lease, and Guarantor has provided this Guaranty as a material inducement for Landlord to enter into and execute the Lease, with the intention that Guarantor shall guarantee the payment and performance of all of Tenant's obligations under the Lease as set forth herein.

NOW, THEREFORE, for the foregoing consideration and other good and valuable consideration, the receipt and legal sufficiency of which are hereby acknowledged, Guarantor hereby agrees as follows:

1. <u>Guaranty.</u> Guarantor hereby unconditionally and irrevocably guarantees to Landlord the full payment and performance of the Guaranteed Lease Obligations (as defined below), as and when the same shall be due and payable or required to be performed under the terms of the Lease. The "<u>Guaranteed Lease Obligations</u>" shall mean the prompt, timely, and complete payment and performance of all of the covenants, conditions, and agreements under the Lease to be performed and observed by Tenant. The Guaranteed Lease Obligations shall expressly include, without limitation, all covenants, conditions, and agreements by Tenant under the Lease in respect of the entire Term of the Lease (as the same may from time to time be extended or renewed) and for any holdover term thereafter, for the entire Premises and its appurtenances thereunder (as the same may from time to time be expanded, contracted, relocated, subleased or licensed in whole or in part, and/or assigned, whether voluntarily or otherwise, and whether or not Landlord has consented to or approved any of the foregoing matters), and all obligations under the Lease that survive the expiration or earlier termination of the Lease as provided in the Lease. Guarantor's obligations under this Guaranty for the Guaranteed Lease Obligations shall survive the expiration or earlier termination of the Lease.

Without limiting any rights or remedies that may be exercised by Landlord under the Lease or this Guaranty, if after the expiration or earlier termination of the Lease Landlord elects to demand performance by Guarantor of any non-monetary Guaranteed Lease Obligation that requires access to the premises under the Lease in order to perform the applicable work for such

non-monetary Guaranteed Lease Obligation (such as repair work or restoration work that Tenant failed to perform) and Tenant no longer has possession of or any right of access to the premises under the Lease due to the expiration or earlier termination of the Lease, then Guarantor's obligation to undertake the performance of such work shall be conditioned on Landlord permitting Guarantor to access the applicable portion of the premises for the limited purpose of performing such work in accordance with the terms of the Lease that would have applied in the absence of such expiration or termination of the Lease, provided that nothing in this sentence shall require Landlord to grant any access rights to Guarantor of any kind (which may be permitted or denied in Landlord's sole discretion) or to limit Guarantor's liability under this Guaranty for any monetary Guaranteed Lease Obligations (including without limitation any monetary Guaranteed Lease Obligations that may arise from expenses and costs incurred by Landlord on account of any failure by Tenant to perform to any non-monetary Guaranteed Lease Obligation).

- 2. <u>Payment by Guarantor</u>. If all or any part of the Guaranteed Lease Obligations shall not be punctually paid when due, Guarantor shall, immediately upon demand by Landlord, and without presentment, protest, notice of protest, notice of non-payment, or any other notice whatsoever, pay to Landlord, in lawful money of the United States of America, the amount of the Guaranteed Lease Obligations due to the Landlord submitting such demand, at the address for Landlord provided below. Such demand(s) may be made at any time and from time to time, either coincident with or after the time for payment of all or part of the Guaranteed Lease Obligations, and may be made from time to time with respect to the same or different items of Guaranteed Lease Obligations. Such demand shall be deemed made, given, and received in accordance with the notice provisions set forth below.
- 3. No Impairment of Guaranteed Obligations. Guarantor expressly agrees that the validity of this Guaranty, the Guaranteed Lease Obligations, and the obligations of Guarantor under this Guaranty shall not be terminated, diminished, discharged, released, or in any way impaired by reason of (a) any amendment to the Lease or any assignment, sublease, or other transfer of any interest in the Lease (Guarantor hereby agreeing that its obligations under this Guaranty shall continue in full force and effect with respect to the Lease as the same may hereafter be amended, assigned, subleased, or otherwise transferred from time to time), or (b) the assertion by Landlord against Tenant of any of the rights or remedies reserved to Landlord pursuant to the provisions of the Lease, or (c) any delay, waiver, or failure by Landlord to enforce any of the terms, covenants, or conditions of the Lease, this Guaranty, or any other guaranty of the Lease (if any) by any other party, or (d) the granting of any indulgence, release, forbearance, or extension to Tenant, or to Guarantor, or to any other guarantor (if any), all of which may be given or done by Landlord from time to time without notice to Guarantor, or (e) any sale or other transfer by Landlord of the Property or any interest therein, including without limitation any assignment or collateral assignment of Landlord's interest in the Lease in connection with any sale or financing of the Property or any interest therein, or (f) any relief or discharge of Tenant from, or any limitation of Tenant's liability for, any of Tenant's obligations under the Lease by operation of law or otherwise, including, without limitation, in connection with any proceedings under the bankruptcy laws now or hereafter enacted or similar laws for the relief of debtors, any rejection or termination of the Lease in any such proceedings, and any limitation of Tenant's liability for its obligations or damages under the Lease under such bankruptcy or similar laws. Guarantor hereby waives notice of non-paymen

other amounts to be paid by Tenant under the Lease and waives notice of default or non-performance of any of Tenant's other covenants, conditions, and agreements contained in the Lease. Guarantor further waives, to the fullest extent permitted by law, any and all legal, equitable, and/or surety defenses to which Guarantor might otherwise be entitled other than (i) that Guarantor has fully performed all of its obligations under this Guaranty and (ii) that Tenant has fully and timely performed all of its obligations under the Lease (which obligations of Tenant, for purposes of this Guaranty, shall be determined without regard to any relief or discharge of Tenant from, and without regard to any limitation on Tenant's liability for, its obligations under the Lease by operation of law or otherwise, including without limitation under any proceedings under the bankruptcy laws or similar laws for the relief of debtors). In the event of the avoidance, disgorgement, reduction, reconveyance, or recovery of any payment made by Tenant to Landlord as a preference under any laws relating to the bankruptcy, reorganization, or liquidation of debtors, or as a so-called fraudulent conveyance or fraudulent transfer, or under any other applicable law, Landlord shall be entitled to recover on demand the amount of such payment from Guarantor as if such payment had never been made by Tenant.

- 4. Nature of Obligations. Guarantor agrees that its liability under this Guaranty shall be primary, direct, and unconditional, and that such liability shall be joint and several with Tenant, any other guarantor (if any) of Tenant's obligations under the Lease, and any other party that may be liable for Tenant's obligations under the Lease. In any right of action that shall from time to time accrue to Landlord under the Lease, Landlord may, at its option, proceed against Tenant, Guarantor, or such other party, either separately or concurrently, in such order and in such proceedings as Landlord may from time to time determine. Without limiting the generality of the foregoing, Landlord may, at its option and without limiting any claim against Tenant or any other party, proceed against Guarantor, without having commenced any action or having obtained any judgment against Tenant, any other guarantor, or any other party liable for Tenant's obligations under the Lease. Guarantor hereby waives any right that Guarantor may have to require Landlord to enforce, pursue, or exhaust any remedies that Landlord may have against Tenant or any other party, including as to any security deposit or other collateral that may have been given to secure the Guaranteed Lease Obligations, or to mitigate damages or to take any other action to reduce, collect, or enforce the Guaranteed Lease Obligations. In any such proceeding, the books and records of Landlord in respect of the status of Tenant's obligations under the Lease (including, without limitation, delinquent rent reports) shall constitute prima facie evidence of the Guaranteed Lease Obligations for purposes of this Guaranty.
- 5. Subordination of Claims against Tenant. Guarantor further agrees that, to the maximum extent permissible (i.e., so as not to trigger a breach of any covenants) under any then applicable debt financing to which Guarantor is a party (including, without limitation, the Credit Agreement (as defined in the Lease)), any and all claims of any nature that Guarantor or any of its affiliates may now or hereafter have against Tenant shall be subordinated to the full and final cash payment to Landlord of all obligations under the Lease and under this Guaranty. Without limiting the generality of, and subject to, the foregoing, Guarantor agrees that, if and for so long as Tenant shall have failed to satisfy any obligation under the Lease beyond any applicable notice and cure period, then prior to the full and final cash payment to Landlord of all obligations under the Lease and under this Guaranty, Guarantor shall not itself (or permit or suffer any affiliate of Guarantor to do any of the following): (i) make any claim of liability of Tenant to Guarantor or any of its affiliates (or assert any set-off or counterclaim against Tenant) by reason

of paying any sum due or recoverable under this Guaranty (whether or not demanded by Landlord) or under the Lease, or by any other means or on any other ground; or (ii) attempt to prove in competition with Landlord any claim regarding any payment made under this Guaranty or under the Lease. Until such time as the Guaranteed Lease Obligations shall have been paid in full, Guarantor waives any rights of subrogation for any claims that Landlord may have against Tenant on account of any payment made by Guarantor to Landlord under this Guaranty. In furtherance of the provisions of this paragraph, Guarantor promptly will execute such instruments as Landlord may from time to time reasonably request to preserve for Landlord the benefits of the foregoing provisions.

- 6. Enforcement Costs and Expenses. In the event that Guarantor shall fail to timely perform any provisions of this Guaranty, Guarantor shall pay to Landlord, upon demand, (a) all costs and expenses, including without limitation reasonable attorneys' fees and court costs, incurred by Landlord in connection with such failure and the enforcement of any obligations of Guarantor under this Guaranty, and (b) interest at the Default Rate set forth in the Lease (not to exceed the maximum rate permitted under applicable law), from the date of such demand to Guarantor until the date paid, on any amounts due under this Guaranty that have not been paid to Landlord within ten (10) days after demand for such amounts has been made to Guarantor. Such obligations shall survive the payment and performance of the applicable Guaranteed Lease Obligations, whether by Tenant, Guarantor, or any other party.
- 7. Notice Address. All notices and demands to Guarantor hereunder shall be deemed given when delivered to Guarantor by hand delivery, by overnight courier, or by certified mail, return receipt requested, addressed to Guarantor at the same address as notices to the Tenant are required to be provided under the Lease, or to such other notice address(es) as Guarantor may from time to time specify to Landlord by written notice delivered to Landlord at Landlord's notice address(es) under the Lease. All notices, demands, and payments to Landlord hereunder shall be deemed given when delivered to Landlord by hand delivery, by overnight courier, or by certified mail, return receipt requested, addressed to Landlord at the same address as notices (or, as applicable, payments) to the Landlord are required to be provided under the Lease, or to such other notice or payment address(es) as Landlord may from time to time specify to Guarantor by written notice delivered to Guarantor at Guarantor's notice address(es) as provided herein.
- 8. <u>Confirmations</u>. The provisions of this Guaranty, including without limitation the continuation of the Guaranteed Lease Obligations under Section 3 above, shall be self-operative without the need for Guarantor to execute any further instrument. In confirmation thereof, if requested by Landlord in connection with any amendment of the Lease or any sale or financing of the Property or any interest therein, Guarantor shall execute and deliver an estoppel certificate, in a commercially reasonable form, confirming the Guaranteed Lease Obligations and the status of Guarantor's obligations under this Guaranty. In the event that Guarantor shall at any time cease to be a publicly traded company, then at Landlord's request in connection with a sale or financing of the Property or any interest therein or otherwise at the times permitted under the Lease for Landlord to request Tenant's financial statements, Guarantor shall provide Landlord with a copy of its most recent financial statements, which may be audited or certified in the manner set forth in the Lease for Tenant's financial statements.

- 9. <u>Jurisdiction, etc.</u> Guarantor hereby irrevocably and unconditionally submits to personal jurisdiction in The Commonwealth of Massachusetts over any suit, action, or proceeding arising out of this Guaranty or the Lease, and Guarantor hereby waives any right to object to personal jurisdiction within The Commonwealth of Massachusetts. The initiation of any suit, action, or proceeding by Landlord against Guarantor or any property of Guarantor in any other jurisdiction shall not constitute a waiver of the agreements contained herein that the law of The Commonwealth of Massachusetts shall govern the rights of Landlord and the rights and obligations of Guarantor under this Guaranty and that Guarantor submits to personal jurisdiction within The Commonwealth of Massachusetts. Guarantor hereby waives any right to a trial by jury for any claim arising under this Guaranty or the Lease.
- 10. Execution. Guarantor represents and warrants to Landlord that (a) Guarantor has a direct or indirect financial interest in the Tenant, (b) the individual(s) executing this Guaranty on behalf of Guarantor are duly authorized to execute and deliver this Guaranty on behalf of Guarantor, (c) this Guaranty is a valid and binding obligation of Guarantor enforceable in accordance with its terms, and (d) Guarantor's execution of and performance under this Guaranty violates no law, rule, regulation, agreement or contract applicable to or binding on Guarantor or Tenant.
- 11. General Provisions. If any term of this Guaranty, or the application thereof to any person or circumstance, shall to any extent be invalid or unenforceable, the remainder of this Guaranty, or the application of such term to persons or circumstances other than those as to which it is invalid or unenforceable, shall not be affected thereby, and each term of this Guaranty shall be valid and enforceable to the fullest extent permitted by law. A facsimile signature or other copy of this executed Guaranty shall be sufficient to prove the execution and delivery of this Guaranty, and any such copy or reproduction thereof shall be admissible in evidence in any judicial or administrative proceeding as an original of this instrument, whether or not the original has been lost or destroyed. This Guaranty may be amended only by an instrument in writing executed by Landlord and expressly stating Landlord's agreement to modify this Guaranty, and no amendment of this Guaranty shall be deemed to arise by reason of any course of conduct, oral or electronic communication, or failure or delay in enforcing any remedy against Tenant under the Lease or against Guarantor under this Guaranty.
- 12. Successors and Assigns. This Guaranty shall inure to the benefit of Landlord and its successors and assigns, including, without limitation, any assignee of Landlord's interest pursuant to an assignment or collateral assignment of Landlord's interest in the Lease in connection with a sale or financing of the Property or any interest therein. This Guaranty shall be binding on Guarantor and its successors and assigns, provided that Guarantor shall have no right, without the prior written consent of Landlord, to assign any of its rights, powers, duties, or obligations under this Guaranty. Any assignment of any rights or obligations under this Guaranty shall not relieve or discharge Guarantor from any obligations under this Guaranty. Accordingly, as used herein, references to "Landlord" shall mean the original Landlord under the Lease as set forth above, together with the successors and assigns from time to time of the interest of the Landlord under the Lease, and references to "Tenant" shall mean the original Tenant under the Lease as set forth above, together with the successors and assigns from time to time of the interest of the Tenant under the Lease.

This Guaranty is executed as a sealed Massachusetts instrument as of the date first set forth above.

GUARANTOR:

TRIPADVISOR, INC., a Delaware corporation

By: /s/Julie M.B. Bradley
Name: Julie M.B. Bradley
Title: Chief Financial Officer

Certification

- I, Stephen Kaufer, Chief Executive Officer of TripAdvisor, Inc., certify that:
- 1. I have reviewed this quarterly report on Form 10-Q for the period ended June 30, 2013 of TripAdvisor, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: July 24, 2013 /s/ STEPHEN KAUFER

Stephen Kaufer
President and Chief Executive Officer

Certification

I, Julie M.B. Bradley, Chief Financial Officer of TripAdvisor, Inc. certify that:

- 1. I have reviewed this quarterly report on Form 10-Q for the period ended June 30, 2013 of TripAdvisor, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: July 24, 2013 /s/ JULIE M.B. BRADLEY

Julie M.B. Bradley
Chief Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report on Form 10-Q of TripAdvisor, Inc. (the "Company") for the quarter ended June 30, 2013, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Stephen Kaufer, Chief Executive Officer of the Company, certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that:

- 1) the Report which this statement accompanies fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: July 24, 2013 /s/ STEPHEN KAUFER

Stephen Kaufer

President and Chief Executive Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report on Form 10-Q of TripAdvisor, Inc. (the "Company") for the quarter ended June 30, 2013, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Julie M.B. Bradley, Chief Financial Officer of the Company, certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that:

- the Report which this statement accompanies fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: July 24, 2013 /s/ JULIE M.B. BRADLEY

Julie M.B. Bradley Chief Financial Officer